ТШ

Closed Loop Current Control of a Grid Connected Transformerless Flying Capacitor Inverter

Master Thesis

to gain the academic degree

Master of Science (M.Sc.)

at the Technical University of Munich (TUM), Department of Electrical and Computer Engineering, Institute for Electrical Drive Systems and Power Electronics.

submitted by	Mirza Abdul Waris Begh, B.Tech. Matriculation Number: 03681054 mirza.begh@tum.de
on	November 28, 2018
1. Supervisor:	Eyke Liegmann, M.Sc. Institute for Electrical Drive Systems and Power Electronics, TUM
2. Supervisor:	Akshay Mahajan, M.Sc. Fraunhofer Institute for Solar Energy Systems, ISE

Abstract

Renewable sources of energy such as solar and wind energy are being increasingly used as alternatives to conventional energy resources. The primary energy of these sources varies widely in its nature due to various factors like weather conditions, irradiance level, wind speed, etc. Therefore, high efficiency power electronic converters are required to interface these sources with the grid. Specifically for solar power, there has been an increasing trend to use transformerless converters with multilevel output. The advantage of using such converters lies in the removal of the bulky transformers, implying smaller converters and better efficiencies. Moreover, control algorithms play a crucial role in the operation of grid-connected converters. Therefore, this thesis aims to examine linear and non-linear control techniques for a recently proposed transformerless inverter namely Siwakoti-H inverter, equipped with a LCL filter.

The linear controller is designed using pole placement to directly set the dominant dynamics of the controller and actively damp the resonant frequency of the filter. In addition, a major portion of this work deals with the application of model predictive control (MPC) for the control of Siwakoti-H inverter. MPC is an advanced control technique that has seen a lot of application in the past decade. Firstly, the direct MPC is introduced as a current controller. As a next step, MPC is used with a modulator to facilitate the use of long prediction horizons and reduce the computational effort. Finally, a non-linear model of the converter is proposed and its accuracy is verified using direct MPC scheme. The working of the proposed control algorithms is verified in simulations. Additionally, some experimental results are presented to prove that this topology can be controlled in the grid-connected mode.

Statement

The work in this thesis is based on the research jointly carried out at the Department of Power Converter Systems (PCS), Fraunhofer Institute for Solar Energy Systems ISE, Freiburg and the Institute for Electrical Drive Systems and Power Electronics, Technische Universität München (TUM). No sections of this thesis were submitted or used to gain another academic degree. Hereby I state, that this thesis was written alone by myself, except of explicitly marked passages.

Munich, November 28, 2018 Mirza Abdul Waris Begh

Acknowledgments

I find it hard to express my gratefulness to the Creator of the heavens and the earth, for bestowing upon me His deepest blessings and for the warmth and kindness He has showered upon me.

This Master's thesis is based on my research jointly carried out at the Department of Power Converter Systems (PCS), Fraunhofer Institute for Solar Energy Systems ISE, Freiburg, and the Institute for Electrical Drive Systems and Power Electronics, Technical University of Munich (TUM).

I would like to express my deepest gratitude to my supervisor Mr. Eyke Liegmann for his constant guidance, helpful comments, advice, and support. His guidance has been the backbone of progress during my thesis and his guidance helped me throughout the whole period of the thesis. Furthermore, I would like to express my sincere gratitude to my second supervisor Mr. Akshay Mahajan for giving me this wonderful opportunity to write my thesis at Fraunhofer ISE and for all the help, support and guidance during my stay in Freiburg. I thank him for his faith in me to work on this topic and for the long hours invested during hardware implementation. Besides, I deeply thank Dr. Yam Siwakoti, UTS Australia the designer of the topology investigated in this thesis, for continuous encouragement and his constant support during the thesis. Additionally, I extend sincere thanks to Dr. Petros Karamanakos, TUT Finland for his constant guidance and expert advice during the complete course of this thesis.

I am particularly grateful to Prof. Ralph Kennel, TUM and Mr. Stephan Liese, Fraunhofer ISE for giving me the opportunity to work on my thesis at their respective research groups with independence and flexibility.

Help from other colleagues is not only motivating but also important. I would therefore like to extend lots of thanks to Mr. Aswin Palanisamy (ISE) for helping me around with the hardware test-bench and investing a lot of time to troubleshoot various hardware issues. Many thanks to Mr. Benjamin Stickan (ISE) for all the healthy discussions and his comments and suggestions during the course of my thesis. I also thank Mr. Manuel Rauscher (ISE) for the fruitful discussions that I had with him and for guiding me during the software implementation on the TI microcontroller.

Words are inadequate to express my heartfelt gratitude to my parents and my siblings for the moral support and the trust they have in me. Without them, I could hardly finish this work.

Munich, November 28, 2018 Mirza Abdul Waris Begh

Contents

A	bstra	nct		iii					
\mathbf{St}	Statement v								
A	cknov	wledgr	nents	vii					
N	omer	nclatur	:e	xiii					
1	Intr	oducti	ion	1					
	1.1	Backg	round	. 1					
	1.2	Motiva	ation \ldots	. 3					
	1.3	Thesis	s objectives	. 4					
	1.4	Thesis	s organization	. 4					
2	The	eoretic	al Background	6					
	2.1	VSI T	opologies	. 6					
		2.1.1	Two-level inverter	. 6					
		2.1.2	Multilevel inverters	. 7					
			2.1.2.1 Neutral point clamped inverter	. 7					
			2.1.2.2 Flying capacitor inverter	. 8					
			2.1.2.3 H-Bridge inverter	. 9					
		2.1.3	Siwakoti-H flying capacitor inverter (sFCI)	. 9					
	2.2	Contro	ol schemes	. 10					
	2.3	State	feedback control	. 11					
		2.3.1	Solution of the state space system	. 12					
		2.3.2	Controllability	. 13					
		2.3.3	Kalman's controllability criterion	. 13					
		2.3.4	State feedback	. 14					
	2.4	Model	l predictive control	. 15					
		2.4.1	MPC: Operating principle	. 15					
		2.4.2	Prediction model	. 17					
		2.4.3	Constraints	. 17					
		2.4.4	Prediction horizon	. 17					
		2.4.5	Cost function	. 17					
		2.4.6	Receding horizon policy	. 19					
	2.5	Summ	nary	. 19					
3	Siw	akoti-I	H Topology: Analysis and System Modeling	20					
	3.1	Novel	flying capacitor topology	. 20					
		3.1.1	Description of the Siwakoti-H topology	. 21					
		3.1.2	Operating principle of the Siwakoti-H topology	. 21					

		3.1.3	Operating modes (states) of the Siwakoti-H topology	22
			3.1.3.1 Positive state (active) $\ldots \ldots 2$	22
			3.1.3.2 Negative state (active) $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 2$	22
			3.1.3.3 Zero state	23
	3.2	Analy	sis of sFCI behavior in open-loop	23
		3.2.1	Open-loop response	23
		3.2.2	Open-loop voltage comparison	24
		3.2.3	Bridge voltage analysis	26
		3.2.4	Three-step voltage behavior	27
		3.2.5	Operating cases of zero-state	28
	3.3	Mathe	ematical basics for system modeling	29
		3.3.1	Three-phase electrical systems	29
		3.3.2	Space vectors and transforms	29
			3.3.2.1 Clarke transformation $(a, b, c) \leftrightarrow (\alpha, \beta)$	30
			3.3.2.2 Park transformation $(\alpha, \beta) \leftrightarrow (d, q)$	31
	34	LCL f	ilters for grid connection $(\alpha, \beta) \lor (\alpha, q) \lor (\alpha, q)$	31
	0.1	3/1	Transfer function of LCL filter	21 21
		349	LCL filter resonance damping	33
		5.4.2	3 4 9 1 Passive Damping	22
			3422 Active Damping	24
	25	Sustor	n modeling of grid connected aFCI equipped with I CI filter	25
	3.0 3.6	Summ	in modeling of grid-connected si of equipped with DOD inter	26
	5.0	Summ	aly	00
4	Stat	te-Feed	Back Current Control3	87
	4.1	Contro	oller design for a single-phase sFCI	37
		4.1.1	Continuous-time model	38
		4.1.2	Discrete-time model	38
		4.1.3	Controllability	39
	4.2	Design	1 of the current controller	39
		4.2.1	Generalized integrators (GI)	40
		4.2.2	Augmented system for pole placement design	42
		4.2.3	Pole placement	42
		4.2.4	DC voltage feedforward compensation (DVFC)	14
	4.3	Perfor	mance evaluation of state-feedback control for single-phase sFCI	15
				tυ
		4.3.1	Steady-state performance	±5 15
		$4.3.1 \\ 4.3.2$	Steady-state performance 4 Dynamic performance 4	45 45 46
		$ \begin{array}{r} 4.3.1 \\ 4.3.2 \\ 4.3.3 \end{array} $	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4	45 45 46 47
	44	4.3.1 4.3.2 4.3.3 Obser	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4	15 15 16 17 17
	4.4 4.5	4.3.1 4.3.2 4.3.3 Obser Contro	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 pller design for a three-phase sFCI 4	15 15 16 17 17 17
	4.4 4.5	4.3.1 4.3.2 4.3.3 Obser Contro 4.5.1	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4	15 16 17 17 17 19
	$4.4 \\ 4.5$	4.3.1 4.3.2 4.3.3 Obser Contro 4.5.1 4.5.2	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Mathematical model of the converter 5	15 15 16 17 17 19 19
	4.4 4.5	4.3.1 4.3.2 4.3.3 Obserr Contro 4.5.1 4.5.2 4.5.3	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5	15 15 16 17 17 19 19 50 50
	4.4 4.5	4.3.1 4.3.2 4.3.3 Obser Contro 4.5.1 4.5.2 4.5.3 4.5.4	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5 Controller design 5	 45 46 47 47 49 49 50 50 50 50
	4.4 4.5	4.3.1 4.3.2 4.3.3 Obser Contro 4.5.1 4.5.2 4.5.3 4.5.4 Perfor	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5 Controller design 5 mance evaluation of state-feedback control for three phase sECI 5	 45 46 47 47 49 49 50 50 52 53
	4.44.54.6	4.3.1 4.3.2 4.3.3 Obserr Contro 4.5.1 4.5.2 4.5.3 4.5.4 Perfor 4.6 1	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5 Controller design 5 mance evaluation of state-feedback control for three-phase sFCI 5 Steady-state performance 5	15 15 16 17 17 19 19 50 50 50 52 53
	4.44.54.6	4.3.1 4.3.2 4.3.3 Obser Contro 4.5.1 4.5.2 4.5.3 4.5.3 4.5.4 Perfor 4.6.1 4.6.2	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5 mance evaluation of state-feedback control for three-phase sFCI 5 Steady-state performance 5 Dynamic performance 5 Dynamic performance 5 Steady-state performance 5 Dynamic performance 5 Dynamic performance 5	15 15 16 17 19 19 50 50 50 50 53 53 54
	4.4 4.5 4.6	4.3.1 4.3.2 4.3.3 Obserr Contro 4.5.1 4.5.2 4.5.3 4.5.4 Perfor 4.6.1 4.6.2	Steady-state performance 4 Dynamic performance 4 Comparison of bode plots 4 ver design 4 oller design for a three-phase sFCI 4 Motivation 4 Mathematical model of the converter 5 Grid synchronization in power converters 5 Controller design 5 mance evaluation of state-feedback control for three-phase sFCI 5 Steady-state performance 5 Dynamic performance 5 Optimic performance 5 Optimic performance 5	43 45 45 46 47 47 49 50 50 50 53 54 55 56 57 58 54 56

5	Mo	del Predictive Control using a Linearized Converter Model5'	7
	5.1	Motivation	7
	5.2	Finite-control-set model predictive control (FCS-MPC)	8
		5.2.1 Continuous-time mathematical model	8
		5.2.2 Converter bridge voltage: Space vectors	9
		5.2.3 Discrete-time model	0
		5.2.4 Control objectives $\ldots \ldots \ldots$	0
		5.2.5 Controller block diagram	0
		5.2.6 Control problem	0
		5.2.7 Reference generation	3
	5.3	Performance evaluation of FCS-MPC	4
		5.3.1 Steady-state performance	5
		5.3.2 Dynamic performance	5
	5.4	Drawbacks of classical MPC	7
		5.4.1 Drawback 1: Heavy computational efforts due to enumeration	7
		5.4.2 Drawback 2: Variable switching frequency	7
	5.5	Continuous-control-set model predictive control (CCS-MPC)	8
		5.5.1 Continuous-time mathematical model	8
		5.5.2 Discrete-time model	9
		5.5.3 Control objectives 6	9
		5.5.4 Control and optimization problem	9
		5.5.4.1 Derivation of optimization problem in vector form 6	9
		5.5.4.2 Solution in terms of the unconstrained minimum 7	'1
		5.5.5 Controller block diagram	'1
		5.5.6 Discussion 7	'1
	56	Performance evaluation of CCS-MPC 7	י יי
	0.0	5.6.1 Stondy state performance	⊿ '२
		5.6.2 Dynamic performance 7	บ '2
	57	Summary 7	ט יב
	0.1		0
6	Mo	del Predictive Control using a Non-linear Converter Model 7	6
	6.1	Motivation	6
	6.2	Analysis of the six working states of the sFCI	7
		6.2.1 Positive cycle	7
		$6.2.1.1 \text{Case 1: Positive state} \dots \dots$	7
		6.2.1.2 Case 2 : Zero-state with $i_m > 0$	8
		6.2.1.3 Case 3 : Zero-state with $i_m < 0$	8
		6.2.2 Negative cycle	9
		$6.2.2.1 \text{Case 4}: \text{Negative state} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	9
		6.2.2.2 Case 5 : Zero-state with $u_{dc} < u_{FC}$	0
		6.2.2.3 Case 6 : Zero-state with $u_{dc} > u_{FC}$	0
	6.3	Non-linear mathematical model	1
		6.3.1 Continuous-time model of the system	2
		6.3.2 Discrete-time model of the system	2
	6.4	Direct model predictive control	2
		6.4.1 Control scheme-I	4
		6.4.2 Control scheme-II	4
		6.4.3 Control problem	4
		6.4.4 Cost function	5
	6.5	Performance evaluation of DMPC	6
		6.5.1 System description	6

	6.6	6.5.2 6.5.3 Summ	Performance of DMPC during control scheme-I	86 88 88 89 90
7	Har	dware	Implementation of State Feedback Control	91
	7.1	Overv	iew	91
	7.2	Descri	ption of the testbench	91
		7.2.1	Hardware setup	91
		7.2.2	Data acquisition and synchronization	92
	7.3	Implei	mentation of the controller using C-code	92
		7.3.1	Current controller	93
		7.3.2	PLL	93
	7.4	Grid c	connection procedure	94
		7.4.1	Pre-charging the dc -link capacitors	94
		7.4.2	Filter capacitor voltage ramp-up	94
		7.4.3	Grid connection and current control	95
	7.5	Exper	imental measurements	95
		7.5.1	Converter Response for 50 V grid	96
		7.5.2	Converter Response for 80 V grid	96
	7.6	Discus	ssion and summary	97
8	Con	clusio	n and Future Outlook	99
	8.1	Conclu	usion	99
	8.2	Future	e outlook	101
AĮ	open	dices	1	103
\mathbf{A}	Des	criptio	on of the Test Bench	L 04
в	Ab B.1 B.2	Dreviat List of List of	ions and symbols 1 f abbreviations 1 f symbols 1	106 106 107
Bi	bliog	graphy	1	107

Nomenclature

Symbol	Meaning					
\mathbb{N}	$:= \{0, 1, 2, 3, \dots\}$, set of natural numbers					
$\mathbb{N}_{>0}$	$:= \{1, 2, 3, \dots\}$, set of positive natural numbers					
\mathbb{R}	$:= (-\infty, \infty)$, set of real numbers					
$\mathbb{R}_{>0}$	$:= (0, \infty)$, set of positive real numbers					
$\mathbb{R}_{\geq 0}$	$:= [0, \infty)$, set of positive real numbers including 0					
\mathbb{C}	$:= [-\infty, \infty)$, set of complex numbers					
In the following let	$n, m \in \mathbb{N}_{>0}.$					
$oldsymbol{x}$	$:= [x_1 \ x_2 \ \dots \ x_n]^\top \in \mathbb{R}^n$ (column) vector with $x_i \in \mathbb{R}$					
	$orall i \in \{1,2,\ldots,n\}$, all vectors are bold					
x	$\in \mathbb{R}, ext{scalar}, ext{all scalars are non-bold}$					
	$\begin{bmatrix} x_{11} & \ldots & x_{1m} \end{bmatrix}$					
X	$:=$ $:$ $:$ $:$ $\in \mathbb{R}^{n \times m}$, matrix with coefficients x_{ij}					
	$\begin{bmatrix} x_{n1} & \dots & x_{nm} \end{bmatrix}$					
	$\forall i \in \{1, 2, \dots, n\} \ \forall i \in \{1, 2, \dots, m\}$, all matrices are upper					
	case and bold					
х	$\in \mathbb{R}$, constant scalar, all constants are non-italic					
\dot{x}	$:= \frac{\mathrm{d}x}{\mathrm{d}t} \in \mathbb{R}$, first time derivative of $x \in \mathbb{R}$					
α, β	orthogonal axes of the stationary reference frame					
d,q	orthogonal axes of the rotating reference frame					
$\mathfrak{Re}\{x\}$	real part of the complex variable x					
$\mathfrak{Im}\{x\}$	imaginary part of the complex variable x					
$\ x\ _{\infty}$	infinity-norm of the vector x (largest absolute value)					

Chapter 1

Introduction

1.1 Background

Due to the uncertainty in the future supply of crude oil and the associated problems of environmental pollution, renewable energy sector has seen an increasing trend over the past decade, especially the solar and wind energy. The continuous growth of the renewable energy market is driven by its low carbon footprint, because carbon emissions are the major cause of global climate change. The global consumption of renewable energy is predicted to increase by an average of 2.6% (per year) till 2040 [1]. The integration of renewable energy resources to the grid requires an interfacing converter. Power electronic converters are considered as the backbone for energy conversion. For example, in solar power generation they convert the direct current (dc) from the photovoltaic (PV) system to alternating current (ac) which is then supplied to the grid [2]. Notable advances in semiconductor technology has enabled better and efficient power conversion. In order to increase the power quality and utilization of the PV system, new inverter topologies with better control algorithms are required to be proposed. This thesis aims to investigate some of these control techniques for a recently proposed transformerless inverter based on the principle of flying capacitor (FC).

In recent years, grid-tied inverters for PV systems have seen a rapid growth for both the utilityscale and distributed generation applications owing to government incentives, declining prices of PV panels and advancement in power electronics technology [3, 4]. The growth of PV has been exponential between 1992-2017. During this period, solar PV evolved from a niche market of small scale applications to a mainstream electricity source. There has been a huge growth from approximately 1.2 GW in 1992 to 401.5 GW in 2017. For example, see Figure 1.1 which shows the annual installed solar capacity for various regions of the world. Although, the European countries collectively have the highest amount of solar installations, China appears to take the lead in terms of the growth in solar installations. The European Photovoltaic Industry Association claims that by 2050 solar power will become the world's largest source of electricity [5]. In Germany it has become common to see PV systems installed on rooftops or PV farms of varied capacities next to the roads in the countryside.

A major part of the generated electricity from solar PVs is consumed immediately, since storage is comparatively expensive. Hence almost 99% of the installed PV capacity corresponds to gridconnected systems [4]. In grid-connected PV systems, storage is not necessary since all of the generated power is directly supplied to the utility grid. In this fashion traditional energy sources, like hydro or fossil fuels, can adapt to the power demand and behave as virtual storage devices to serve the purpose of power regulation and backup. However, solar power is inherently variable and intermittent in nature due to the day/night cycles and weather conditions. Moreover, due to its availability only during the day, storage of solar energy is potentially an important issue particularly in off-grid systems, to have continuous availability of electricity [5].

A typical configuration of a grid-connected PV system is presented in Figure 1.2. The PV



Figure 1.1: Yearly installed solar capacity [5].

panels generate a dc current that varies according to solar irradiance, temperature, and the terminal voltage [4]. The generated dc power must be converted and supplied to the grid as ac power. Additional components used include a grid connection filter, a transformer and a typical grid angle tracking system, e.g., a phase-locked-loop (PLL). An optional dc/dc conversion stage can be used between the PV panels and the inverter. A dc/dc converter is normally used to decouple the PV system operating point from the inverter and allow maximum power point tracking (MPPT) control. Additionally, the dc/dc stage can be used to boost the PV output voltage [4].



Figure 1.2: A generic structure of a grid connected PV system [4].

In PV applications, a transformer is often used for galvanic isolation and voltage transformations [3]. However, transformers are quite bulky and thus increase the size/weight of the inverter while reducing efficiency and power density. Due to the massive development of grid-connected PV systems, there has been an increasing trend to avoid the use of interfacing transformers and replace them by using transformerless inverter topologies. Although, transformerless topologies offer better efficiencies, less weight/volume ratio, and hence lower costs; removing the transformer gives rise to common-mode currents or leakage currents, due to the presence of parasitic capacitance between the PV panel and its frame [3]. Hence, additional measures have to be undertaken to eliminate the leakage currents and prevent any safety hazards. Consequently, grid-connected transformerless PV inverters must comply with strict safety standards [4]. Various research works have proposed different techniques to mitigate the common-mode currents, e.g., decoupling the dc from ac side and/or clamping the common-mode voltage (CMV) during the freewheeling period, or using common ground configurations [3]. Common-ground-type PV inverter has attracted a lot of interest from both industry and academia as it can effectively reduce the leakage current of the PV system. Many common ground transformer less inverters have been proposed, for e.g. in [6-12], but they either use more switches or more passive components. In [13] a new common-ground-type transformerless

inverter was proposed, consisting of only four switches and works on the principle of a flying capacitor. The new inverter allows direct connection between the grid-neutral and the negative-pole of the PV system in order to eliminate the leakage currents. In [3] a detailed design and analysis of the working of a family of transformerless inverters was presented.

1.2 Motivation

Power electronic converters are enabling technologies for control and conversion of electric power by virtue of a certain pattern of switching operations applied to semiconductor switches. In gridconnected power electronic systems quality of the current injected into the grid is a priority, whereas for renewable energy conversion efficiency of the converter is very important. These two objectives have been the main drivers behind the progress in the PV converter market and hence more and more PV systems are connected to the electrical grid via power electronics. The use of grid-connected converters allows independent control of active and reactive power exchanged with the grid. However, grid-connected converters generate harmonics which might be harmful for other devices connected to the same grid. In order to meet the grid-code requirement of injecting a clean current, with less total harmonic distortion (THD), it is crucial to use a filter (L or LCL filter) between the converter bridge and the grid [14]. Compared to the L filter, the LCL filter provides much higher attenuation of switching harmonics, and uses smaller components that reduce the volume and weight while increasing power density [15]. However, control of the LCL filters is complicated due to the presence of a resonant peak in its response. The resonant behavior of the LCL filter amplifies any harmonic components in its vicinity, thereby deteriorating the performance of the converter [16]. This behavior can either be damped by using passive methods which cause additional power losses or by using active damping methods [17].

Conventionally, the control of grid-connected inverters is carried out by employing voltage oriented control (VOC) [18, 19]. VOC uses linear controllers based on a simple proportional-integral (PI) controller, designed in the synchronous reference frame. Although these controllers are simple and can ensure a moderate quality of the injected grid current, the design of controller gains is often based on the linear model and hence does not include any non-linearity present in the system. Additionally, this controller suffers from the problem of integral-windup which necessitates the uses of anti-windup methods, hence the dynamic response of this scheme is slow.

Among other approaches, a proportional-resonant (PR) controller [20], state-feedback controller [21–24] and a predictive controller [25] have been proposed in literature. The PR controllers have been widely used for the control of grid-connected inverters due to their property to introduce infinite gain at selected frequencies, thereby mitigating the resonant behavior of LCL filters. In [21–23] a pole-placement based control strategy has been proposed that has inherent resonance damping and does not require any additional damping solutions. While [21] deals with the control design in continuous-time, [22,23] have proposed a controller in the discrete-time domain. Since pole-placement is generally not based on a systematic design procedure, [24] presents a controller design based on linear quadratic regulator (LQR), where a cost-function is minimized to place the poles in an optimal way.

Predictive Controllers are an alternative to linear controllers as they allow control of the system constraints, non-linearities and other requirements by using a single cost function. Model predictive control (MPC) uses a model of the system, called prediction model, to calculate (or predict) the future behavior of the controlled variables and then solve for the best possible prediction according to an optimization criterion [72]. Calculation of the predicted behavior and simultaneous optimization imposes an enormous calculation burden on the controller. However, due to continuous development in the semiconductor technology modern microprocessors have very high calculation power and are well suited for MPC applications.

MPC can handle multiple objectives simultaneously by incorporating them into a single cost

function. Moreover, MPC can successfully handle system and operational constraints that can be imposed on the variables of concern and thus allows the system to operate at its limits. Thanks to these characteristics, MPC is particularly effective for systems with complex dynamics, such as many power electronic systems [27]. MPC applied to power converters can be generally classified into two main approaches, continuous control set MPC (CCS-MPC) and finite control set MPC (FCS-MPC) [28]. While CCS-MPC generates a duty cycle and uses a modulator for generation of gating signals, FCS-MPC takes the advantage of the discrete nature of the power converter and hence the switches of the converter are directly manipulated without the need of a modulator [27, 29–31].

Motivated by the increasing trend in the use of transformerless topologies for grid-connected PV systems, the focus of this thesis is the design and implementation of various control strategies for a recently proposed Type-III, Siwakoti-H flying capacitor inverter (sFCI), see [3, Sec. II B], to enable subsequent connection to the grid.

1.3 Thesis objectives

Unlike the conventional three-level inverter topologies like neutral point clamped (NPC) and H-bridge, the sFCI topology has not been extensively analyzed for the design of a control algorithm. This thesis aims to design and implement different control techniques that can enable reference current tracking and voltage ripple control on the flying capacitor. This thesis aims to fulfill the following objectives:

- A thorough analysis of the operating states in order to understand the working of the sFCI and derivation of a mathematical model that describes the behavior of the linearized system.
- Realization and experimental evaluation of state-feedback controller and observer, using the pole placement approach for feedback gain calculation.
- Design and simulation of enumeration based FCS-MPC scheme using a linear model of the grid-connected three-phase sFCI.
- Design and simulation of a controller based on the CCS-MPC approach using the linear converter model.
- Derivation of a non-linear model of the grid-connected sFCI by incorporating the dynamics of the flying capacitor into the converter model, and validation of the derived non-linear model using direct MPC scheme.

1.4 Thesis organization

The upcoming chapters of this thesis are organized as follows:

Chapter 2 presents a literature review regarding the research problem. Firstly, it addresses the different multi-level inverter topologies which are widely used. Then, the sFCI is introduced as an interesting alternative to the conventional topologies. In addition, the basics of state-feedback control and MPC techniques are also discussed.

Chapter 3 discusses the sFCI topology in detail. The design, working of the sFCI and the open loop response for a resistive load is discussed. Further, a detailed analysis of the additional non-linearities of the sFCI is presented and a comparison is drawn with a conventional NPC converter. Finally, in chapter 3 a three-phase sFCI with LCL filter is presented and a linear system model is derived.

Chapter 4 proposes the design, simulation and analysis of a state-feedback controller and observer, in discrete time, for a single-phase sFCI. Further, the state-feedback controller is extended to the three-phase sFCI. Firstly the issues related to the three phase system control are discussed and then a controller, designed in the (d, q)-reference frame, is presented. Performance of the controller for grid current control is verified by simulations.

Chapter 5 uses the linearized converter model and is further divided mainly into two parts, where the first part deals with the design of a controller using FCS-MPC scheme, and the second part deals with a controller using the CCS-MPC scheme. Mathematical modeling, design and working are presented for both the schemes and each part concludes with the simulation results.

Chapter 6 discusses the problems associated with the new topology and proposes solutions. In particular, a non-linear model is derived and a control scheme based on this model is presented. Finally, the operation is verified using Matlab simulations.

Chapter 7 briefly discusses the hardware implementation of the state-feedback control on a singlephase testbench. The various modules used and methodology adopted is presented. Experimental results are included which affirm the working of the controller and the fact that this new inverter can be controlled in the grid-connected mode. Chapter 8 concludes this thesis.

Chapter 2

Theoretical Background

This chapter introduces the three-level inverter as an attractive alternative to the conventional two level voltage source inverter (VSI). First the different converter topologies used for grid-connected systems will be presented and later the sFCI will be introduced briefly. Finally, the conventional control and advanced control strategies for the multi-level inverters will be highlighted. This chapter is concluded by deriving a model for the system consisting of sFCI and LCL filter.

2.1 VSI Topologies

2.1.1 Two-level inverter

The most commonly used topology for a three-phase VSI with bi-directional energy flow capability is the two-level inverter (2LI), which can produce an output voltage having two levels [33]. This topology has a simple construction as there are only 6 switches with anti-parallel diodes, as depicted in Fig 2.1(a). The switches can be typically controlled with simple sine-wave pulse-width-modulation (PWM) signals. Since the semiconductor switches are switched at a very fast rate, the output current contains a lot of harmonics and specific filters have to be employed. Each phase in the 2LI has two states, namely 'P' and '0'. If the upper switch is turned *on*, the state is termed as 'P'. On the contrary, when the lower switch is *on* the state is termed as '0'. In general an *n*-level inverter can have a total of n^3 switching states. In case of 2LI there are 8 possible switching states, which give rise to 8 different voltage vectors, depicted in Fig. 2.1(b), with discrete voltage values $\in \{0, \pm \frac{1}{2}u_{dc}, \pm \frac{2}{3}u_{dc}\}$.



Figure 2.1: (a) Schematic diagram of the two level inverter. (b) Switching vectors of the 2LI.

However, the two-level inverters have several drawbacks including higher THD, high electromagnetic interference (EMI) and higher dv/dt stress. To overcome these drawbacks multi-level inverters are preferred as they offer more output voltages, improved current THD and hence better performance.

2.1.2 Multilevel inverters

Multilevel Inverters (MLIs) have received a lot of attention in the industry during the last decade because of their increased power rating, less EMI and lower switching losses owing to their ability to work at lesser switching frequencies. The term MLI was coined with the advent of the three-level inverters which have a staircase voltage output [32]. Over the last two decades a number of multilevel topologies have emerged, even with more than three voltage levels. The elementary concept of a multilevel converter is to achieve higher power by synthesizing a staircase voltage waveform by employing several lower voltage dc sources. Capacitors, batteries, and renewable energy sources can be used as the dc voltage sources [33].

A MLI has several advantages over a conventional two-level inverter including reduced dv/dt stress of the switches, less electromagnetic compatibility (EMC) problems, lower current distortion and ability to operate for a wide range of switching frequencies. However, the main drawback of MLIs is the usage of more semiconductor switches, which increases the gate driver requirement and possibly the conduction losses. Traditionally, the MLIs are categorized into three main topologies: 1) Neutral Point Clamped (NPC), 2) Cascaded H-Bridge (CHB) and 3) Flying Capacitor (FC).

2.1.2.1 Neutral point clamped inverter

Fig 2.2(a) shows the schematic of a diode-clamped inverter also known as neutral point clamped (NPC) inverter. It is the most widely used topology and was introduced by Nabae, et al., in 1981 [32]. NPC has been widely implemented in medium voltage applications as it exhibits a relatively high efficiency [34]. It consists of four switches and two diodes connected in a fashion as shown in Fig. 2.2(a). The NPC utilizes a single dc-link that is subdivided into a number of voltage levels by a string of series capacitors. For a three-level NPC (3L-NPC), the dc-link uses two capacitors with the mid-point termed as the neutral point. A diode is connected between the neutral point and the mid-point of the upper and lower legs. Considering the neutral point to be at zero potential the output bridge voltage of the NPC has three values $(-\frac{1}{2}u_{dc}, 0, +\frac{1}{2}u_{dc})$. Unlike a 2LI, each phase of the 3L-NPC has three switching states, denoted as $\{N, 0, P\}$ and summarized in Table 2.1.

Switching state	Notation	S_1	S_2	S_3	S_4	Bridge Voltage
Positive	P	1	1	0	0	$+\frac{1}{2}u_{dc}$
Zero	0	0	1	1	0	0
Negative	N	0	0	1	1	$-\frac{1}{2}u_{dc}$

Table 2.1: Switching states of 3L-NPC inverter.

In total the 3L-NPC can produce as total of $3^3 = 27$ different voltage vectors, see Figure 2.2(b). The voltage vectors can take discrete voltage values as

$$u \in \left\{0, \pm \frac{u_{dc}}{6}, \pm \frac{u_{dc}}{3}, \pm \frac{u_{dc}}{2}, \pm \frac{2u_{dc}}{3}\right\}$$

Out of these 27 vectors, only a subset of 19 vectors are unique and rest of them can be considered as redundant. These redundant vectors serve the purpose of voltage balancing in control applications. Although higher levels for a NPC can be obtained by increasing the number of switches and passive devices, balancing of voltages on the input capacitors becomes quite tricky and therefore, NPC is



Figure 2.2: (a) Schematic diagram of the three level NPC inverter [32]. (b) Switching vectors of the 3L-NPC [35].

limited to three levels. Its applications include Static VAR Compensator, variable speed drives and high power interconnections [34].

2.1.2.2 Flying capacitor inverter



Figure 2.3: Schematic diagram of three-level flying capacitor multilevel inverter [36].

Flying Capacitor multilevel inverter (FC-MLI) was introduced by Meynard et al., in 1992 [36]. The structure of the FC-MLI is similar to the NPC except the clamping diodes are now replaced by the clamping capacitors, as depicted in Fig. 2.3. This inverter topology provides more flexibility in waveform synthesis and voltage balancing, although control algorithms are more complex. This topology has a ladder structure of dc side capacitors, where the voltage on each capacitor differs from that of the next capacitor. The voltage increment between two adjacent capacitor legs gives the size of the voltage steps in the output waveform [33].

One advantage of the flying-capacitor based inverter is that it has redundancies for inner voltage levels; in other words, two or more valid switch combinations can synthesize an output voltage. Unlike the diode-clamped inverter, the flying-capacitor inverter does not require all of the switches that are conducting, to be in a consecutive series. Moreover, the redundancies allow a choice of charging/discharging specific capacitors and can be incorporated in the control system for balancing the voltages across the various levels [33].

2.1.2.3 H-Bridge inverter

Figure 2.4 shows a schematic construction of a single phase H-bridge converter. These inverters have attracted a lot of interest due to their ability to scale up to higher voltages by simply cascading several units. Thus to construct a multilevel phase-leg, series of single-phase full-bridge inverters will be used, with separate dc sources [37]. The bridge output voltage of the each unit has three values $(-\frac{1}{2}u_{dc}, 0, +\frac{1}{2}u_{dc})$, resulting in a staircase waveform.



Figure 2.4: Phase-leg of the cascaded H-bridge (CHB) inverter.

Output is obtained by sequential control of the four switches S_1 , S_2 , S_3 and S_4 . When S_1 and S_2 are switched on, the output is $+u_{dc}$; when S_3 and S_4 are switched on, the output is $-u_{dc}$; when either pair S_1 , S_3 or S_2 , S_4 are on, the output is 0. Addition of identical units in series would increase the output voltage levels. Although cascaded H-bridge inverter uses the least number of semiconductor devices when compared to other multi-level topologies, each unit requires a separate dc source and hence a large isolating transformer is required.

2.1.3 Siwakoti-H flying capacitor inverter (sFCI)



Figure 2.5: Phase leg of the Siwakoti-H inverter.

The sFCI is based on the topology proposed in [3, 13]. It is a common-ground-type transformerless inverter that works on the principle of a flying capacitor and has only four active switching elements, shown in Figure 2.5. The uniqueness of this topology is that, the negative voltage bus, required for the negative cycle, is fulfilled by using only a single input dc supply. This is achieved by the cyclic charging and discharging of the flying capacitor thus creating a virtual negative dc-link, as illustrated in Figure 2.6. Since the neutral of the grid can be directly connected to the negative pole of the dc-link, leakage current is automatically eliminated [37]. Phase leg of the three-level sFCI includes four power switches and one capacitor, as depicted in the Figure 2.5. Among the four switches, S_1 and S_4 are devices with bipolar voltage blocking capability whereas switches S_2 and S_3 are unipolar voltage devices. Therefore, the switches S_2 and S_3 are realized using a MOSFET while S_1 and S_4 are realized using a reverse blocking IGBT (RB-IGBT).



Figure 2.6: (a) Charging cycle of the flying capacitor, (b) discharging cycle of the flying capacitor where it acts as a virtual dc-link.

2.2 Control schemes

Figure 2.7 presents a classification of the control strategies used in grid-connected power electronic systems. As can be observed, the control of converters can be broadly classified into conventional and advanced strategies.



Figure 2.7: Control methods for grid-connected converters [38].

Among the linear control techniques, PI based Voltage Oriented Control (VOC) is very common as its implementation is simple and gives satisfactory steady-state performance. The non-linear behavior of the power converter is approximated to be linear for small time instants and hence a modulator can be employed for PWM generation [39]. A PWM modulator compares a sinusoidal reference signal to a triangular carrier signal, generating a pulsed waveform for switching the devices. In this fashion the inverter can operate between different states and hence generate voltages as desired. Since the dynamic performance of the VOC is quite slow, it is not preferred in practical systems. On the contrary, state feedback control (SFC) has attracted a lot of attention and has many applications in grid connected systems [21–24]. It has superior dynamic performance compared to PI control and better steady state performance. However, the drawback of applying linear schemes to control non-linear systems is that they can produce uneven performance throughout the dynamic range. With linear controller design, the various system constraints and requirements (like the maximum current and switching frequency or total harmonic distortion) cannot be directly incorporated [39].

Non-linear techniques have several advantages over their linear counterparts. In particular, Model Predictive Control is a technique which can control multiple conflicting objectives and allows to incorporate system constraints as well. With predictive control the cascaded structure, found in linear schemes, can be avoided so as to produce very fast transient responses [39].

2.3 State feedback control

Grid-connected converters offer uni/bi-directional power transfer with low current distortion. Grid codes set stringent limits on the grid current THD, which can be divided into two parts. The first issue is related to low frequency harmonics due to grid voltage background distortion. This is connected to the disturbance rejection capability of the current controller [23, 40]. High control bandwidth is required for good harmonic rejection. The second issue of grid current harmonic distortion is related to harmonics at the switching frequency of the converter. LCL filters are increasingly used for mitigation of switching harmonics due to the superior filtering characteristic above their resonant frequency [23].

For damping the resonant behavior of the LCL filter, active damping is usually preferred as it is more flexible and efficient compared to passive damping. However, as the LCL resonance dynamics get closer to the desired control bandwidth, it becomes difficult to simultaneously achieve resonance damping and high bandwidth with reasonable robustness [22]. State Feedback Control¹ (SFC) [21–24, 41–43] enables high dynamic performance of a converter, connected to the grid via an LCL filter. With SFC, the dominant and resonant dynamics can be simultaneously set according to desired specifications and provide a convenient way for resonance damping [21].

In State Feedback Control, the closed loop poles can be placed using different approaches : 1) dead-beat control; 2) using Bessel Functions; 3) optimizing a cost function as in linear quadratic (LQ) control; 4) selecting the desired pole locations directly [21]. LQ control is very attractive for design of a controller when dealing with very complex systems, because it provides an indirect method for optimal pole placement. However, solution of the Riccati equation and selection of the cost-function weights is a tedious process [21]. A simpler alternative to LQ control is direct pole placement (see [21-23]) where the controller gains can be expressed using the parameters of the system and dynamic performance specifications (i.e., the bandwidth of current controller and the resonance damping of the LCL filter) [21].

The idea of pole placement is to specify the closed-loop performance in terms of pole and zero locations. The control parameters can be tuned to achieve the specified locations. Although parameter calculation is straightforward, proper specification of closed-loop poles and zeros is necessary [22]. In the following section a general solution for state-space system is derived and the criteria for controllability is deduced.

¹In literature it is also called State Space Control.

2.3.1 Solution of the state space system

The dynamics of a linear time invariant (LTI) system can be represented using ordinary differential equations (ODE). A general LTI system in state space can be written as

$$\dot{\boldsymbol{x}}(t) = \boldsymbol{A}\boldsymbol{x}(t) + \boldsymbol{B}\boldsymbol{u}(t) \tag{2.1a}$$

$$\boldsymbol{y}(t) = \mathbf{C}\boldsymbol{x}(t) + \mathbf{D}\boldsymbol{u}(t) \tag{2.1b}$$

For the given LTI system, the state vector $\boldsymbol{x} \in \mathbb{R}^n$ has dimension n, the input vector $\boldsymbol{u} \in \mathbb{R}^p$ has dimension p and the output vector $\boldsymbol{y} \in \mathbb{R}^q$ has dimension q. Therefore, the state space matrices have dimension: $\mathbf{A} \in \mathbb{R}^{n \times n}$, $\mathbf{B} \in \mathbb{R}^{n \times p}$, $\mathbf{C} \in \mathbb{R}^{q \times n}$ and $\mathbf{D} \in \mathbb{R}^{q \times p}$. This set of equations including the dimensions of vectors and matrices is drawn in the following block diagram [44].



Let us consider the complete response of a linear system of first-order to an input u(t). The state equation $\dot{x} = ax + bu$ can be written in the form [45]

$$\dot{x}(t) - ax(t) = bu(t). \tag{2.2}$$

If both sides are multiplied by an integrating factor e^{-at} , the left-hand side becomes a perfect differential

$$e^{-at}\dot{x} - e^{-at}ax = \frac{d}{dt}\left(e^{-at}x(t)\right) = e^{-at}bu$$
(2.3)

which can be integrated directly to give

$$\int_{0}^{t} \frac{d}{d\tau} \left(e^{-a\tau} x(\tau) \right) d\tau = e^{-at} x(t) - x(0) = \int_{0}^{t} e^{-a\tau} b u(\tau) d\tau$$
(2.4)

and rearranged to give the state variable response explicitly:

$$x(t) = e^{at}x(0) + \int_0^t e^{-a(t-\tau)}bu(\tau)d\tau$$
(2.5)

The development of the expression for the response of higher order systems may be performed in a similar manner using the matrix exponential $e^{-\mathbf{A}t}$ as an integrating factor [45]. Matrix differentiation and integration are defined to be element by element operations, therefore if the state equation $\dot{\mathbf{x}}(t) = \mathbf{A}\mathbf{x}(t) + \mathbf{B}\mathbf{u}(t)$ is rearranged, and all terms pre-multiplied by the square matrix $e^{-\mathbf{A}t}$, we get:

$$e^{-\mathbf{A}t}\dot{\boldsymbol{x}}(t) - e^{-\mathbf{A}t}\mathbf{A}\boldsymbol{x}(t) = \frac{d}{dt}\left(e^{-\mathbf{A}t}\boldsymbol{x}(t)\right) = e^{-\mathbf{A}t}\mathbf{B}\boldsymbol{u}(t).$$
(2.6)

Integration of (2.6) gives

$$\int_0^t \frac{d}{d\tau} \left(e^{-\mathbf{A}\tau} \boldsymbol{x}(\tau) \right) d\tau = e^{-\mathbf{A}t} \boldsymbol{x}(t) - e^{-\mathbf{A}0} \boldsymbol{x}(0) = \int_0^t e^{-\mathbf{A}\tau} \mathbf{B} \boldsymbol{u}(\tau) d\tau, \qquad (2.7)$$

and because $e^{-\mathbf{A}0} = \mathbf{I}$ and $[e^{-\mathbf{A}t}]^{-1} = e^{\mathbf{A}t}$ the complete state space response may be written as

$$\boldsymbol{x}(t) = e^{-\mathbf{A}t}\boldsymbol{x}(0) + \int_0^t e^{\mathbf{A}(t-\tau)} \mathbf{B}\boldsymbol{u}(\tau) d\tau.$$
 (2.8)

To design a controller for the LTI system, linearization of the ODE is carried out in order to consider the behavior of the system around a reference or steady-state point. Additionally, controllability of the system has to be checked to verify that a controller can actually be designed.

2.3.2 Controllability

A system is controllable, if in finite time t_f any initial state $\boldsymbol{x}(0)$ can be driven to any given final state $\boldsymbol{x}(t_f)$ by appropriate choice of the control signal $\boldsymbol{u}(t)$ for $0 \le t \le t_f$, depicted in Figure 2.8. Considering the solution (2.8) of the state space system (2.1), for $0 \le t \le t_f$ we get

$$\underbrace{\boldsymbol{x}(t_f) - e^{-\mathbf{A}t_f} \boldsymbol{x}(0)}_{:= -e^{-\mathbf{A}t_f} \boldsymbol{x}_i} = \int_0^{t_f} e^{\mathbf{A}(t_f - \tau)} \mathbf{B} \boldsymbol{u}(\tau) d\tau.$$
(2.9)

The value \boldsymbol{x}_i is defined by setting the LHS equal to $-e^{-\mathbf{A}t_f}\boldsymbol{x}_i$. As the equation has to be valid for any $\boldsymbol{x}(t_f)$ and any $\boldsymbol{x}(0)$, the following equation has to hold for all $\boldsymbol{x}_i \in \mathbb{R}^n$ [44].

$$-e^{-\mathbf{A}t_f}\boldsymbol{x}_i = \int_0^{t_f} e^{\mathbf{A}(t_f - \tau)} \mathbf{B}\boldsymbol{u}(\tau) d\tau.$$
(2.10)

The system is controllable, if for any $\boldsymbol{x}_i \in \mathbb{R}^n$, a finite t_f and a control input $\boldsymbol{u}(t)$ for $0 \leq t \leq t_f$ can be found, such that (2.10) holds. In other words: by appropriate choice of $\boldsymbol{u}(t)$, the system can be driven from any initial state \boldsymbol{x}_i to the zero state in finite time t_f [44].



Figure 2.8: Driving a state $\boldsymbol{x}(0)$ to $\boldsymbol{x}(t_f)$ [44].

2.3.3 Kalman's controllability criterion

Consider (2.10) for a single-input single-output (SISO) system

$$-e^{-\mathbf{A}t_f}\boldsymbol{x}_i = \int_0^{t_f} e^{\mathbf{A}(t_f - \tau)} \mathbf{b}u(\tau) d\tau.$$
(2.11)

Using the definition of the matrix exponential we can write

$$-\boldsymbol{x}_{i} = \int_{0}^{t_{f}} e^{\mathbf{A}(\tau)} \mathbf{b} u(\tau) d\tau = \int_{0}^{t_{f}} \left(\sum_{\nu=0}^{\infty} \frac{(-\mathbf{A})^{\nu} \tau^{\nu}}{\nu!} \right) \mathbf{b} u(\tau) d\tau$$
$$= \sum_{\nu=0}^{\infty} \mathbf{A}^{\nu} \mathbf{b} \underbrace{\int_{0}^{t_{f}} \frac{(-1)^{\nu} \tau^{\nu}}{\nu!} u(\tau) d\tau}_{:=u_{\nu}}$$
(2.12)

Thus we get for x_i

$$\boldsymbol{x}_i = -\sum_{\nu=0}^{\infty} \mathbf{A}^{\nu} \mathbf{b} u_{\nu} \tag{2.13}$$

Hence for controllability, an arbitrary value x_i should be representable as a linear combination of the column vectors

$$\mathbf{b}, \mathbf{Ab}, \mathbf{A}^2 \mathbf{b}, \dots$$

As a consequence, the matrix composed of these n column vectors is called the controllability matrix \mathbf{M}_{c} and it must have a rank = n [45].

$$\mathbf{M}_{c} = \begin{bmatrix} \mathbf{b}, \mathbf{A}\mathbf{b}, \mathbf{A}^{2}\mathbf{b}, \dots, \mathbf{A}^{n-1}\mathbf{b} \end{bmatrix}$$
(2.14)

For MIMO systems the Kalman Controllability Criteria states: The LTI system $(\mathbf{A}; \mathbf{B}; \mathbf{C})$ is controllable if and only if the controllability matrix \mathbf{M}_{c} has full rank = n, where

$$\mathbf{M}_{c} = \begin{bmatrix} \mathbf{B}, \mathbf{A}\mathbf{B}, \mathbf{A}^{2}\mathbf{B}, \dots, \mathbf{A}^{n-1}\mathbf{B} \end{bmatrix}$$
(2.15)

2.3.4 State feedback

Once it is determined that the system (2.1) is controllable, assuming $\mathbf{D} = 0$, a feedback can be added to the system as shown below,



The state feedback controller is defined by

$$\boldsymbol{u}(t) = -\mathbf{K}\boldsymbol{x}(t) \tag{2.16}$$

Substituting (2.16) into the state ODE $\dot{\boldsymbol{x}}(t) = \boldsymbol{A}\boldsymbol{x}(t) + \boldsymbol{B}\boldsymbol{u}(t)$ yields the feedback system

$$\dot{\boldsymbol{x}}(t) = (\mathbf{A} - \mathbf{B}\mathbf{K})\boldsymbol{x}(t) \tag{2.17}$$

The control design has two objectives:

- Design a prefilter **N** such that for a reference input \boldsymbol{w} , the output vector $\boldsymbol{y}(t) \to \boldsymbol{w}$ for $t \to \infty$. \updownarrow The control law becomes $\boldsymbol{u}(t) = -\boldsymbol{K}\boldsymbol{x}(t) + \mathbf{N}\boldsymbol{a}\boldsymbol{x}$

The control law becomes $\boldsymbol{u}(t) = -\mathbf{K}\boldsymbol{x}(t) + \mathbf{N}\boldsymbol{w}$

2.4 Model predictive control

Power electronic converters are inherently non-linear due to the switching nature of the semiconductor devices. Moreover, a converter system has operational restrictions such as limits on the voltage and current stress of the switches, and protective operation limits of associated components. Thus linear control techniques cannot capture all the associated dynamics of these non-linear systems. Since linearization of the system is carried out around a specific operating point for linear control design, the system behavior might deviate at a different operating point. In order to improve the performance of the controlled system and achieve higher efficiencies, the true nature of the power converter and its characteristics must be taken into consideration. Additionally, the discrete nature of the semiconductor switches must also be considered.



Figure 2.9: Classification of MPC strategies applied to power electronics and drives [28].

Among the control strategies adopted in power electronics, model predictive control (MPC) has gained a lot of popularity due to its various advantages including the ability to easily handle system constraints and non-linearities. MPC has been shown to outperform traditional control methods because of its easy handling of system constraints that are imposed by defining the control problem as a constrained optimization problem [46]. With the advancement in microprocessors, the computational abilities of control platforms has seen a steep improvement over the years, making it possible to implement computationally demanding algorithms like MPC [39]. MPC is an advanced control method that relies on the system model to predict its future behavior. Using the future predictions, the controller decides a sequence of appropriate control actions based on a predefined optimization criteria [27]. For example, optimization criterion for hysteresis-based predictive control requires the actuation to maintain the controlled variable within the defined hysteresis bounds around the reference signal [47]. On the contrary MPC can incorporate flexible criteria as all the control objectives can be tackled in a single performance criterion, i.e., the cost function [28]. The easy handling of system constraints and non-linearities eliminates the need for linearization, and therefore the system does not have any restrictions w.r.t operation points.

Figure 2.9 shows the most common MPC techniques used for power converters and drives. Table 2.2 summarizes the main features of the different MPC techniques.

2.4.1 MPC: Operating principle

The prerequisite for designing a predictive controller is the model of the system to be controlled. MPC uses this model to predict the future behavior of the state variables of the system. Based on the predicted states and other system constraints, minimization of a cost function is carried out that describes the desired system behavior [39]. In general, the cost function is based on the error between the predicted output with the reference values. The MPC controller computes a set of optimal switching sequences based on the minimal cost and these sequences act as the firing signals for the semiconductor switches. Although, for each sampling period, a set of switching sequences that minimizes the cost function is calculated, only the first element of the switching sequence is

Parameter	GPC	EMPC	OSV-MPC	OSS-MPC
Modulator	PWM	PWM	Not required	Not required
Fixed switching frequency	Yes	Yes	No	Yes
Optimization	Online	Offline	Online	Online
Constraints	Can be included but increases the computational cost	Yes	Yes	Yes
Long prediction horizon	Yes	Yes	Can be used	Can be used
Formulation	Complex	Complex	Very intuitive	Intuitive

Table 2.2: MPC techniques used in power electronics [28].

applied. Moreover, MPC controllers solve an open-loop optimal control problem where the MPC algorithm is repeated in a receding horizon fashion at every sampling instant, thus providing a feedback loop and potential robustness to system uncertainties [28]. Figure 2.10 shows a basic block diagram of the MPC strategy, where i^* denotes the reference current and \hat{i} denotes the predicted current. Note that i^* , \hat{i} at the time step k + 2 are used instead of the time step k to compensate for the digital implementation delay [28, 48]. For detailed operation of this technique see [28, Sec. II].



Figure 2.10: Block diagram of a basic MPC strategy applied for the current control in a VSI with output RL load [28].

As shown in Figure 2.9, MPC is broadly classified into two methods depending on the kind of optimization problem. Out of these methods, Continuous Control Set MPC (CCS-MPC) calculates a continuous control signal and employs a modulator to generate the desired switching signals. The main advantage of CCS-MPC is that it produces a fixed switching frequency [28].

Finite Control Set MPC (FCS-MPC) does not need a modulator as it utilizes the discrete nature of the power converter to formulate the MPC algorithm, and hence generates the switching signals directly. It uses a control set of output voltage vectors to calculate predictions, reducing the optimal problem to an enumerated search algorithm [28]. The main disadvantage is that FCS-MPC generates a variable switching frequency [39].

In general, MPC algorithms require a significant number of computations. CCS-MPC usually has a lower computational cost than FCS-MPC because it computes part or all of the optimization problem offline. For this reason, CCS-MPC can address long prediction horizon problems. On the contrary, FCS-MPC requires that the optimization problem to be solved online. Since FCS-MPC involves a lot of calculations, it is usually limited to short prediction horizons in power electronic applications [28].

2.4.2 Prediction model

The dynamic evolution of a system can be described in the continuous time domain using a statespace model, described in section 2.3. Generally, the switching behavior of the power converter can be neglected and the mathematical model of the filter/grid system is sufficiently accurate for prediction purposes. Moreover, control algorithms like MPC are usually implemented using digital hardware platforms like FPGAs or DSPs. Hence the continuous time model of the system has to be discretized. For linear systems discretization can be carried by using the Forward Euler method or the Zero-Order-Hold method, although non-linear systems usually require a more complex approach (see for example [52, 53]). For a sampling time of T_s the discrete-time state space representation of the continuous time model (2.1) becomes

$$\boldsymbol{x}(k+1) = \mathbf{A}\boldsymbol{x}(k) + \mathbf{B}\boldsymbol{u}(k) \tag{2.18a}$$

$$\boldsymbol{y}(k) = \mathbf{C}\boldsymbol{x}(k) + \mathbf{D}\boldsymbol{u}(k) \tag{2.18b}$$

2.4.3 Constraints

Using semiconductor switches imposes certain restrictions or constraints on the controller and therefore, only a finite set of actuation options are allowed. In literature these restrictions are referred as *switching constraints*.

The control scheme CCS-MPC requires a modulator. The controller in this case generates the voltage reference for the modulator, which is restricted to a bounded continuous set, such as $S = [-1,1] \in \mathbb{R}$. On the contrary, the control scheme FCS-MPC directly generates the required switch positions of the converter, which is constrained to a finite set of integers. For example, a two-level converter (section 2.1.1) can synthesize two voltage levels per phase. This characteristic can be captured by the input constraint $S = \{0, P\} \approx \{0, 1\}$. Similarly, for a three level converter (section 2.1.2) the input constraint can be captured by the discrete set $S = \{N, 0, P\} \approx \{-1, 0, 1\}$. This kind of restrictions are known as "hard" constraints, as they cannot be relaxed [55].

In addition to this, restrictions can be imposed on the system states to prevent operation beyond the safe limits. For example, the upper constraint on the value of currents can be set to a value which is slightly less than the maximum allowed limit, in order to prevent damages due to over currents. This kind of restrictions are known as "soft" constraints [55].

2.4.4 Prediction horizon

Based on the prediction model the controller makes predictions for a defined length of N_p time-steps, known as the prediction horizon. This generates a sequence of outputs $\mathbf{Y} = [\mathbf{y}^{\top}(k+1), \mathbf{y}^{\top}(k+2), \ldots, \mathbf{y}^{\top}(k+N_p))]^{\top}$. The controller is tasked to drive the system in such a manner that it follows a reference trajectory (denoted as \mathbf{Y}_{ref}) over the prediction horizon. In order to minimize the deviations from the reference trajectory an optimization criterion is evaluated at each time step, for all possible switching states (which fulfill the system constraints). This is accomplished using a cost function.

2.4.5 Cost function

The cost function in MPC is a measure to define the behavior of the system. The control objectives can be fulfilled using the cost function, which can incorporate the different system states, outputs, and constraints into a scalar cost. It can be complex depending on which variables and control objectives are considered. The cost function facilitates a comparison between the predicted and the reference values [55]. A general definition of the cost function is

$$J = \sum_{l=k}^{k+N_p-1} (\boldsymbol{y}(l) - \boldsymbol{y}_{\text{ref}}(l))^\top \mathbf{Q}(\boldsymbol{y}(l) - \boldsymbol{y}_{\text{ref}}(l)) + \sum_{r=k}^{k+N_c-1} (\boldsymbol{s}(r))^\top \mathbf{R} \, \boldsymbol{s}(r)$$
(2.19)



(a) Trajectories of predicted current \mathbf{Y} , the reference trajectory \mathbf{Y}_{ref} (green dotted), and minimum cost trajectory (red).

(b) Corresponding candidate switching sequences.

Figure 2.11: Typical operation of FCS-MPC for a three level inverter.

where $\mathbf{y}(l)$ is a vector containing the predicted outputs, and $\mathbf{y}_{ref}(l)$ is a vector containing the reference values at the instant l, $\mathbf{s}(r)$ is the input vector at the instant r, N_p and N_c are the prediction and control horizons, respectively [28,56]. \mathbf{Q} and \mathbf{R} are matrices containing the weighting factors. Choice of the weighting factors and the states that are controlled impact the system's performance and stability. Therefore, it is a matter of tuning and requires some trial and error. To summarize, the cost function uses the difference vector $\Delta \mathbf{y}(l) = (\mathbf{y}(l) - \mathbf{y}_{ref}(l))$ and the sequence of manipulated variables (switching states that fulfill the system constraints)²

$$\mathbf{S}(k) = \left[\mathbf{s}^{\top}(k), \ \mathbf{s}^{\top}(k+1), \dots, \ \mathbf{s}^{\top}(k+N_c-1)\right]^{\top}$$
(2.20)

as arguments. Based on these two arguments, the future states and controlled variables can be predicted over the prediction horizon using the prediction model [55].

A typical example of a cost function where the matrices \mathbf{Q} and \mathbf{R} are diagonal, can be written as

$$J = \sum_{l=k}^{k+N_p-1} (\lambda_e J_e(l) + \lambda_s J_s(l))$$
(2.21)

For FCS-MPC, the controller does not need an external modulator and can therefore directly generate the set of optimal switching positions to be applied at the time instant k + 1. The cost function J is evaluated for all the candidate switching sequences $\mathbf{S}(k)$ over the prediction horizon N_p , and the set that corresponds to minimum cost is selected as the optimal switching sequence $\mathbf{s}_{opt}(k)$. The first term J_e corresponds to minimization of the reference tracking error, penalized by the weighting factor λ_e . The second term is J_s with a weighting factor λ_s , is related to penalization of the switchings for the power converter, which indirectly corresponds to reduction of switching frequency [47]. Considering a three level inverter where the controller has to decide between three discrete possibilities $\{-1, 0, 1\}$, evolution of predicted trajectories and corresponding switching sequences is presented in Figure 2.11.

²In literature these are also known as candidate switching sequences.

2.4.6 Receding horizon policy

The solution to the optimization problem at time step k, generates a sequence of manipulated variables \mathbf{S}_{opt} from time step k to $k + N_p - 1$. To provide feedback, only the first element of this sequence, namely $\mathbf{s}_{opt}(k)$, is applied to the system. At the next time step k + 1, a new state estimate is obtained and the optimization problem is solved again over the shifted horizon from k + 1 to $k + N_p$. This policy is known as receding horizon control [55].

2.5 Summary

In this chapter, different VSI topologies were discussed. Additionally, sFCI was introduced as an attractive alternative to conventional topologies, for PV applications. sFCI has a lot of potential for application in the PV field because it uses less number of switches and smaller additional components. Further, the basics and working principles of famous control techniques SFC and MPC were discussed. SFC and MPC are quite popular for grid connected applications. SFC, with its straightforward design can be used for control of linearized systems. On the other hand, MPC has a lot of potential to control non-linear and constrained systems. In this thesis an attempt is made to analyse the sFCI topology and design a closed loop for grid connected operation.

Chapter 3

Siwakoti-H Topology: Analysis and System Modeling

This chapter discusses the working of Siwakoti-H Flying Capacitor Inverter (sFCI). The design of the three level sFCI topology, it's working in different modes, various switching states will be discussed and the additional non-linearities present in the topology will be analysed. Furthermore, open-loop behavior of the sFCI in conjunction with an LCL filter will be presented. Finally, a continuous-time model of the inverter will be derived.

3.1 Novel flying capacitor topology

The sFCI was presented briefly in section 2.1.3, and is based on the topology proposed in [3,13]. Due to its novel nature, the Siwakoti-H flying capacitor inverter (sFCI) presents an interesting problem for analysis and system modeling. In particular, the behavior of the flying capacitor and its interaction with the LCL filter adds to the problem at hand. This topology is similar to a conventional T-type (3 level NPC) inverter, where it only uses four active switches; two of them with unipolar voltage blocking capability and two with bipolar voltage blocking capability [3]. However, the dc-link voltage requirement of this topology is only half of that required in NPC converter [32], or ANPC [57]. This reduces the front-end voltage boost requirement and associated circuitry, which helps in improving the efficiency and size of the system. In addition, it also reduces the size and number of passive component (capacitor) requirements.

Figure 3.1 illustrates a single phase sFCI with an LCL filter connected to a resistive load. It



Figure 3.1: Single-phase sFCI with LCL filter, resistive load and various measurements.

Parameter	Symbol	Value
Nominal power (base power)	P_n	5 kVA
Converter side inductance	L_m	$400\mu H$
Converter side resistance	R_m	$50\mathrm{m}\Omega$
Grid side inductance	L_{g}	$56\mu\mathrm{H}$
Grid side resistance	R_{g}	$30\mathrm{m}\Omega$
Filter Capacitance	C_{f}	$5\mu\mathrm{F}$
ESR of filter capacitor	\vec{R}	$7.4\mathrm{m}\Omega$
Flying Capacitor	C_{FC}	$680\mu\mathrm{F}$
DC link voltage	u_{dc}	$400\mathrm{V}$
Load Resistor	R_{load}	31.7Ω

 Table 3.1: System parameters from hardware test bench.

consists of a dc-link with a constant voltage $u_{dc} > 0$ which can be supplied from a renewable energy resource like PV. The LCL-filter consists of main inductor¹ and grid-side inductor, L_m and L_g , with internal resistances R_m and R_g respectively, and the filter capacitance C_f with electrostatic resistance (ESR) R. The hardware test-bench design of the converter and associated components was carried out in [37]. The various system parameters of the hardware test-bench are detailed in Table 3.1. For further analysis and simulations the same parameters will be utilized.

3.1.1 Description of the Siwakoti-H topology

The single phase of the sFCI, see Figure 3.1, consists of four active switching elements, $S_1 - S_4$, and a flying capacitor C_{FC} . Among the four switches, S_1 and S_4 are devices with bipolar voltage blocking capability whereas switches S_2 and S_3 are unipolar voltage devices. Hence, the switches S_2 and S_3 are realized using a MOSFET while S_1 and S_4 are realized using a reverse blocking IGBT (RB-IGBT). A RB-IGBT is a series combination of a normal IGBT and a power diode, depicted in Figure 3.2(a). It only conducts in the forward direction when the gating signal is supplied and the series diode is forward biased. Moreover, due to the presence of the series diode, RB-IGBT offers bidirectional voltage blocking capability.

In this topology, only one switch carries the load current during the positive or the negative cycle. Switch S_3 conducts the load current during the positive cycle and creates a positive operation state. Switch S_2 conducts the load current during the negative cycle and creates a negative operation state. When switches S_1 and S_4 are turned on, charging of the flying capacitor C_{FC} takes place, giving rise to the zero state. C_{FC} is continuously charged in both the positive and negative cycles. The inverter is modulated using a standard unipolar sinusoidal PWM (SPWM), where S_1 and S_4 experience high frequency switching in both positive and negative cycles for charging C_{FC} . The reduction in the number of active and passive components and a simple modulation technique makes this topology very versatile and useful for applications which require common bus bar configurations (positive or negative) [3].

3.1.2 Operating principle of the Siwakoti-H topology

The sFCI is a common ground type transformerless inverter that works on the principle of a flying capacitor. Figure 3.2(b) illustrates the the charging and discharging process of the flying capacitor to create a negative supply voltage for the inverter, during the negative cycle. When the switches are in position 1, the flying capacitor C_{FC} connects to the input voltage and charges up to the input voltage. However, when the switches turn to position 2, a negative voltage, with magnitude approximately

¹also referred as converter-side inductor.

equal to input voltage, appear at the output side. The process is continuously repeated at a high switching frequency to maintain a desired voltage across the output port [3].



Figure 3.2: (a) Realization of a reverse blocking IGBT (RB-IGBT). (b) Illustration of the charging and discharging of the flying capacitor (C_{FC}) to create a negative bus voltage in the inverter during the negative cycle [3].

3.1.3 Operating modes (states) of the Siwakoti-H topology

This topology has three modes of operation, i.e., positive active (P), negative active (N) and zero state (O). The zero state is common in both positive and negative cycles, e.g., in positive cycle, the switching sequence is POPO and in negative cycle, the switching sequence is NONO [3]. Figure 3.3 shows the different switching states of the topology and paths of current flow (red-dotted line represents the active current path, blue-dotted line represents the reactive current path, and green-dotted line represents the charging current path for C_{FC}). Switching table of the single phase sFCI is shown in Table 3.2.

States	Notation	S_1	S_2	S_3	S_4	Bridge Output
Positive state	Р	0	0	1	0	$+u_{dc}$
Negative state	Ν	0	1	0	0	$-u_{FC}$
Zero state	0	1	0	0	1	0

Table 3.2: Switching table of the sFCI.

3.1.3.1 Positive state (active)

During this state the positive modulating signal is compared with a triangular carrier wave to create switching pulses for the switch S_3 , which creates a unipolar positive voltage u_m at the output of the converter bridge. Switches S_1 and S_4 are off during this state, while the switch S_2 is off for the complete positive cycle. Since a MOSFET is used for the switch S_3 , the circuit has capability to conduct reactive current via the reverse diode, thus offering reactive power support. It should be noted that power in this state is completely supplied from the input dc supply. Figure 3.3(a) gives a clear picture of the positive state.

3.1.3.2 Negative state (active)

During this state the negative modulating signal is compared with a triangular carrier wave to create switching pulses for the switch S_2 . The Flying Capacitor acts as a virtual dc-link for the inverter during this operation state, and creates a negative unipolar voltage u_m at the output of the converter bridge. Switches S_1 and S_4 are off during this state, while the switch S_3 is off for the


Figure 3.3: Schematic overview of the three operation states and with respective paths for the current flows (red: active current path, blue: reactive current path, green: C_{FC} charging current path).

complete negative cycle. This state can also allow reactive power flow via the reverse diode of S_2 . Figure 3.3(b) gives a clear picture of the negative state.

3.1.3.3 Zero state

This state exists in both the positive and negative cycles. The main purpose of this state is to continuously charge the flying capacitor C_{FC} and maintain the voltage across C_{FC} around the input dc-link voltage ($\approx 400 \text{ V}$). The flying capacitor is designed to sustain a ripple voltage $(\delta_{FC} = max(u_{FC}) - min(u_{FC}))$, where the voltage u_{FC} is restricted by a maximum limit of 450 V. Zero state follows the active states, where both the switches S_1 and S_4 are turned on. Charging of the capacitor in both positive and negative cycles helps in reducing the size of the capacitor with the switching frequency, see Figure 3.3(c).

3.2 Analysis of sFCI behavior in open-loop

In order to design the closed loop controller, a model of the system under consideration must be derived. Conventional topologies like NPC, ML-FCI and H-bridge have been extensively studied in literature and mathematical models are readily available [32–34, 36]. However, sFCI is a novel topology and has been recently designed [3,13,37]. Hence, a detailed analysis of the working modes and system responses is necessary to understand the behavior. In the following discussion, analysis of the open-loop response of three-level sFCI (3L-sFCI) is undertaken and comparison to a three-level NPC (3L-NPC) converter is carried out to highlight the differences and additional non-linearities.

3.2.1 Open-loop response

Using the parameters give in Table 3.1 the open-loop system is simulated, in standalone version of PLECS (\mathbb{R}) , for a resistive load R_{load} . A unipolar sinusoidal PWM (SPWM) with a carrier frequency of 40kHz is employed to generate the switching signals for the four switches, see Figure 3.4.



Figure 3.4: SPWM strategy for generation of gating signals [37].

Figures 3.5 and 3.6 show the response of the sFCI equipped with an LCL filter and connected to a resistive load of 31.7Ω . Figure 3.5 shows a comparison between the voltage and current waveforms of the two *dc*-links (u_{dc} and u_{FC}). Unlike a three-level NPC converter where the two input capacitors are coupled and dependent on each other, it can be concluded that the input capacitor C_{in} and the flying capacitor C_{FC} do not have any specific coupling, although C_{FC} is charged by the energy supplied by C_{in} . Figure 3.6(a) shows the bridge voltage at the output of the converter and the filtered voltage across the load. Figure 3.6(b) shows the three states of the LCL filter. Comparing the main inductor current (red) and grid side current (blue) we can conclude that the LCL filter offers very good attenuation of switching frequency harmonics and filters out the current ripple.



Figure 3.5: Voltage and current waveforms of (a) the input dc supply and (b) the flying capacitor.

3.2.2 Open-loop voltage comparison

Figure 3.7 shows the voltage comparison in per unit (p.u.) form. Each plot consists of the load voltage u_{load} (red color) and the sinusoidal PWM reference u_{ref} (blue color). Compared to the NPC which has a sinusoidal response for the load current, sFCI does not appear to give a purely sinusoidal response as there is quite visible deviation compared to the reference voltage. Particularly, the deviation is prominent in the negative half cycle. To understand this behavior further, investigation of the converter bridge voltage provides more insight.



Figure 3.6: (a) Voltage across the converter bridge and the load and (b) measurements of the states of the LCL filter.



Figure 3.7: Comparison of load voltages during open-loop operation



Figure 3.8: Comparison of the output voltage u_m of the converter bridge.

Figure 3.8 shows a comparison between the bridge voltages u_m of the sFCI and NPC converter. During the positive cycle, the sFCI has an input dc voltage of 400 V and hence the voltage across the bridge is $\approx u_{dc}$. However, for the negative cycle flying capacitor acts as the dc supply and supplies the load. In Figure 3.5(b) it was observed that the flying capacitor has a voltage ripple $\delta_{FC} \approx 40 \text{ V}$, and it discharges continuously during the negative state. This slow decrease in the voltage of the flying capacitor is visible at the bridge output and hence responsible for distorted voltage output. On the contrary, the bridge voltage output of the NPC is sinusoidal and has minimal distortions, because the input dc supply is reasonably constant.



3.2.3 Bridge voltage analysis

Figure 3.9: Bridge output voltage with distortion regions marked with red/blue, and flying capacitor voltage.

Figure 3.9 shows the converter bridge voltage and flying capacitor voltage for a single cycle with regions marked in red/blue specifying the distorted regions. Figure 3.10 presents a zoomed view of three specific regions. A detailed explanation is as follows:

- (a) The peak bridge output voltage follows the slow decrease in the C_{FC} voltage. As can be seen, peak output voltage u_m varies between -430 V and -390 V due to the voltage ripple of the flying capacitor (δ_{FC}). From 0.03 s to 0.036 s the flying capacitor discharges without charging during the zero state, whereas 0.036 s onwards the flying capacitor discharges and subsequently charges during the zero state, hence a slight increase is observed.
- (b) It shows the bridge voltage u_m during the positive cycle when zero state is active. Unlike the NPC converter, where bridge voltage during the zero state = 0, sFCI does not show similar behavior. The main reason for this is that the voltage of the C_{FC} is higher than the input *dc*-link. Due to this a negative voltage drop appears across the switch S_4 , i.e., $u_{S_4} = (u_{dc} - u_{FC}) < 0$, and subsequently the series diode of the RB-IGBT is reverse biased. Hence, S_4 does not conduct any current for most of the positive cycle (S_4 conducts during the region (d), discussed later), even though it is supplied with switching signals. This negative voltage drop u_{S_4} appears at the output and keeps increasing as the FC is charging continuously (only during zero state).



Figure 3.10: Open-loop converter bridge voltage u_m : zoom-in view (a) distortion during negative state, (b) distortion during zero-state of positive half cycle, (c) distortion during zero-state of negative half cycle.

(c) It shows the bridge voltage u_m during the negative cycle when zero state is active. Again, this is in contrast to the behavior of conventional topologies like NPC, where voltage at the bridge is essentially zero. In this case, from 0.03 s to 0.035 s a small voltage drop appears across the output because switch S_4 conducts the main inductor current, while no charging currents flow as $(u_{dc} < u_{FC})$. Beyond 0.035 s $(u_{dc} > u_{FC})$, hence charging currents flow and switch S_4 conducts both the main inductor current i_m and C_{FC} charging current i_{FC} . Therefore, increase in voltage drop across the switch S_4 is observed.

3.2.4 Three-step voltage behavior

In addition to the distorted regions (a,b,c) discussed above, the bridge voltage u_m , see Figure 3.9, has an additional region marked in blue. A zoomed-in view of this region is presented in Figure 3.11. In general, the conventional topologies like NPC can only generate two voltage values during one switching period, i.e., either positive and zero or negative and zero. However, from Figure 3.11 it is observed that the Siwakoti-H inverter generates three different voltage values in a single switching period. This behavior is primarily due to the ripple of the flying capacitor. As discussed earlier, switch S_4 does not conduct for most part of the positive cycle. But, near the transition from positive to negative cycle the main inductor current i_m goes negative due to the current ripple. It is clear from Figure 3.6(b) that the main inductor current i_m has a current ripple ≈ 5 A. In this situation, series diode of the switch S_4 becomes forward biased due to high di/dt and hence S_4 conducts for a part of the switching period. Whenever S_4 conducts, the output voltage changes from a negative value (u_{S_4}) to a voltage ≈ 0 . Hence, instead of a normal two voltage level behavior, three stepped voltages are observed in the sFCI.



Figure 3.11: Open-loop converter bridge voltage: three-step behavior.

3.2.5 Operating cases of zero-state



Figure 3.12: Zero-state: four different operating modes.

In [3,13] only three working states of the sFCI were proposed, and described in section 3.1.3. However, from the previous discussion we can conclude that *zero state* does not actually generate zero voltage at the output of the converter. Rather the zero state can be divided into four cases based on the cycle of operation and other conditions. The proposed operating modes of the zero state are illustrated in Figure 3.12. In all the four cases, switches S_1 and S_4 are supplied with gating signals. Explanation of the four cases is as follows:

(a) This case exists during the positive cycle. When the main inductor current $i_m > 0$, the voltage across switch S_4 is negative due to the voltage ripple δ_{FC} and therefore S_4 does not conduct. Moreover, the charging current i_{FC} flows via $S_1 \rightarrow C_{FC} \rightarrow L_m$.

- (b) This case also exists during the positive cycle. When the main inductor current i_m becomes negative due to its ripple, the series diode of S_4 is forward biased [for details see section 3.2.5]. Hence charging current i_{FC} flows via $S_1 \to C_{FC} \to S_4$.
- (c) This case is existent during the negative cycle when $(u_{dc} < u_{FC})$. In this case charging current i_{FC} does not flow as the voltage u_{FC} is more than the input *dc*-link. The main inductor current i_m is negative and flows through S_4 , as the switch S_2 is turned OFF.
- (d) This case also occurs during the negative cycle. When $(u_{dc} > u_{FC})$, charging current i_{FC} flows via $S_1 \rightarrow C_{FC} \rightarrow S_4$. Additionally, i_m also flows through S_4 as the switch S_2 is OFF. Hence, the current stress on S_4 is maximum in this case.

3.3 Mathematical basics for system modeling

In this section, the utilized nomenclature for three-phase electrical systems is introduced. In contrast to the general complex representation/notation, the space vectors are introduced in the *vector-/matrix* representation.

3.3.1 Three-phase electrical systems

Three phase systems are described using vector variables. In general, the following vector

$$\boldsymbol{x}^{abc}(t) := \begin{bmatrix} x_a(t) \\ x_b(t) \\ x_c(t) \end{bmatrix} := \begin{bmatrix} \tilde{x}_a(t)\cos(\phi_a(t)) \\ \tilde{x}_b(t)\cos(\phi_b(t)) \\ \tilde{x}_c(t)\cos(\phi_c(t)) \end{bmatrix}, \qquad \boldsymbol{x}^{abc} \in \mathbb{R}^3$$
(3.1)

with sinusoidal components is defined, see [58, Sec. 2]. The phase quantities $x_a(t), x_b(t), x_c(t)$ have amplitudes $\tilde{x}_a(t), \tilde{x}_b(t), \tilde{x}_c(t)$ and phase angles $\phi_a(t), \phi_b(t), \phi_c(t) \in \mathbb{R}$ (in rad), respectively. The variable \boldsymbol{x} corresponds to the phase voltage vector \boldsymbol{u} (in V)³ or the phase current vector \boldsymbol{i} (in A)³.

In this thesis, mainly *balanced* three phase electrical grid is considered. Hence, the following assumptions can be imposed on the phase quantities:

Assumption (A.1): All phases have same amplitude, i.e.

$$\forall t \ge 0 \,\mathrm{s}: \qquad \tilde{x}(t) := \tilde{x}_a(t) = \tilde{x}_b(t) = \tilde{x}_c(t). \tag{3.2}$$

Assumption (A.2): Each phase is shifted from one another by an angle of $\frac{2}{3}\pi$, i.e.

$$\forall t \ge 0 \,\mathrm{s}: \qquad \phi(t) := \phi_a(t) = \phi_b(t) + \frac{2}{3}\pi = \phi_c(t) + \frac{4}{3}\pi. \tag{3.3}$$

3.3.2 Space vectors and transforms

The use of space vectors to represent the three-phase systems is well established in practice [58]. The three phases of the an electrical grid are *spatially* displaced from each other by $120^{\circ} = \frac{2}{3}\pi$. The three phases *a*, *b*, *c* are plotted as vectors in Figure 3.13, where all the electrical quantities can be described along the three axes. In order to simplify the calculations and reduce the number of the manipulated variables, orthogonal coordinate system is generally used, where the three phase quantities can be transformed to a two phase vector space. In Figure 3.13, two orthogonal coordinate systems are shown:

1. The stationary $s = (\alpha, \beta)$ -reference frame with the axes α, β and the vector $\mathbf{x}^s = (x^{\alpha}, x^{\beta})^{\top}$. The α -axis of the *s*-reference frame/coordinate system is aligned with the *a*-axis of the general (a, b, c) coordinate system.



Figure 3.13: Various reference frames (or coordinate systems): general (a, b, c)-reference frame, stationary $s = (\alpha, \beta)$ -reference frame, and rotating k = (d, q)-reference frame. The vector $\boldsymbol{x}^s = (x^{\alpha}, x^{\beta})^{\top}$ has a length $\| \boldsymbol{x}^s \| = \sqrt{(x^{\alpha})^2 + (x^{\beta})^2}$ [58].

2. The rotating k = (d, q)-reference frame with the axes d, q and the vector $\mathbf{x}^k = (x^d, x^q)^{\top}$. The d-axis of the k-reference frame/coordinate system is displaced by an angle ϕ_k from the a-axis of the general (a, b, c) coordinate system.

3.3.2.1 Clarke transformation $(a, b, c) \leftrightarrow (\alpha, \beta)$

The mathematical conversion of the three phase quantities $\boldsymbol{x}^{abc}(t)$ to the stationary *s*-reference frame is known as the *Clarke Transformation*. The space vector \boldsymbol{x}^s can be described in the Cartesian (α, β) -coordinate system as shown in Figure 3.13, since three linearly dependent quantities can be expressed by two linearly independent variables. The space vector \boldsymbol{x}^s can be summarized as a complex number:

$$\boldsymbol{x}^{s} = \frac{2}{3} (x_{a} + x_{b} e^{j \cdot \frac{2}{3}\pi} + x_{c} e^{j \cdot \frac{4}{3}\pi}) = \frac{2}{3} \left[x_{a} - \frac{1}{2} x_{b} - \frac{1}{2} x_{c} + j \left(\frac{\sqrt{3}}{2} x_{b} - \frac{\sqrt{3}}{2} x_{c} \right) \right]$$
(3.4)

The transformation from phase-values to space vector can also be expressed in the matrix notation:

$$\boldsymbol{x}^{\alpha\beta}(t) = \begin{bmatrix} \boldsymbol{x}^{\alpha} \\ \boldsymbol{x}^{\beta} \end{bmatrix} = \frac{2}{3} \cdot \underbrace{\begin{bmatrix} 1 & -\frac{1}{2} & -\frac{1}{2} \\ 0 & \frac{\sqrt{3}}{2} & -\frac{\sqrt{3}}{2} \end{bmatrix}}_{=:\mathbf{T}_{c} \in \mathbb{R}^{2 \times 3}} \cdot \boldsymbol{x}^{abc}(t)$$
(3.5)

Here the factor $\frac{2}{3}$ is use to scale the length of the space vector to the amplitude of the phase values, and hence yields an amplitude correct transformation. Also, the inverse Clarke transformation $(\alpha, \beta) \leftrightarrow (a, b, c)$ can be written as

$$\boldsymbol{x}^{abc}(t) = \begin{bmatrix} x_a \\ x_b \\ x_c \end{bmatrix} = \underbrace{\begin{bmatrix} 1 & 0 \\ -\frac{1}{2} & \frac{\sqrt{3}}{2} \\ -\frac{1}{2} & -\frac{\sqrt{3}}{2} \end{bmatrix}}_{=:\mathbf{T}_c^{-1} \in \mathbb{R}^{3 \times 2}} \cdot \boldsymbol{x}^{\alpha\beta}(t).$$
(3.6)

3.3.2.2 Park transformation $(\alpha, \beta) \leftrightarrow (d, q)$

The previous section discussed the stationary coordinate system (α, β) . If any space vector is rotated counterclockwise by an angle ϕ_k , the rotation can be described by the matrix:

$$\mathbf{T}_{\mathrm{p}}(\phi_{\mathrm{k}}) = \begin{bmatrix} \cos\left(\phi_{\mathrm{k}}\right) & -\sin\left(\phi_{\mathrm{k}}\right) \\ \sin\left(\phi_{\mathrm{k}}\right) & \cos\left(\phi_{\mathrm{k}}\right) \end{bmatrix}$$
(3.7)

For three-phase electrical systems this transformation is famous as *Park Transformation* [59]. Figure 3.13 illustrates the definition of the angle ϕ_k . By using the *Park transformation* stationary quantities $\boldsymbol{x}^s = (x^{\alpha}, x^{\beta})^{\top}$ can be represented in the rotating k = (d, q)-reference frame. The conversion between the stationary quantities and quantities in the k-reference frame is given by

$$\boldsymbol{x}^{dq}(t) = \begin{bmatrix} \boldsymbol{x}^{d} \\ \boldsymbol{x}^{q} \end{bmatrix} = \underbrace{\begin{bmatrix} \cos\left(\phi_{k}\right) & \sin\left(\phi_{k}\right) \\ -\sin\left(\phi_{k}\right) & \cos\left(\phi_{k}\right) \end{bmatrix}}_{=:\mathbf{T}_{p}^{-1}(\phi_{k})\in\mathbb{R}^{2\times2}} \cdot \boldsymbol{x}^{\alpha\beta}(t) \quad \Leftrightarrow \quad \boldsymbol{x}^{\alpha\beta}(t) = \underbrace{\begin{bmatrix} \cos\left(\phi_{k}\right) & -\sin\left(\phi_{k}\right) \\ \sin\left(\phi_{k}\right) & \cos\left(\phi_{k}\right) \end{bmatrix}}_{=:\mathbf{T}_{p}(\phi_{k})\in\mathbb{R}^{2\times2}} \cdot \boldsymbol{x}^{dq}(t) \quad (3.8)$$

Here, $\mathbf{T}_{p}(\phi_{k})$ is the Park transformation matrix and $\mathbf{T}_{p}(\phi_{k})^{-1}$ is the inverse Park transformation matrix. Table 3.3 presents an overview of the different coordinate systems relevant for our application.

Reference Frame	Abbrev.	Angle	Notation	Transform
General	(a, b, c)	-	$oldsymbol{x}^{abc}$	-
Stationary	(lpha,eta)	-	$oldsymbol{x}^{lphaeta}$	Clarke
Rotating	(d,q)	ϕ_k	$oldsymbol{x}^{dq}$	Clarke & Park

Table 3.3: Useful frames of reference.

3.4 LCL filters for grid connection

3.4.1 Transfer function of LCL filter

A single phase LCL filter, depicted in Figure 3.14, is considered for the derivation of the transfer function. Considering the LCL filter and invoking Kirchhoff's laws, the differential equations that define the dynamics of the filter can be summarized as:

$$0 = u_m - L_m \frac{\mathrm{d}i_m}{\mathrm{d}t} - R_m i_m - u_f \tag{3.9a}$$

$$0 = u_f - R_g i_g - L_g \frac{\mathrm{d}i_g}{\mathrm{d}t} - u_g \tag{3.9b}$$

$$0 = i_m - i_f - i_g \tag{3.9c}$$

$$0 = i_f - C_f \frac{\mathrm{d}}{\mathrm{d}t} (u_f - Ri_f) \tag{3.9d}$$

Using Laplace transform and rearranging, the set of equations (3.9) can be written as:

$$i_m = i_f + i_g \tag{3.10}$$

$$u_m - u_f = i_m (sL_m + R_m)$$
(3.11)

$$u_f - u_g = i_g(sL_g + R_g) (3.12)$$

$$u_f = i_f \left(\frac{1}{sC_f} + R\right) \tag{3.13}$$



Figure 3.14: Single-phase schematic of LCL filter.

The notations used in the LCL filter are as follows:

- u_m : Converter Bridge output voltage.
- L_m : Main inductor.
- R_m : Internal resistance of main inductor.
- i_m : Main inductor current.
- u_f : Voltage drop across filter capacitor.
- i_f : Filter capacitor current.

- C_f : Filter capacitance.
- R: Parasitic resistance of filter capacitor.
- L_g : Grid side inductor.
- R_g : Internal resistance of grid side inductor.
- i_g : Grid current.
- u_g : Grid voltage



Figure 3.15: Block diagram of the LCL filter [60].

The block diagram of the filter is shown in Figure 3.15. From (3.12) and (3.13), we can write

$$i_f\left(\frac{1}{sC_f} + R\right) = i_g(sL_g + R_g) \tag{3.14a}$$

$$\Rightarrow \qquad i_f \qquad = i_g \frac{s^2 C_f L_g + s C_f R_g}{s C_f R + 1} \tag{3.14b}$$

Substituting (3.10), (3.12) and (3.14b) into (3.11) and rearranging the terms, the bridge voltage can

be expressed as:

$$u_m = i_g (sL_g + R_g) + (i_g + i_f)(sL_m + R_m)$$

= $i_g (sL_g + R_g) + \left(i_g + i_g \frac{s^2 C_f L_g + s C_f R_g}{sC_f R + 1}\right) (sL_m + R_m)$ (3.15)

$$\Rightarrow u_m = i_g \left(sL_g + R_g + sL_m + R_m + \frac{(sL_m + R_m)(s^2C_fL_g + sC_fR_g)}{sC_fR + 1} \right)$$
(3.16)

Hence, the output-to-input transfer function $i_g(s)/u_m(s)$ of the LCL filter can be written as

$$\boldsymbol{H}(s) = \frac{sC_f R + 1}{s^3 L_g L_m C_f + s^2 C_f (L_g (R + R_m) + L_m (R + R_g)) + s(L_g + L_m + C_f (RR_g + RR_m + R_m R_g)) + R_g + R_g}$$
(3.17)

Using the system parameters from Table 3.1, the resonant frequency of the LCL filter can be calculated as

$$\omega_r = \sqrt{\frac{L_m + L_g}{L_m L_g C_f}} = 63.808 \cdot 10^3 \text{rad/s} \qquad \Rightarrow f_r = 10.155 \text{kHz}$$
(3.18)

Figure 3.16 shows the bode plot considering a loss-less filter. The resonant peak is near 10kHz.



Figure 3.16: Bode plot of the LCL filter.

3.4.2 LCL filter resonance damping

Figure 3.16 shows the bode plot of the uncompensated and undamped LCL filter. The resonant peak is clearly visible and poses a challenge for the design of a controller. This resonant behavior of the LCL filter must be damped adequately using either a passive damping component(s) or an active damping (AD) strategy. If the resonant frequency is not adequately damped, the introduction of an LCL filter may worsen the performance of the system due to increased grid current distortion [25].

3.4.2.1 Passive Damping

Passive damping is achieved by adding a resistance in series or in parallel with the capacitance C_f or inductance L_m of the filter. Two possible solutions in which a series resistance is added are shown in Fig. 3.17.



Figure 3.17: Possible positions for adding a damping resistance.

Addition of damping resistance R_d should adequately damp the resonant behavior of the LCL filter, although it is accompanied with increased power losses. Figure 3.18 shows the bode plots of two cases. It is clear that addition of damping resistance in series with the filter capacitor C_f provides good damping with less losses, compared to a series resistance for the inductor L_m .



(a) Damping resistance in series with filter capacitor C_f . (b) Damping resistance in series with main inductor L_m .

Figure 3.18: Bode plots of LCL filter with passive damping.

In Figure 3.18(a), adding a damping resistance of $R_d = 1 \Omega$ provides good damping with a phase margin > 90°. On the contrary, adding a damping resistance $R_d = 20 \Omega$ in series with L_m provides very less damping with significant increase in power losses.

3.4.2.2 Active Damping

Although passive damping is a simple solution it leads to a decrease in the system efficiency. Other approaches that can be employed for resonance damping include 1) virtual resistance method, 2) addition of a lag-lead compensator, and 3) using a Notch-filter. Virtual resistance method has been extensively use in literature [60-62] as it offers a workaround for resistance damping method. Instead of adding a real resistor, behavior of the resistor is emulated in the controller by using feedback signals and proportional terms. If tuned properly, virtual resistance method offers similar performance like passive damping without additional power losses.



Figure 3.19: Three-phase sFCI with LCL filter, point of common coupling (PCC) and the grid.

3.5 System modeling of grid-connected sFCI equipped with LCL filter

Figure 3.19 shows a grid connected sFCI with an LCL filter. The converter has a common dc-link u_{dc} for all the three phases, employing a single dc-link capacitor C_{in} and separate flying capacitors (C_{FC}) for each phase. The LCL-filter consists of main inductor L_m and grid-side inductor L_g , with internal resistances R_m and R_g respectively. The filter capacitors C_f with parasitic resistance R are connected in a star configuration. The LCL-filter offers attenuation to the harmonics generated by the converter, before it is connected to the grid. It is assumed that the amplitude and phase of the grid voltage remains constant for the model and the direction of current flow is from converter to grid. Modeling of the system is based on the assumption that the converter is fed from an ideal energy source, meaning that the dc-link voltage u_{dc}/u_{FC} is constant. Modeling is carried out in the general (a, b, c)-reference frame.

For modeling we impose the following assumptions :

- Assumption (A.3): The grid has a positive magnitude, a constant angular frequency $\omega_g > 0$, and is symmetric, i.e., $u_g^a(t) + u_g^b(t) + u_g^c(t) = 0$.
- Assumption (A.4): The grid voltage $u_g^{abc}(t) = [u_g^a(t) \ u_g^b(t) \ u_g^c(t)]^{\top}$ is considered as a disturbance input and is available for feedback for $t \ge 0$.
- **Assumption (A.5):** The switching behavior of the converter bridge is neglected i.e. the non-linear behavior of the flying capacitor due to its interaction with the switches is not considered for modeling.

Based on the assumption (A.5) the sFCI converter bridge can be neglected and the system model can be derived for the filter/grid system. Invoking assumption (A.4) and the Kirchoff's current and voltage laws yields the dynamics of the filter/grid system (see Figure 3.14) in the (a, b, c)-reference frame as follows

$$0 = -\boldsymbol{u}_m^{abc}(t) + R_m \boldsymbol{i}_m^{abc}(t) + L_m \frac{\mathrm{d}}{\mathrm{d}t} \boldsymbol{i}_m^{abc}(t) + R(\boldsymbol{i}_m^{abc}(t) - \boldsymbol{i}_g^{abc}(t)) + \boldsymbol{u}_f^{abc}(t)$$
(3.19a)

$$0 = \boldsymbol{i}_m^{abc}(t) - \boldsymbol{i}_g^{abc}(t) - C_f \frac{\mathrm{d}}{\mathrm{d}t} \boldsymbol{u}_f^{abc}(t)$$
(3.19b)

$$0 = R(\boldsymbol{i}_m^{abc}(t) - \boldsymbol{i}_g^{abc}(t)) + \boldsymbol{u}_f^{abc}(t) - \boldsymbol{u}_g^{abc}(t) - R_g \boldsymbol{i}_g^{abc}(t) + -L_g \frac{\mathrm{d}}{\mathrm{d}t} \boldsymbol{i}_g^{abc}(t)$$
(3.19c)

Rearranging the terms to derive a state-space model of the system, we can write:

$$\frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{i}_{m}^{abc}(t) = \frac{1}{L_{m}}\boldsymbol{u}_{m}^{abc}(t) - \frac{R_{m}+R}{L_{m}}\boldsymbol{i}_{m}^{abc}(t) - \frac{1}{L_{m}}\boldsymbol{u}_{f}^{abc}(t) - \frac{R}{L_{m}}\boldsymbol{i}_{g}^{abc}(t)$$
(3.20a)

$$\frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{u}_{f}^{abc}(t) = \frac{1}{C_{f}}\boldsymbol{i}_{m}^{abc}(t) - \frac{1}{C_{f}}\boldsymbol{i}_{g}^{abc}(t)$$
(3.20b)

$$\frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{i}_{g}^{abc}(t) = \frac{R}{L_{g}}\boldsymbol{i}_{m}^{abc}(t) + \frac{1}{L_{g}}\boldsymbol{u}_{f}^{abc}(t) - \frac{R+R_{g}}{L_{g}}\boldsymbol{i}_{g}^{abc}(t) - \frac{1}{L_{g}}\boldsymbol{u}_{g}^{abc}(t)$$
(3.20c)

For the model of the inverter, we denote the state, control input, and disturbance as

$$\begin{aligned} \boldsymbol{x} &:= \begin{bmatrix} \boldsymbol{i}_m^{abc} & \boldsymbol{u}_f^{abc} & \boldsymbol{i}_g^{abc} \end{bmatrix}^\top \\ \boldsymbol{u} &:= \boldsymbol{u}_m^{abc} \\ \boldsymbol{d} &:= \boldsymbol{u}_g^{abc} \end{aligned}$$
 (3.21)

Hence the continuous-time state-space model of the system can be written as

$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}\boldsymbol{x}(t) + \mathbf{G}\boldsymbol{u}(t) + \mathbf{T}\boldsymbol{d}(t)$$
(3.22a)

$$\boldsymbol{y} = \mathbf{C}\boldsymbol{x}(t) \tag{3.22b}$$

where the system matrices are

$$\mathbf{F} = \begin{bmatrix} -\frac{R_m + R}{L_m} & -\frac{1}{L_m} & \frac{R}{L_m} \\ \frac{1}{C_f} & 0 & -\frac{1}{C_f} \\ \frac{R}{L_g} & \frac{1}{L_g} & \frac{R + R_g}{L_m} \end{bmatrix}; \mathbf{G} = \begin{bmatrix} \frac{1}{L_m} \\ 0 \\ 0 \end{bmatrix}; \mathbf{T} = \begin{bmatrix} 0 \\ 0 \\ -\frac{1}{L_g} \end{bmatrix}; \mathbf{C} = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix}.$$
(3.23)

3.6 Summary

In this chapter the design and working of the sFCI was discussed. The operation of the topology was analyzed in detail and preliminaries for system modeling were presented. As can be inferred, the sFCI has a simple construction but its response is not essentially linear, and exhibits a different behavior compared to the conventional inverter topologies due to additional non-linearities. Finally, the non-linear nature of the converter is neglected and behavior of the LCL filter/grid is modeled using mathematical equations.

Chapter 4

State-Feedback Current Control

This chapter proposes the design and digital implementation of a State Feedback Current Controller (SFCC) for the single phase and three phase sFCI converter. The objective of the controller is grid current reference $(i_{g,ref})$ tracking, maintaining the voltage ripple (δ_{FC}) of the flying capacitors below the maximum allowed value, and minimizing the THD of the grid side current. The design of the control structure for single-phase (phase leg) and three phase system are covered separately. The chapter is concluded by simulation results.

4.1 Controller design for a single-phase sFCI

Due to the novel nature of this inverter and lack of previous literature that deals with its control problem, separate approach was followed for the design of control loop for single-phase and three-phase sFCI. As a general notion, a single-phase controller can be extended to the three-phase system, at a later stage, without much effort. Hence, the following sections deal with the controller design for the single-phase sFCI.

A single phase grid-connected sFCI converter is shown in Figure 4.1. The current controller is implemented in the general (a, b, c)-reference frame. In the following sections, the continuous time and discrete time models of the system, are introduced. The losses in the filter components are neglected for the reasons: 1) internal resistances offer additional damping and can be neglected to consider the worst-case scenario for the LCL filter resonance [22]; and 2) the complexity of the discrete-time model and the control algorithm reduces.



Figure 4.1: Single-phase grid-connected sFCI.

- **Assumption (A.6):** The losses in the filter components are neglected i.e. internal resistances R_m, R_g , and R are considered zero, and therefore not shown in Figure 4.1.
- Assumption (A.7): The bridge voltage $u_m(t)$ is assumed to be constant during $kT_s < t < (k+1)T_s$, where T_s is the sampling time [22].

Assumption (A.8): The sampling frequency f_s and the switching frequency f_{sw} are equal.

4.1.1 Continuous-time model

Invoking the assumption (A.5) the dynamics of the flying capacitor are neglected, and hence the system to be controlled reduces to an LCL filter. Based on the Assumption (A.6) the continuous time dynamics of the LCL filter described by the system of equations (3.22), (3.23) can be rewritten for the single phase system as:

$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \underbrace{\begin{bmatrix} 0 & -\frac{1}{L_m} & 0\\ \frac{1}{C_f} & 0 & -\frac{1}{C_f} \\ 0 & \frac{1}{L_g} & 0 \end{bmatrix}}_{\mathbf{F}} \boldsymbol{x}(t) + \underbrace{\begin{bmatrix} \frac{1}{L_m} \\ 0\\ 0 \end{bmatrix}}_{\mathbf{G}} \boldsymbol{u}_m(t) + \underbrace{\begin{bmatrix} 0\\ 0\\ -\frac{1}{L_g} \end{bmatrix}}_{\mathbf{T}} \boldsymbol{u}_g(t)$$

$$y(t) = \underbrace{\begin{bmatrix} 0 & 0 & 1 \end{bmatrix}}_{\mathbf{C}} \boldsymbol{x}(t)$$
(4.1)

where $\boldsymbol{x} = [i_m \ u_f \ i_g]^{\top}$ is the state vector, \boldsymbol{y} is the output i.e., grid current i_g , and input disturbance matrix **D** is assumed to be zero. Essentially the model of (4.1) can be summarized as:

$$\dot{\boldsymbol{x}}(t) = \mathbf{F}\boldsymbol{x}(t) + \mathbf{G}\boldsymbol{u}_m(t) + \mathbf{T}\boldsymbol{u}_g(t)$$
(4.2a)

$$y(t) = \mathbf{C}\boldsymbol{x}(t) \tag{4.2b}$$

The transfer function of the LCL filter from the bridge voltage to the converter current is given as

$$Y(s) = \mathbf{C}(s\mathbf{I} - \mathbf{F})^{-1}\mathbf{G} = \frac{1}{L_m} \frac{s^2 + \omega_z^2}{s(s^2 + \omega_r^2)}$$
(4.3)

where
$$\omega_r = \sqrt{\frac{L_m + L_g}{L_m L_g C_f}}$$
 and $\omega_z = \sqrt{\frac{1}{L_g C_f}}$ (4.4)

are the resonance and anti-resonance frequencies of the LCL filter, respectively.

4.1.2 Discrete-time model

In general, a control algorithm has to be implemented on a DSP or an FPGA. One method is to design the controller in continuous time and then use discrete time equivalent for implementation, see [21,23]. A better way is to discretize the system model and design the controller in discrete time. Therefore, this section discusses the derivation of a Zero-Order-Hold equivalent discrete-time model for the LCL filter/grid system. Based on the assumption (A.8) sampling frequency and switching frequency are equal. Under the assumptions (A.5), (A.6), (A.7), (A.8) and considering that the bridge voltage is averaged over the switching-cycle, the discrete time model of (4.1) becomes

$$\boldsymbol{x}(k+1) = \mathbf{A}\boldsymbol{x}(k) + \mathbf{B}\boldsymbol{u}_m(k) + \mathbf{E}\boldsymbol{u}_g(k)$$
(4.5a)

$$y(k) = \mathbf{C}\boldsymbol{x}(k) \tag{4.5b}$$

where the discretized system matrices are

$$\mathbf{A} = e^{\mathbf{F}T_s} = \begin{bmatrix} \frac{L_m + L_g \cos(\omega_r T_s)}{L_t} & -\frac{\sin(\omega_r T_s)}{\omega_r L_m} & \frac{L_g [1 - \cos(\omega_r T_s)]}{L_t} \\ \frac{\sin(\omega_r T_s)}{\omega_r C_f} & \cos(\omega_r T_s) & -\frac{\sin(\omega_r T_s)}{\omega_r C_f} \\ \frac{L_m [1 - \cos(\omega_r T_s)]}{L_t} & \frac{\sin(\omega_r T_s)}{\omega_r L_g} & \frac{L_g + L_m \cos(\omega_r T_s)}{L_t} \end{bmatrix} \\ \mathbf{B} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{\mathbf{j}\omega_g \tau} d\tau \right) \mathbf{G} = \begin{bmatrix} \frac{T_s}{L_t} + \frac{L_g \sin(\omega_r T_s)}{\omega_r L_m L_t} \\ \frac{L_g [1 - \cos(\omega_r T_s)]}{L_t} \\ \frac{T_s}{L_t} + \frac{\sin(\omega_r T_s)}{\omega_r L_t} \end{bmatrix} \\ \mathbf{E} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{\mathbf{j}\omega_g \tau} d\tau \right) \mathbf{T} = \begin{bmatrix} -\frac{T_s}{L_t} + \frac{\sin(\omega_r T_s)}{\omega_r L_t} \\ \frac{L_m [1 - \cos(\omega_r T_s)]}{L_t} \\ \frac{L_m [1 - \cos(\omega_r T_s)]}{L_t} \end{bmatrix} \\ \mathbf{E} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{\mathbf{j}\omega_g \tau} d\tau \right) \mathbf{T} = \begin{bmatrix} -\frac{T_s}{L_t} + \frac{\sin(\omega_r T_s)}{\omega_r L_t} \\ -\frac{T_s}{L_t} + \frac{L_m \sin(\omega_r T_s)}{\omega_r L_g L_t} \end{bmatrix} \\ \end{bmatrix}$$

Due to the finite computation time, there exists a computational delay between the calculation of the reference voltage $u_{m,ref}$ and its application to the converter bridge u_m [63]. This delay can modeled as $u_m(k) = u_{m,ref}(k-1)$ [22]. Taking the computational delay into account, the discrete-time model is written as

$$\boldsymbol{x}_{d}(k+1) = \underbrace{\begin{bmatrix} \mathbf{A} & \mathbf{B} \\ \mathbf{0} & 0 \end{bmatrix}}_{\mathbf{A}_{d}} \boldsymbol{x}_{d}(k) + \underbrace{\begin{bmatrix} 0 \\ 1 \end{bmatrix}}_{\mathbf{B}_{d}} u_{m,ref}(k) + \underbrace{\begin{bmatrix} \mathbf{E} \\ 0 \end{bmatrix}}_{\mathbf{E}_{d}} u_{g}(k)$$
$$\boldsymbol{y}(k) = \underbrace{\begin{bmatrix} \mathbf{C} & 0 \end{bmatrix}}_{\mathbf{C}_{d}} \boldsymbol{x}_{d}(k)$$
(4.7)

where the modified state vector is $\boldsymbol{x}_d = [\boldsymbol{x}^\top \ u_m]^\top$.

4.1.3 Controllability

For the design of the state-feedback controller, it must be assured that the system (3.22) is controllable. According to the Kalman Controllability criterion (see section 2.3.3), system (3.22) is controllable if

$$\operatorname{rank}\left[\underbrace{\mathbf{G} \quad \mathbf{F}\mathbf{G} \quad \mathbf{F}^{2}\mathbf{G}}_{=:\mathbf{M}_{c}} \right] = 3.$$
(4.8)

Solving for \mathbf{M}_{c} we get

$$\mathbf{M}_{c} = \frac{1}{L_{m}} \begin{bmatrix} 1 & 0 & -\frac{1}{L_{m}C_{f}} \\ 0 & \frac{1}{C_{f}} & 0 \\ 0 & 0 & \frac{1}{C_{f}L_{g}} \end{bmatrix}$$
(4.9)

Rank of $\mathbf{M}_c = 3$ if, and only if, $L_m > 0$, $C_f > 0$, and $L_g > 0$. Since all these parameters are > 0, see Table 3.1, controllability matrix has a full rank and hence the system (3.22) is controllable.

4.2 Design of the current controller

The scheme of the current controller based on state-feedback is shown in Figure 4.2. For improved disturbance rejection the controller is extended using an integral-state, by introducing an integrator into the control loop. The integral state is defined as

$$x_I(k+1) = x_I(k) + i_{g,ref}(k) - i_g(k)$$
(4.10)

where $i_{g,\text{ref}}$ is the grid current reference. The state-space control law becomes

$$u_{m,\text{ref}}(k) = k_I x_I(k) + k_f i_{g,\text{ref}}(k) - \mathbf{K} \mathbf{x}_d(k)$$

$$(4.11)$$

39



Figure 4.2: Closed loop control structure for the single-phase sFCI.

where k_I is the gain of the integral state, k_f is the feedforward gain, $\mathbf{K} = [k_1 \ k_2 \ k_3 \ k_4]$ is the state-feedback gain, and $\mathbf{x}_d = [\mathbf{x}^\top \ u_m]^\top$ is the state matrix augmented with the delayed voltage reference. The state feedback controller with the integral action included, can be designed to have good damping of the LCL filter resonance and disturbance rejection. The feedforward term is an additional degree of freedom for the design of reference-tracking based controllers [22].

4.2.1 Generalized integrators (GI)

Second order generalized integrators (SOGI) are used for obtaining zero steady state error when using stationary reference frames in grid-connected inverters, rectifiers, active filters and power supplies. These integrators have also been used in algorithms for grid synchronization, detection of sequences and harmonic compensation [64]. A common approach for control design of a single-phase grid-connected inverter is to use stationary frame controllers. Since the control variables are sinusoidal and time varying, proportional resonant (PR) controllers based on a proportional gain and SOGI are employed, due to their ability to eliminate steady state error for sinusoidal references. However, in state-feedback based control approaches, SOGIs can be employed to enable grid phase detection and/or variation in the grid parameters and subsequently minimize or eliminate the generated phase delay [65]. The SOGI can also be employed to generate quadrature-signals, which are very useful in single-phase PLLs (phase-locked-loop). Again, implementation of these algorithms is done using microcontrollers, hence design in discrete time is a necessity.

According to [66], the transfer function of a second order generalized integrator for a single sinusoidal signal is

$$G(s) = \frac{2s}{s^2 + \omega_o^2}.$$
 (4.12)

Discrete time implementation consists of discretization of the continuous-time transfer function and calculation of difference equations. For the SOGI better results are obtained using zero-order-hold (ZOH) technique [64]. The following transfer function shows the implementation of the SOGI using ZOH discretization.

$$H_{zoh}(z) = \frac{\sin(\omega_o T_s)(z-1)}{z^2 - 2\cos(\omega_o T_s)z + 1}$$
(4.13)

Figure 4.3 shows the bode plot of (4.13). It shows that the SOGI has a high gain at 50 Hz. Generally,



Figure 4.3: Bode plot of ZOH discretization based SOGI.

the basic functionality of the integrator in the controller is to introduce an infinite gain at the selected frequency, for eliminating steady state error at that particular frequency. A disadvantage of the ZOH based approach is that the difference equations have to be calculated off-line as opposed to the Euler approach. However, the discretized system model (4.1) is also based on the ZOH-equivalent model. Hence it makes sense to use ZOH implementation of the SOGI. Figure 4.4 depicts the structure of the SOGI where the sine and cosine gains are related to the tuning frequency of the SOGI, i.e., 50Hz in our case.



Figure 4.4: SOGI based on ZOH method.

The (discrete time) difference equations for the SOGI can be written as

$$\begin{aligned} x_{gi_1}(k+1) &= \cos(\omega_g T_s) x_{gi_1}(k) - \sin(\omega_g T_s) x_{gi_2}(k) - i_g(k) + i_{g,\text{ref}}(k) \\ x_{qi_2}(k+1) &= \sin(\omega_g T_s) x_{qi_1}(k) + \cos(\omega_g T_s) x_{qi_2}(k) \end{aligned}$$
(4.14)

Hence the generalized integrator model becomes

or

$$\underbrace{\begin{bmatrix} x_{gi_1}(k+1) \\ x_{gi_2}(k+1) \end{bmatrix}}_{\boldsymbol{x}_{gi}(k+1)} = \underbrace{\begin{bmatrix} \cos(\omega_g T_s) & -\sin(\omega_g T_s) \\ \sin(\omega_g T_s) & \cos(\omega_g T_s) \end{bmatrix}}_{\boldsymbol{\Gamma}_{gi}} \underbrace{\begin{bmatrix} x_{gi_1}(k) \\ x_{gi_2}(k) \end{bmatrix}}_{\boldsymbol{x}_{gi}(k)} + (i_{g,ref}(k) - i_g(k))$$

$$(4.15)$$

41

4.2.2 Augmented system for pole placement design

For pole-placement, the system model of (4.7) is augmented with the integral state (4.10) and the generalized integrator (4.15), resulting in

$$\boldsymbol{x}_{a}(k+1) = \mathbf{A}_{a}\boldsymbol{x}_{a}(k) + \mathbf{B}_{a}\boldsymbol{u}_{m,\text{ref}}(k) + \mathbf{E}_{a}\boldsymbol{u}_{g}(k) + \mathbf{P}_{a}\boldsymbol{i}_{g,\text{ref}}(k)$$
(4.16)

where $\boldsymbol{x}_a = [\boldsymbol{x}_d^\top \ x_I \ \boldsymbol{x}_{gi}^\top]^\top$ is the augmented state vector, and

$$\mathbf{A}_{a} = \begin{bmatrix} \mathbf{A}_{d} & \mathbf{0}_{4x1} & \mathbf{0}_{4x2} \\ -\mathbf{C}_{d} & 1 & \mathbf{0}_{1x2} \\ -\mathbf{\Gamma}_{a} & \mathbf{0}_{2x1} & \mathbf{\Gamma}_{gi} \end{bmatrix}_{\mathbb{R}\in7x7}; \mathbf{B}_{a} = \begin{bmatrix} \mathbf{B}_{d} \\ 0 \\ \mathbf{0}_{2x1} \end{bmatrix}_{\mathbb{R}\in7x1}; \mathbf{E}_{a} = \begin{bmatrix} \mathbf{E}_{d} \\ 0 \\ \mathbf{0}_{2x1} \end{bmatrix}_{\mathbb{R}\in7x1}; \mathbf{G}_{a} = \begin{bmatrix} \mathbf{C}_{d} & 0 \\ \mathbf{0}_{1x4} & 0 \end{bmatrix}_{\mathbb{R}\in2x5}; \mathbf{C}_{d} = \begin{bmatrix} \mathbf{C} & 0 \end{bmatrix}_{\mathbb{R}\in1x4}; \mathbf{P}_{a} = \begin{bmatrix} \mathbf{0}_{1x4} & 1 & 1 & 0 \end{bmatrix}_{\mathbb{R}\in7x1}^{\top}$$
(4.17)

are the augmented system matrices. Hence, the current control loop is based on a multi-variable pole placement approach, which combines a state-feedback regulator with an integral block and an SOGI to achieve null tracking error. From (4.10), (4.15) and (4.16) the closed loop dynamics become

$$\boldsymbol{x}_{a}(k+1) = (\mathbf{A}_{a} - \mathbf{B}_{a}\mathbf{K}_{a})\boldsymbol{x}_{a}(k) + (\mathbf{B}_{a}k_{t} + \mathbf{P}_{a})i_{g,ref}(k) + \mathbf{E}_{a}u_{g}(k)$$
$$\boldsymbol{y}(k) = \mathbf{C}_{a}\boldsymbol{x}_{a}(k)$$
(4.18)

The augmented state-feedback gain vector $\mathbf{K}_{a} = [\mathbf{K} - k_{I} \ k_{6} \ k_{7}]$ where the gains k_{6}, k_{7} are related to the GI. Also the output vector $\mathbf{C}_{a} = [0, 0, 1, 0, 0, 0, 0]$. The transfer function from the reference current $i_{g,\text{ref}}(z)$ to the grid side current $i_{g}(z)$ is written as

$$H_{CL}(z) = \frac{\boldsymbol{n}(z)}{\boldsymbol{d}(z)} = \mathbf{C}_{\mathrm{a}}(z\mathbf{I} - \mathbf{A}_{\mathrm{a}} + \mathbf{B}_{\mathrm{a}}\mathbf{K}_{\mathrm{a}})^{-1}(\mathbf{B}_{\mathrm{a}}k_f + \mathbf{P}_{\mathrm{a}}).$$
(4.19)

The denominator polynomial (i.e. the characteristic polynomial) written as

$$d(z) = \det(z\mathbf{I} - \mathbf{A}_{a} + \mathbf{B}_{a}\mathbf{K}_{a})$$
(4.20)

has seven poles, where one pole is originating from the ZOH property of the system, four poles depend on the dynamics of the LCL filter and the remaining two poles depend on the SOGI. The gain vector \mathbf{K}_{a} required to calculate the characteristic polynomial can either be solved using numerical tools like MATLAB or using analytical expressions. The former approach was used in this thesis.

4.2.3 Pole placement

A fully controllable system can be controlled by designing the feedback gain matrix using direct pole placement. Selection of the pole locations depends on a compromise between robustness and dynamic response [22,23]. Figure 4.5(a) shows the poles and zeros of the open-loop transfer function from $u_{m,ref}(z)$ to $i_g(z)$, obtained from (4.7). Poles and zeros are discrete counterparts of those in (4.2). Furthermore, there is a pole at the origin due to the ZOH property. Let the desired characteristic polynomial be

$$d^{*}(z) = z(z - \alpha_{1})(z - \alpha_{2})(z - \alpha_{3})(z - \alpha_{4})(z - \alpha_{5})(z - \alpha_{6}).$$
(4.21)

The poles of the closed-loop system can be arbitrarily placed at desired locations, and the system would respond accordingly. Although, root locus analysis would suggest that the closed-loop poles be placed near the open-loop poles in order to avoid aggressive inputs and/or compromise loop stability. Nevertheless, *near* or *close* is quite ill defined when the number of poles to be placed is significant. It implies, placing n poles explicitly rather than just a nominal area for dominant poles,



Figure 4.5: Pole-zero plots. (a) Poles and zeros of the open-loop transfer function (red). (b) Closed loop poles set to desired positions (blue).

results in an overly specified design [67]. In practice, poles are placed such that the controller has sufficient bandwidth and the closed-loop offers sufficient damping to the LCL filter resonance.

In the desired characteristic polynomial (4.21), the two complex poles $\alpha_{1,2}$ are placed to determine the dominant behavior (i.e., bandwidth of the controller) and two poles $\alpha_{3,4}$ are placed to determine the resonant behavior (i.e., the resonance damping). To simplify the procedure, poles are specified in the continuous-time domain and them mapped to the discrete domain [22]. The dominant and the resonant behavior can be specified by two second order polynomials as

$$\underbrace{(s^2 + 2\zeta_1\omega_1 s + \omega_1^2)}_{\text{Dominant behavior}} \underbrace{(s^2 + 2\zeta_2\omega_2 s + \omega_2^2)}_{\text{Resonant behavior}}.$$
(4.22)

Let us first consider the poles related to dominant behavior. The frequency ω_1 specifies the desired bandwidth, selected as $1/20^{th}$ of the switching frequency. The damping ratio ζ_1 is set to a large value to prevent large overshoots. The corresponding discrete poles are

$$\boldsymbol{\alpha}_{1,2} = \exp\left[\left(-\zeta_1 \pm j\sqrt{1-\zeta_1^2}\right)\omega_1 T_s\right].$$
(4.23)

For damping of the resonant behavior, frequency ω_2 is kept near the natural frequency of the LCL filter ω_r , whereas the damping ratio ζ_2 is chosen to be small. Selecting higher values for ζ_2 is not recommended due to increased control effort [68]. The discrete poles for the resonance damping are

$$\boldsymbol{\alpha}_{3,4} = \exp\left[\left(-\zeta_2 \pm j\sqrt{1-\zeta_2^2}\right)\omega_2 T_s\right].$$
(4.24)

Furthermore, the pole at the origin due to the ZOH discretization, is not moved. The SOGI introduces a complex-conjugate pair of poles corresponding to its tuning frequency $\omega_g = 2\pi \cdot 50$ rad/s. Since the main aim of using an SOGI is to eliminate the phase-shift, it is sufficient to provide a slight damping to its poles (ζ_d), in order to keep them in the vicinity of the open loop poles and avoid aggressive control inputs which might compromise the loop stability. It should be noted that state-feedback control does not affect the location of zeros. The discrete poles for the SOGI are

$$\boldsymbol{\alpha}_{5,6} = \zeta_d \cdot \exp\left[\omega_g T_s\right]. \tag{4.25}$$

Parameter	Value
ζ_1	0.8
ω_1	$2\pi \cdot 1950 \mathrm{rad}\mathrm{s}^{-1}$
ζ_2	0.204
ω_2	$2\pi \cdot 10.155 \mathrm{rad} \mathrm{s}^{-1}$
ζ_d	0.1

Note that in Figure 4.5 there exists a zero at z = -3.14 which is not shown for clarity. Table 4.1 gives an overview of the tuning parameters chosen for pole placement.

 Table 4.1: Tuning parameters for pole placement controller design.

4.2.4 DC voltage feedforward compensation (DVFC)

The input supply of the three-level sFCI is essentially composed of two separate capacitors, i.e., the input dc-link capacitor C_{in} and the flying capacitor C_{FC} , that are independently responsible for supplying power in the positive and negative cycles, respectively. Although the input capacitor C_{in} has a constant voltage across it, the flying capacitor has a voltage ripple equal to δ_{FC} across it, see Figure 4.7(b). This variation in the supply voltage has a prominent effect on the output of the converter bridge, discussed previously in the section 3.2. Moreover, the system model (3.22) was derived by neglecting the switching behavior of the converter bridge and the variation in the supply voltage. Therefore, for tackling this abrupt variation of supply voltage a feedforward approach is proposed, depicted in Figure 4.6.

As a general practice, the output of the state-feedback controller $u_{m,ref}(k)$ [see Figure 4.2], which is fed to the PWM block, is normalized with the input *dc*-link voltage measurement. However, in the present scenario the normalization should be carried out separately for the positive and negative cycles using the voltage measurements across C_{in} and C_{FC} , respectively. In this way the effect of the voltage ripple δ_{FC} and the variation in the supply voltage can be incorporated into the control structure. This approach is termed as DC voltage feedforward compensation (DVFC). DVFC ensures that the controller indirectly maintains the voltage ripple of the flying capacitor below the maximum allowable limit.



Figure 4.6: Illustration of dc voltage feedforward compensation (DVFC).

4.3 Performance evaluation of state-feedback control for singlephase sFCI

To investigate the performance of the proposed state-feedback current control (SFCC) strategy for the sFCI, several simulations using Matlab/Simulink and PLECS blockset have been conducted. The system parameters for the hardware test-bench are given in Table 3.1 and additional parameters used in the simulations are listed in Table 4.2. The system under consideration is a single-phase sFCI rated for 1.67 kW. Since the resonant frequency of the LCL filter is 10.155 kHz, the switching frequency is set to 40 kHz, i.e., almost 4 times the resonant frequency, in order to avoid any excitation of the filter modes and provide sufficient bandwidth to the control loop. Additionally, a level shifted sinusoidal PWM (SPWM) (see Figure 3.4) with a carrier frequency of 40 kHz is employed to generate the switching signals for the four switches.

Parameter	Symbol	Value
Switching frequency	f_{sw}	$40\mathrm{kHz}$
Resonant frequency	ω_r	$2\pi \cdot 10.155 \mathrm{rad} \mathrm{s}^{-1}$
Sampling time	T_s	$25\mu{ m s}$
Dead time	T_d	$300\mathrm{ns}$
Nominal grid voltage	u_g	$230\mathrm{V(rms)}$
Grid Inductance	L_{grid}	$0.01\mathrm{mH}$
Grid Resistance	$\tilde{R_{grid}}$	0.1Ω

 Table 4.2: Additional parameters for simulations.

4.3.1 Steady-state performance

Based on the desired active power of 1.31 kW, the grid current reference (peak) is set to 8 A. Note that the reactive power demand is set to zero. Figure 4.7 shows the steady-state performance of the closed loop system. Figures 4.7(a) and (b) show the voltage and current waveforms of the two dc sources of the sFCI. Unlike the conventional NPC converter, where the two dc-link capacitors are coupled and make voltage balancing a possibility, the two input dc capacitors of the sFCI do not exhibit any coupling between each other, although the flying capacitor is charged using the input capacitor. While the input capacitor C_{in} has a constant voltage across it, the flying capacitor exhibits a voltage ripple $\delta_{FC} \approx 30 - 40$ V. Note that the maximum allowable voltage on the flying capacitor is restricted to 450 V, and therefore the ripple should not exceed this limit. From Figure 4.7(b) we conclude that the proposed SFCC can indirectly control the voltage ripple of the flying capacitor and thus achieves voltage balancing for the flying capacitor.

Figure 4.7(c) shows the output waveform of the converter bridge and compares it to the grid voltage. As can be observed, the controller is able to track the grid voltage properly and generates a voltage pattern that corresponds to the grid voltage in magnitude (average) and phase. From Figure 4.7(d) it is clear that the proposed SFCC achieves reference tracking, with a slight deviation at the zero crossing. This behavior is primarily due to the additional non-linearities, discussed in section 3.2. Moreover, due to the inability of the integral-state to respond quickly to a change in the dc source, i.e., from input capacitor to the flying capacitor, the grid current deviates from the reference current.

The total harmonic distortion (THD) of the grid current in Figure 4.7(d) is 2.02% (dead-time is 0 ns). However, for a dead-time of 300 ns, see Figure 4.8, the THD of the grid current increases to 3.7%, which also conforms with the standard limit of 5% set for the grid-connected PV inverters [69].



(a) Voltage and current waveforms for the input capacitor.(b) Voltage and current waveforms for the flying capacitor.



(c) Converter Bridge voltage and the grid voltage. (d) Comparison of reference current and the grid current.

Figure 4.7: Steady-state performance of the closed loop system using state-feedback control.



Figure 4.8: Comparison of reference current and the grid current when dead-time = 300 ns.

4.3.2 Dynamic performance

Figures 4.9(a)-(c) shows the dynamic response of the closed loop system for an initial current reference of $6 A \approx 0.98 \text{ kW}$ with a step at 0.025 s to 8 A. Figure 4.9(a) shows a comparison between the grid voltage and the grid current. Since the reactive power demand is set to zero, the grid current follows the grid voltage, i.e., it is in-phase with the grid, implying that the controller achieves grid synchronization. The dynamic response of the SFCC is apparent from Figure 4.9(b) which shows a zoomed view of the comparison between the reference current and the grid current. At 0.025 s the current reference is increased from 6 A to 8 A and it is observed that the controller settles within 2 ms after a small overshoot. From Figure 4.9(c) it is observed that the converter output voltage and the flying capacitor have the same behavior as in steady-state, with an increase in the voltage ripple



Figure 4.9: Dynamic performance of the closed loop system.

 δ_{FC} due to increased current reference. Furthermore, Figure 4.9(d) shows the response of the grid current when the reference current is stepped from 0 A to 8 A and then again to 0 A. Compared to conventional control schemes based on PI approach, the pole placement based SFC offers improved dynamic response and less settling time.

4.3.3 Comparison of bode plots

Figure 4.10 presents the frequency response of the open-loop and closed-loop transfer functions for the single-phase grid-connected sFCI. As can be seen, the closed loop system offers a negative peak at the resonant frequency of the LCL filter. Hence, the state-feedback controller provides inherent resonance damping and does not require any additional damping methods.

4.4 Observer design

For hardware implementation an observer is designed, using the Luenberger Observer concept [70], to estimate the unmeasured states u_f and i_g . The observer dynamics, in discrete-time, with estimated state vector $\hat{\boldsymbol{x}} = [\hat{i}_m, \hat{u}_f, \hat{i}_g]^{\top}$ are given by

$$\widehat{\boldsymbol{x}}(k+1) = \mathbf{A}\widehat{\boldsymbol{x}}(k) + \mathbf{B}\widehat{\boldsymbol{u}}_m(k) + \mathbf{E}\widehat{\boldsymbol{u}}_g(k) + \widehat{\boldsymbol{u}}_{obs}$$

$$\widehat{\boldsymbol{y}}(k) = \mathbf{C}\widehat{\boldsymbol{x}}(k)$$
(4.26)

with the observer feedback defined as

$$\widehat{u}_{obs} = \mathbf{L}(y(k) - \widehat{y}(k)) = \mathbf{LC}(\boldsymbol{x}(k) - \widehat{\boldsymbol{x}}(k))$$
(4.27)

47



Figure 4.10: Frequency response of the closed-loop transfer function (4.19) (H_{CL}) from the reference current $i_{g,ref}(z)$ to the grid side current $i_g(z)$. The open-loop transfer function is obtained from (4.5) as $H_{OL} = \mathbf{C}(z\mathbf{I} - \mathbf{A})^{-1}\mathbf{B}$.

where **L** is the observer gain vector, and has to be chosen such that the eigenvalues of $[\mathbf{A} - \mathbf{LC}]$ lie within the unit circle. Sensitivity to measurement noise and dynamic response of the observer play a vital role while selecting the pole locations. A rule of thumb to select the observer poles is to set them at least twice as fast as the controller dynamics [22,68]. The characteristic equation for the estimation-error dynamics is

$$\boldsymbol{\alpha}_o(z) = \det(z\mathbf{I} - \mathbf{A} + \mathbf{L}\mathbf{C}) \tag{4.28}$$

The poles of $\alpha_o(z)$ can be mapped via continuous-time placement to independently set the dominant dynamics and the complex-conjugate parts. However, since the system is completely observable, pole locations can be chosen arbitrarily inside the unit circle. The latter approach was preferred in this thesis and the observer poles were placed in the vicinity of the origin, as per the dead-beat approach. Due to its inherent filtering property, this approach allows the observer to have higher dynamic performance.

Figure 4.11 shows the response of the observer where (a) depicts the original state measurements and (b) depicts the estimated states from the observer. It is apparent that the estimated states from the observer correspond to the actual state measurements.



Figure 4.11: Steady-state performance of the observer.

4.5 Controller design for a three-phase sFCI

4.5.1 Motivation

The design of state-feedback controller for the single-phase sFCI was discussed in the previous sections. Generally, a controller designed in the (a, b, c)-reference frame for a single-phase inverter can be extended to a three-phase inverter by creating separate controllers for each phase. In this manner no extra transformations are required, and the control design is simple.

Figure 4.12 shows the response of the three-phase sFCI for a controller designed in (a, b, c)-reference frame, as indicated above. For the few initial cycles, the controller synchronizes with the grid and achieves reference tracking for the grid current. However, balancing of the voltage ripple on the flying capacitors only works for one of the phases while for the other two phases the voltage on the flying capacitors increases beyond the allowed limit, i.e., 450 V (termed as voltage fly-away) after a few cycles of operation, see Figure 4.12(b). Hence, a controller designed in (a, b, c)-reference frame does not fulfill all the control objectives.

Furthermore, design of the controller in the stationary (α, β) -reference frame and the rotating (d, q)-reference frame was investigated. The stationary (α, β) -reference frame controller gives a response similar to the (a, b, c)-reference frame controller, with the flying capacitor voltage exceeding the allowed limit (450 V) in phases b and c. However, the (d, q)-reference frame controller shows promising performance and is able to achieve both reference tracking and flying capacitor voltage control. Hence, it can be concluded that the three phases of the sFCI possess some inherent coupling due to the presence of flying capacitors.

In the following sections, design of the controller for three-phase sFCI in the rotating (d, q)-reference frame is discussed.



Figure 4.12: Closed loop results for three phase sFCI using controller designed in (a, b, c)-reference frame.

4.5.2 Mathematical model of the converter

The rotating (d, q)-reference frame is aligned with the grid voltage¹ (voltage orientation), i.e.,

$$\boldsymbol{u}_{g}^{k}(t) = u_{g}^{d}(t) + ju_{g}^{q}(t) = \hat{u}_{g}(t) + j0.$$
(4.29)

Using the Clarke transformation (3.5) and Park transformation (3.8) in (3.22) yields the system model in the (d, q)-reference frame. The continuous time model in the (d, q) frame can be written as

$$\frac{\mathrm{d}\boldsymbol{x}^{dq}(t)}{\mathrm{d}t} = \mathbf{F}\boldsymbol{x}^{dq}(t) + \mathbf{G}\boldsymbol{u}_m^{dq}(t) + \mathbf{T}\boldsymbol{u}_g^{dq}(t)$$
(4.30a)

$$\boldsymbol{y}(t) = \mathbf{C}\boldsymbol{x}^{dq}(t) \tag{4.30b}$$

where the state vector is $x^{dq} = [i_m^d \ i_m^q \ u_f^d \ u_f^q \ i_g^d \ i_g^q]^{\top}$ and the system matrices are

$$\mathbf{F} = \begin{bmatrix} -\frac{R+R_m}{L_m} & \omega_g & -\frac{1}{L_m} & 0 & \frac{R}{L_m} & 0 \\ -\omega_g & -\frac{R+R_m}{L_m} & 0 & -\frac{1}{L_m} & 0 & \frac{R}{L_m} \\ \frac{1}{C_f} & 0 & 0 & \omega_g & -\frac{1}{C_f} & 0 \\ 0 & \frac{1}{C_f} & -\omega_g & 0 & 0 & -\frac{1}{C_f} \\ \frac{R}{L_g} & 0 & \frac{1}{L_g} & 0 & -\frac{R+R_g}{L_g} & \omega_g \\ 0 & \frac{R}{L_g} & 0 & \frac{1}{L_g} & -\omega_g & -\frac{R+R_g}{L_g} \end{bmatrix}; \mathbf{G} = \begin{bmatrix} \frac{1}{L_f} & 0 \\ 0 & \frac{1}{L_f} \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \end{bmatrix}; \mathbf{T} = \begin{bmatrix} 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ -\frac{1}{L_g} & 0 \\ 0 & -\frac{1}{L_g} \end{bmatrix}.$$

$$(4.31)$$

Since the q-axis is 90° ahead of the d-axis, i.e., $\xi^q = \mathbf{j} \cdot \xi^d$ or $\xi^d = -\mathbf{j} \cdot \xi^q$, complex valued space vectors (e.g., the main inductor current $\mathbf{i}_m = \mathbf{i}_m^d + \mathbf{j} \, \mathbf{i}_m^q$) can be used to represent the system (4.30), (4.31) as follows

$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \underbrace{\begin{bmatrix} -\frac{R+R_m}{L_m} - \mathrm{j}\omega_g & -\frac{1}{L_m} & \frac{R}{L_m} \\ \frac{1}{C_f} & -\mathrm{j}\omega_g & -\frac{1}{C_f} \\ \frac{R}{L_g} & \frac{1}{L_g} & -\frac{R+R_g}{L_g} - \mathrm{j}\omega_g \end{bmatrix}}_{\mathbf{F}} \boldsymbol{x}(t) + \underbrace{\begin{bmatrix} 1\\L_m \\ 0 \\ 0 \end{bmatrix}}_{\mathbf{G}} \boldsymbol{u}_m(t) + \underbrace{\begin{bmatrix} 0\\ 0 \\ -\frac{1}{L_g} \end{bmatrix}}_{\mathbf{T}} \boldsymbol{u}_g(t) \quad (4.32a)$$

$$\boldsymbol{y}(t) = \underbrace{\begin{bmatrix} 0 & 0 & 1 \end{bmatrix}}_{\mathbf{C}} \boldsymbol{x}(t) \quad (4.32b)$$

The input matrices remain unchanged as there are no coupling terms. Since the imaginary part signifies the coupling between respective components, a complex-valued gain vector $\mathbf{K} \in \mathbb{R}^n$ can be back-transformed to the (d, q) frame using:

$$\begin{bmatrix} \mathbf{K}^{d} \\ \mathbf{K}^{q} \end{bmatrix} = \begin{bmatrix} \mathfrak{Re}(K_{0}) & \mathfrak{Im}(K_{0}) & \mathfrak{Re}(K_{1}) & \mathfrak{Im}(K_{1}) & \dots & \mathfrak{Re}(K_{n-1}) & \mathfrak{Im}(K_{n-1}) \\ -\mathfrak{Im}(K_{0}) & \mathfrak{Re}(K_{0}) & -\mathfrak{Im}(K_{1}) & \mathfrak{Re}(K_{1}) & \dots & -\mathfrak{Im}(K_{n-1}) & \mathfrak{Re}(K_{n-1}) \end{bmatrix}$$
(4.33)

4.5.3 Grid synchronization in power converters

Grid synchronization implies, the accurate detection of the phase angle and other attributes of the grid voltage, in order to tune an internal oscillator of the power converter to the oscillatory dynamics imposed by the grid. Usually the main attributes to enable grid connection, via power converters, are the phase angle and the amplitude of the fundamental frequency component of the grid voltage [71]. Grid synchronization techniques can be classified into two groups, namely the time-domain and the frequency-domain methods.

The time-domain detection methods are based on some kind of adaptive loop that tracks the specific component of input signal. The most commonly employed synchronization method is based

¹Note: $\boldsymbol{u}_{g}^{k}(t) = \mathbf{T}_{\mathbf{p}}(\phi_{g})^{-1} \mathbf{T}_{\mathbf{c}} \boldsymbol{u}_{g}^{abc}(t)$, where the angle $\phi_{g}(t) = \omega_{g} t$.



Figure 4.13: (a) Phase-locked-loop (PLL) for grid synchronization, and (b) block diagram of a PLL.

on a phase-locked-loop (PLL). A PLL is a closed loop system in which an oscillator is used to track the fundamental component of the grid voltage and calculate its phase-angle and amplitude. The structure of a PLL used for grid synchronization is shown in Figure 4.13(a). The aim of the PLL is to drive the quadrature q-component of the grid voltage to zero i.e., $u_g = u_g^d + j0$. From Figure 4.13(a) the q-component of the grid voltage can be written as

$$u_g^q = -u_g^\alpha \sin(\hat{\phi}_g) + u_g^\beta \cos(\hat{\phi}_g)$$

= $-u_g \cos(\phi_g) \sin(\hat{\phi}_g) + u_g \sin(\phi_g) \cos(\hat{\phi}_g)$
= $u_g \sin(\phi_g - \hat{\phi}_g) \approx u_g(\phi_g - \hat{\phi}_g)$ (4.34)

where u_g^{α} and u_g^{β} are the components of the grid voltage u_g^s in stationary reference frame, and $\hat{\phi}_g$ is the angle estimated by the PLL. Similarly, the *d*-component of the grid voltage can be written as

$$u_g^d = u_g^\alpha \cos(\hat{\phi}_g) - u_g^\beta \sin(\hat{\phi}_g)$$

= $u_g \cos(\phi_g) \cos(\hat{\phi}_g) - u_g \sin(\phi_g) \sin(\hat{\phi}_g)$
= $u_g \cos(\phi_g - \hat{\phi}_g)$ (4.35)

If the estimated angle of the PLL $(\hat{\phi}_g)$ and the grid phase-angle (ϕ_g) are equal, i.e., $\hat{\phi}_g = \phi_g$, the q-component u_g^q will become negligible, and the d-component u_g^d will be equal to the amplitude of the grid voltage \boldsymbol{u}_g^s . A linearized block diagram of the PLL, considering (4.34), is shown in Fig. 4.13(b). The transfer function from (ϕ_g) to $(\hat{\phi}_g)$ is written as

$$\boldsymbol{H}_{PLL}(s) = \frac{u_g(k_p s + k_i)}{s^2 + u_g k_p s + u_g k_i}$$
(4.36)

A comparison between the coefficients of the denominator polynomial of $H_{PLL}(s)$ and a second-order polynomial $s^2 + 2\zeta_n\omega_n s + \omega_n^2$ yields the PI controller gains k_p and k_i of the PLL as

$$k_p = \frac{2\zeta_n \omega_n}{u_g}; \qquad k_i = \frac{\omega_n^2}{u_g}. \tag{4.37}$$

where k_p is the proportional gain, k_i is the integral gain, ω_n is the undamped natural frequency, and ζ_n is the damping factor. Figure 4.14 shows the output of the PLL block w.r.t the voltage of the phase-*a*.



Figure 4.14: Output of the phase-locked-loop (PLL).

4.5.4 Controller design

Invoking the assumptions (A.4), (A.5) and (A.6), model of the LCL filter/grid with complex valued terms (4.32) is re-written as

$$\frac{d\boldsymbol{x}(t)}{dt} = \underbrace{\begin{bmatrix} -j\omega_g & -\frac{1}{L_m} & 0\\ \frac{1}{C_f} & -j\omega_g & -\frac{1}{C_f}\\ 0 & \frac{1}{L_g} & -j\omega_g \end{bmatrix}}_{\mathbf{F}} \boldsymbol{x}(t) + \underbrace{\begin{bmatrix} \frac{1}{L_m}\\ 0\\ 0 \end{bmatrix}}_{\mathbf{G}} \boldsymbol{u}_m(t) + \underbrace{\begin{bmatrix} 0\\ 0\\ -\frac{1}{L_g} \end{bmatrix}}_{\mathbf{T}} \boldsymbol{u}_g(t)$$
(4.38a)
$$\boldsymbol{y}(t) = \underbrace{\begin{bmatrix} 0 & 0 & 1 \end{bmatrix}}_{\mathbf{G}} \boldsymbol{x}(t)$$
(4.38b)

The current control structure for the three-phase sFCI is shown in Figure 4.15. The model (4.38) is discretized and extended with an integral state, following the steps described in the sections 4.1 and 4.2. For pole placement, the augmented system matrix for the three-phase sFCI has five poles that are to be placed to set the dominant and resonant dynamics of the current controller. The pole placement procedure is similar to the method described in section 4.2.3. However, the calculated gain vector **K** is complex valued and must be back-transformed to the (d, q)-frame using (4.33). Note that the grid synchronization is carried using the PLL output, therefore poles related to the SOGI are not used in the three-phase approach.



Figure 4.15: Closed loop control structure for three-phase sFCI.

4.6 Performance evaluation of state-feedback control for threephase sFCI

To investigate the performance of the proposed state-feedback current control (SFCC) scheme for the three-phase sFCI, several simulations using Matlab/Simulink and PLECS blockset have been conducted. The system parameters for the hardware test-bench are given in Table 3.1 and Table 4.2. The system under consideration is a three-phase sFCI rated for 5 kW. Please note that a dead-time of 300 ns is valid for all the simulations in this section.



Figure 4.16: Steady-state simulation results for three-phase sFCI using controller designed in (d, q)-reference frame.

4.6.1 Steady-state performance

Based on the desired active power of $3.9 \,\mathrm{kW}$, the grid current reference is set to 8 A for each phase. The reactive power demand is set to zero. Figures 4.16 and 4.17 shows the steady-state performance of the three-phase sFCI. From Figure 4.16, which shows the voltage and current waveforms for the input *dc*-link and the three flying capacitors, we can infer that the controller achieves voltage control of all the three flying capacitors. Figure 4.17(a) shows the output of the converter phase-*a* and the grid voltage of the three phases. It also shows the output angle of the PLL as it tracks the phase-*a* and makes grid synchronization possible. Figure 4.17(b) shows the waveforms of the three states of the LCL filter. Compared to the main inductor current, the grid current has negligible current-ripple and thus lower harmonic content, implying that the LCL filter offers excellent attenuation of the switching frequency harmonics. Figure 4.17(c), which shows the steady-state performance of the *d* and *q* axis currents, highlights the reference tracking of the SFCC.



(a) Voltage waveforms and grid angle from the PLL.





(c) Steady-state performance of i_d and i_q w.r.t. their reference values.

Figure 4.17: Steady-state performance of the three-phase sFCI using state-feedback control.



Figure 4.18: Dynamic performance of SFC: comparison of grid current and grid voltage of the phase-*a*.

4.6.2 Dynamic performance

Figures 4.18, 4.19 and 4.20 illustrate the dynamic performance of the SFCC. At 0.01 s the desired active power is set to $3.9 \,\mathrm{kW} \approx i_d = 8 \,\mathrm{A}$ (per phase), and at 0.04 s the reactive power demand is set



(a) Dynamic performance of i_d and i_q w.r.t. their reference (b) Dynamic performance of the controller (zoomed values. view).

Figure 4.19: Closed loop simulation results for three-phase sFCI.

to $1.5 \text{ kVAr} \approx i_q = 3 \text{ A}$ (per phase). From Figure 4.18 it is observed that the grid current follows the grid voltage when there is no reactive power demand, and has a phase-shift w.r.t the grid voltage when the reactive power demand is set to 1.5 kVAr. From Figures 4.19(a) and (b) it is concluded that the SFCC offers excellent dynamic performance and has a settling time of less than 10 ms.

Figure 4.20, which presents a comparison between the actual grid current and its reference values, signifies the reference tracking of the proposed state-feedback current controller. In Figure 4.20, the sharp peaks in the grid current response arise due to two reasons: 1) due to the introduction of dead-time into the PWM block, and 2) due to the inherent coupling between the three flying capacitors. From this figure we can also observe that whenever any one of the phases undergoes transition from positive to negative cycle, the deviation in grid current affects all the other phases, thereby confirming that there exists some inherent coupling. This behavior requires a deeper study of this topology and is out of the scope of this thesis. Nevertheless, the grid current with all these distortions is observed to have an average THD of 2.25 %.



Figure 4.20: Dynamic performance of the (d, q)-reference frame controller: comparison of grid currents and their references.

4.7 Summary

This chapter proposed a state-feedback based current control scheme for the single-phase and three-phase sFCI, connected to the grid via an LCL filter. To achieve an improved dynamic and steady-state performance, a pole-placement based control design approach was discussed. The simulation results show that higher dynamic performance and better resonance damping can be achieved with the proposed controller in comparison with conventional linear control techniques. The results also show that the resonance of the LCL filter is well damped, and the dynamic performance specified by direct pole placement is obtained for the reference tracking, grid voltage disturbance rejection, and balancing of flying capacitor voltages.

Chapter 5

Model Predictive Control using a Linearized Converter Model

This chapter begins with the motivation for using model predictive control (MPC) to control a threephase sFCI. The first part of this chapter presents a finite-control-set MPC scheme with a prediction horizon of three. The second part of this chapter discusses the drawbacks of finite-control-set MPC and presents a model predictive control scheme using a continuous-control-set, where a modulator is required for generation of firing signals.

5.1 Motivation

Considering the additional non-linearities present in the sFCI topology, particularly the deviation of the grid current from its reference when the inverter operation changes from the positive cycle to the negative cycle, see Figure 5.1, it can be reasoned that a linear control technique like state-feedback control does not provide sufficient dynamic performance. Figure 5.1(b) shows a zoomed view of the grid current at the zero crossing where the inverter operation changes scheme. At this moment,



Figure 5.1: Comparison of the grid current and its reference to highlight the deviation at zero crossing. (a) comparison for a single cycle, (b) zoomed view around zero crossing when the inverter supply changes from input dc-link to flying capacitor.

the flying capacitor starts discharging (negative state) and supplies power to the load, instead of the input dc-link. This changeover of input sources was tackled using a feed-forward approach known as DVFC, see section 4.2. However, the state-feedback controller is unable to react abruptly to this behavior and hence the output current deviates from its reference. From Figure 5.1 it is



Figure 5.2: Three-phase Siwakoti-H flying capacitor inverter connected to the grid using LCL filter.

observed that the initial overshoot is followed by an oscillatory response due to the SOGI and the later undershoot is when the integral state reacts to this deviation and tries to minimize the error.

Unlike state-feedback control, predictive control techniques offer faster dynamic response, lower settling time and a provision to handle various non-linearities including system constraints. In the following sections a model predictive control (MPC) based scheme is designed to simultaneously control the main-inductor current, the filter capacitor voltage and the grid-side current by means of reference tracking.

5.2 Finite-control-set model predictive control (FCS-MPC)

In this section a FCS-MPC based strategy is introduced to control the sFCI. FCS-MPC departs from the concept of using a modulator as switching signals are generated directly from the controller stage. The objectives of the controller are to minimize tracking error and maintain the ripple voltage of the flying capacitor within the allowed limits. To achieve these control goals, a model of the system is derived that accurately predicts its behavior over the whole operating regime.

5.2.1 Continuous-time mathematical model

Figure 5.2 shows the configuration of a three-phase sFCI connected to the grid using an LCL filter. The LCL filter with main inductor L_m , grid-side inductor L_g and filter capacitor C_f provides sufficient attenuation to the switching frequency harmonics present in the main inductor current i_m . To simplify the modeling and ease the control design, the variables are expressed in the stationary (α, β) -reference frame instead of the three-phase system (a, b, c). A variable $\boldsymbol{\xi}^{abc} := [\boldsymbol{\xi}_a \ \boldsymbol{\xi}_b \ \boldsymbol{\xi}_c]^{\top}$ can be transformed to a variable $\boldsymbol{\xi}^{\alpha\beta} := [\boldsymbol{\xi}^{\alpha} \ \boldsymbol{\xi}^{\beta}]^{\top}$ in the (α, β) system through $\boldsymbol{\xi}^{\alpha\beta} = \mathbf{T}_c \ \boldsymbol{\xi}^{abc}$, where \mathbf{T}_c is the Clarke transformation matrix defined in section 3.3.

The system states include the main inductor currents, the grid currents and filter capacitor voltages. Thus, the state vector is $\boldsymbol{x} = [i_m^{\alpha} \ i_m^{\beta} \ u_f^{\alpha} \ u_f^{\beta} \ i_g^{\alpha} \ i_g^{\beta}]^{\top} \in \mathbb{R}^6$. The three-phase switch position $\boldsymbol{s}^{abc} \in S^3$ is considered as the input to the system, with $\boldsymbol{s}^{abc} = [s_a \ s_b \ s_c]^{\top}$ and $S^3 = \{-1, 0, 1\}$. Moreover, the grid voltage is considered as a disturbance to the system. A continuous-time model of the three-phase LCL filter/grid system, in the rotating (d, q)-reference frame was derived in section 4.5. Neglecting the coupling terms between the various axes transforms the model to the (α, β) system. The system model in the (α, β) -frame is thus written as

$$\frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{x}^{\alpha\beta}(t) = \mathbf{F}\boldsymbol{x}^{\alpha\beta}(t) + \mathbf{G}\boldsymbol{u}_m^{\alpha\beta}(t) + \mathbf{T}\boldsymbol{u}_g^{\alpha\beta}(t)$$
(5.1a)

$$\boldsymbol{y} = \mathbf{C}\boldsymbol{x}^{\alpha\beta}(t) \tag{5.1b}$$
where the system matrices are

$$\mathbf{F} = \begin{bmatrix} -\frac{R+R_m}{L_m} & 0 & -\frac{1}{L_m} & 0 & \frac{R}{L_m} & 0 \\ 0 & -\frac{R+R_m}{L_m} & 0 & -\frac{1}{L_m} & 0 & \frac{R}{L_m} \\ \frac{1}{C_f} & 0 & 0 & 0 & -\frac{1}{C_f} & 0 \\ 0 & \frac{1}{C_f} & 0 & 0 & 0 & -\frac{1}{C_f} \\ \frac{R}{L_g} & 0 & \frac{1}{L_g} & 0 & -\frac{R+R_g}{L_g} & 0 \\ 0 & \frac{R}{L_g} & 0 & \frac{1}{L_g} & 0 & -\frac{R+R_g}{L_g} \end{bmatrix}, \quad \mathbf{G} = \begin{bmatrix} \frac{1}{L_f} & 0 \\ 0 & \frac{1}{L_f} \\ 0 & 0 \\ 0 & 0 \\ 0 & 0 \end{bmatrix}, \quad \mathbf{T} = \begin{bmatrix} 0 & 0 \\ 0 & 0 \\ 0 & 0 \\ -\frac{1}{L_g} & 0 \\ 0 & -\frac{1}{L_g} \end{bmatrix},$$
$$\mathbf{C} = \mathbf{I}_{6\times 6}, \text{ and } \boldsymbol{\xi}^{\alpha\beta} = \mathbf{T}_c \, \boldsymbol{\xi}^{abc}.$$
(5.2)

In the mathematical model, all the states and inputs for the three-phase sFCI are transformed to the stationary frame. The output matrix \mathbf{C} is an identity matrix of $\mathbb{R}^{6\times 6}$, as all the three states of the LCL filter will be controlled simultaneously by the predictive controller. Control of all the three states improves the transient response and provides stable control even with short horizons.

5.2.2 Converter bridge voltage: Space vectors

For a three-level inverter, each phase can have three discrete states, i.e., $S = \{-1, 0, 1\} \approx \{N, 0, P\}$. Based on the switch positions of the three phases¹ s_{abc} , the converter generates a set of discrete voltages at the output of the bridge. These can be expressed using space vectors, as defined in section 3.3. Considering the negative port of the dc-link as the reference, the space vector of the converter bridge voltage is defined as

$$\boldsymbol{u}_{m}^{abc} = \frac{2}{3}(u_{a} + e^{j2\pi/3}u_{b} + e^{j4\pi/3}u_{c})$$
(5.3)

where u_a , u_b and u_c are the phase-to-neutral voltages of the three phases. The phase-to-neutral voltages are defined in terms of the input supply voltage u_{in} and the switch positions $s_{abc} = [s_a \ s_b \ s_c]^{\top}$ as

$$u_a = s_a u_{in}$$

$$u_b = s_b u_{in}$$

$$u_c = s_c u_{in}$$
(5.4)

where $u_{in} = S^+ u_{dc} + (1 - S^+) u_{FC}$, and S^+ is a binary variable which is true for positive half cycle, i.e.,

$$S^{+} = \begin{cases} 1 & \text{for positive half cycle} \\ 0 & \text{for negative half cycle.} \end{cases}$$
(5.5)

 S^+ is used to distinguish between the two half-cycles of operation, as input dc-link supplies during the positive cycle and flying capacitor supplies during the negative cycle.

Using Clarke transformation to convert the space vectors to (α, β) -reference frame gives a total of $3^3 = 27$ voltage vectors (see Figure 2.2(b)). Neglecting the ripple on the flying capacitor and assuming $u_{in} \approx u_{dc} \approx u_{FC}$, the voltage vectors generated by the three-phase sFCI can be represented in the (α, β) -reference frame as follows

$$\boldsymbol{u}_{m}^{\alpha\beta} = \begin{cases} \frac{4}{3} \cdot u_{dc} \cdot e^{j((n-21)\pi/3)} & \text{for } n = \{21, 22, \dots, 26\} & 6 \text{ long vectors} \\ \frac{2}{\sqrt{3}} \cdot u_{dc} \cdot e^{j((n-15)\pi/3 + \pi/6)} & \text{for } n = \{15, 16, \dots, 20\} & 6 \text{ intermediate vectors} \\ \frac{2}{3} \cdot u_{dc} \cdot e^{j((n-3)\pi/3)} & \text{for } n = \{3, 4, \dots, 14\} & 12 \text{ short vectors} \\ 0 & \text{for } n = \{0, 1, 2\} & \text{zero vectors} \end{cases}$$
(5.6)

¹Note: A switching combination $s_{abc} = \{PN0\}$ denotes that the positive, negative and zero states are active in the phases a, b and c, respectively. For details of operating states see section 3.1.

Out of these 27 voltage vectors, a set of 19 vectors is active and the rest of the vectors are redundant. The active vectors in combination with the zero vectors can be used to generate the required bridge voltage.

5.2.3 Discrete-time model

Generally, a continuous-time state space model can be discretized using the Euler method or Tustin's equivalent. The Forward Euler method is adequately precise when the sampling interval is of the order of tens of microseconds. However, the performance of the controller degrades if the sampling frequency is decreased. Therefore, exact discretization is required to improve the controller performance.

Assumption (A.9): The filter parameters $L_m, R_m, L_g, R_g, C_f, R$ are considered to be constant between two successive sampling instants.

Considering the converter bridge output $u_m(t)$ to be piecewise constant during each sampling instant and invoking assumptions (A.4), (A.5) and (A.9), the continuous-time model (5.1), (5.2) can be discretized using zero-order-hold method. The exact discrete-time model is written as

$$\boldsymbol{x}^{\alpha\beta}(k+1) = \mathbf{A}\boldsymbol{x}^{\alpha\beta}(k) + \mathbf{B}\boldsymbol{u}_m^{\alpha\beta}(k) + \mathbf{E}\boldsymbol{u}_q^{\alpha\beta}(k)$$
(5.7a)

$$\boldsymbol{y}(k) = \mathbf{C}\boldsymbol{x}^{\alpha\beta}(k) \tag{5.7b}$$

where k is the discrete-time index, T_s is the sampling interval, and the discretized system matrices are

$$\mathbf{A} = e^{\mathbf{F}T_s}, \ \mathbf{B} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{j\omega_g\tau} d\tau\right) \mathbf{G}, \ \mathbf{E} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{j\omega_g\tau} d\tau\right) \mathbf{T}.$$
 (5.8)

5.2.4 Control objectives

For the sFCI, the control objective is twofold. First, the grid current i_g should accurately track the reference current $i_{g,ref}$. In addition, the main inductor current i_m and the filter capacitor voltage u_f should be regulated along their reference trajectories $i_{m,ref}$ and $u_{f,ref}$, derived from a separate reference calculation block. Moreover, the switching losses are to be kept relatively low, which can be achieved indirectly by controlling the switching frequency. Finally, during transients, the above-mentioned controlled variables should quickly reach their desired values and with as little overshoot as possible.

5.2.5 Controller block diagram

The scheme of the proposed predictive controller with grid current reference tracking is illustrated in Figure 5.3. As can be seen, the desired system performance is achieved by directly manipulating the inverter switches, without the presence of a modulator. The proposed MPC algorithm first computes the evolution of the plant over the prediction horizon (i.e. the trajectories of the variables of concern) based on the measurements of the grid current, main inductor current, filter capacitor voltage, and grid voltage. Following this, the optimal control action (i.e. the switching signals) is chosen by minimizing a performance criterion in real time.

5.2.6 Control problem

The discrete-time model (5.7), (5.8) is used to predict the output $\mathbf{y}(t)$ of the system. At time-step k, the cost function that penalizes the error of the output variables and the switching effort over the finite prediction horizon of N_p time steps is written as

$$J(k) = \sum_{l=k}^{k+N_p-1} \|\boldsymbol{y}_{\text{ref}}(l+1) - \boldsymbol{y}(l+1)\|_{\boldsymbol{Q}}^2 + \|\Delta \boldsymbol{s}_{abc}(l)\|_{\boldsymbol{R}}^2.$$
(5.9)



Figure 5.3: Finite-control-set model predictive control with reference tracking for the three-phase sFCI.

In (5.9) $\boldsymbol{y}_{ref} \in \mathbb{R}^6$ is a vector encompassing the reference values of the controlled variables (main inductor current, filter capacitor voltage and the grid current), i.e.

$$\boldsymbol{y}_{\text{ref}} = \begin{bmatrix} i_{m,\text{ref}}^{\alpha} & i_{f,\text{ref}}^{\beta} & u_{f,\text{ref}}^{\alpha} & i_{g,\text{ref}}^{\beta} & i_{g,\text{ref}}^{\beta} \end{bmatrix}^{\top}.$$
(5.10)

Moreover, the term $\Delta s_{abc}(k) = s_{abc}(k) - s_{abc}(k-1)$ is added to control the inverter switching frequency by penalizing the switching transitions. The diagonal matrices Q and $R \in \mathbb{R}^{6\times 6}$ are the weighting matrices² that set the trade-off between the overall tracking accuracy and the switching frequency. Note that Q and R are positive semidefinite, and the diagonal entries of Q are chosen in a way that the tracking accuracy among the three controlled variables is prioritized. More specifically, priority is given to the grid current by penalizing the corresponding error more heavily. This is achieved by choosing larger values for the corresponding diagonal entries in Q. It implies that when more weight is put into the tracking of the grid current reference $i_{g,ref}$, the trade-off between the tracking accuracy and the switching frequency is simplified to the trade-off between the grid current THD and the switching frequency of the converter. The switching frequency can be calculated by averaging the total number of *on* transitions, for each phase of the sFCI, over a time interval (MT_s) as

$$f_{sw,avg} = \lim_{M \to \infty} \frac{1}{MT_s} \cdot \frac{1}{12} \sum_{l=0}^{M-1} \|\boldsymbol{s}_{abc}(l) - \boldsymbol{s}_{abc}(l-1)\|.$$
(5.11)

Here, the *average switching frequency* is obtained by dividing the total number of transitions by 12, i.e., the total number of the controllable switches of the three-phase converter.

The optimal sequence of control actions is then computed by minimizing the cost function (5.9) over the optimization variable, i.e., the switching sequences

$$\mathbf{S}(k) = \left[\mathbf{s}_{abc}^{\top}(k), \ \mathbf{s}_{abc}^{\top}(k+1), \dots, \ \mathbf{s}_{abc}^{\top}(k+N_p-1)\right]^{\top}$$

²The squared norm weighted with the positive (semi)definite matrix \boldsymbol{W} is given by $\|\boldsymbol{\xi}\|_{\boldsymbol{W}}^2 = \boldsymbol{\xi}^\top \boldsymbol{W} \boldsymbol{\xi}$

for a prediction horizon N_p . The optimization problem can be summarized as

$$\mathbf{S}_{opt}(k) = \arg \min_{\mathbf{S}(k)} J(k)$$
(5.12a)

subject to eq.
$$(5.7)$$
 $(5.12b)$

$$\mathbf{S}(k) \in \mathbb{S} \tag{5.12c}$$

where $S = S^{3N_p}$. The output of the optimization problem is a set of optimal switching sequences $S_{opt}(k)$, out of which only the first element $s_{abc,opt}(k)$ is applied to the sFCI, whereas the rest of the elements are discarded. At the next time step k + 1, the complete procedure is repeated for the updated measurements over a one-step shifted horizon, according to the receding horizon policy [27]. Figure 5.4 presents a typical flowchart of the FCS-MPC approach.



Figure 5.4: Flowchart of the FCS-MPC strategy applied to a three-phase sFCI.

5.2.7 Reference generation

The proposed FCS-MPC controller is based on the idea of reference tracking, which requires the generation of reference signals for all the three controlled variables. In Figure 5.3 a separate block is used for reference calculation, where the input power references $P_{o,\text{ref}}$, $Q_{o,\text{ref}}$ are used to calculate the reference signals for the grid-side current $i_{g,\text{ref}}^{\alpha\beta}$. First the reference currents in the (d, q)-coordinates are calculated as

$$i_{g,\text{ref}}^{d} = \frac{P_{o,\text{ref}}}{1.5 \, u_{g}^{d}}, \qquad i_{g,\text{ref}}^{q} = \frac{Q_{o,\text{ref}}}{1.5 \, u_{g}^{d}} \qquad (5.13)$$



Figure 5.5: Per-phase model of the LCL filter.

where the q-axis components of the grid $u_g^q \approx 0$. Further the currents $i_{g,\text{ref}}^{dq}$ are transformed to $i_{g,\text{ref}}^{\alpha\beta}$ using Clarke transformation. Figure 5.5 shows a per phase model of the LCL filter which is referred to calculate the reference signals for the other controlled variables using the grid-current reference. From Figure 5.5 we can write

$$u_x = u_g + i_{g,\text{ref}}(R_g + s \cdot L_g)$$

$$u_{f,\text{ref}} = \frac{u_x}{1 + s \cdot RC_f}$$

$$i_{m,\text{ref}} = i_{g,\text{ref}} + \frac{s \cdot u_x C_f}{1 + s \cdot RC_f}$$
(5.14)

The system of equations (5.14) generates a set of signals, in the three-phase (a, b, c) system, which are transformed to the stationary $(\alpha\beta)$ -reference frame. To summarize, the reference vector is

$$\boldsymbol{y}_{\text{ref}} = [(\boldsymbol{i}_{m,\text{ref}}^{\alpha\beta})^{\top} \quad (\boldsymbol{u}_{f,\text{ref}}^{\alpha\beta})^{\top} \quad (\boldsymbol{i}_{g,\text{ref}}^{\alpha\beta})^{\top}]^{\top}$$

where $\boldsymbol{i}_{g,\text{ref}}^{\alpha\beta} = \mathbf{T}_{\text{p}} \boldsymbol{i}_{g,\text{ref}}^{dq}, \qquad \boldsymbol{u}_{f,\text{ref}}^{\alpha\beta} = \mathbf{T}_{\text{c}} \boldsymbol{u}_{f,\text{ref}}^{abc}, \text{ and } \qquad \boldsymbol{i}_{m,\text{ref}}^{\alpha\beta} = \mathbf{T}_{\text{c}} \boldsymbol{i}_{m,\text{ref}}^{abc}.$ (5.15)

Figure 5.6 shows the generated reference signals for the controlled variables, where the power references are $P_{o,\text{ref}} = 4.9 \text{ kW}$, and $Q_{o,\text{ref}} = 0$.



Figure 5.6: Reference signals for the main inductor current $i_{m,ref}$, grid current $i_{g,ref}$ and filter capacitor voltage $u_{f,ref}$ for an active power reference of 4.9 kW, which is enabled at 0.05 s.

5.3 Performance evaluation of FCS-MPC

The performance of the proposed predictive controller was verified with simulations in Matlab/Simulink. The system under consideration is a three-phase sFCI rated for 5 kW. In addition to the system parameters, given is Table 3.1, Table 5.1 contains the implementation and controller data for the predictive controller. Generally the switching frequency of FCS predictive controller is variable and lesser compared to the conventional control techniques. This is due to the fact that only one output voltage vector is applied during the complete switching cycle. Furthermore, the same voltage vector might be used during subsequent switching periods, if any additional constraints are not specified, therefore the generated switching frequency is variable.

In order to compare the proposed controller with the previously designed SFCC, see section 4.2, the average switching frequency of the predictive controller must be comparable to the fixed switching frequency of the SFCC. Hence, the sampling time and controller step time have been reduced to 3.5 µs, and switching frequency penalization is minimal.

Parameter	Symbol	Value
Sampling time	T_s	3.5 µs
Controller step time	T_{ctl}	$3.5\mathrm{\mu s}$
Nominal grid voltage	u_g	$230\mathrm{V(rms)}$
Grid Inductance	L_{qrid}	$0.01\mathrm{mH}$
Grid Resistance	$\tilde{R_{qrid}}$	0.1Ω
Damping Resistor	R_d	1.6733Ω
Weighting factor matrices	$oldsymbol{Q},oldsymbol{R}$	Q = diag(20, 20, 5, 5, 150, 150)
		$R = diag(1, 1, 1, 1, 1, 1) \times 1e^{-4}$

Table 5.1: Implementation and controller data.



Figure 5.7: Steady-state simulation results for the grid-connected three-phase sFCI with FCS-MPC.

5.3.1 Steady-state performance

For an active-power reference of 4.9 kW and zero reactive power, the grid-current reference (peak) $i_{g,\text{ref}}$ is set to 10 A. The steady-state performance of the three-phase sFCI is shown in Figures 5.7(a) and (b). The flying capacitor voltages are maintained below the allowable limit of 450 V, as seen in the Figure 5.7(a).

Despite the variation in the input dc supply, the grid current remains sinusoidal with a THD of 2% (see Figure 5.7(b)). Unlike the SFCC where the grid current deviates from the sinusoidal behavior due to an uneven bump around zero crossing (see Figure 4.8), the current response of the FCS-MPC is superior and has comparable THD.

5.3.2 Dynamic performance

The dynamic response of the proposed MPC strategy is shown in Figures 5.8, 5.9, and 5.10. At 0.005 s the active power demand is stepped to 4.9 kW and at 0.035 s the reactive power demand is set to 1.5 kVAr (see Figure 5.10). Figure 5.8 compares the three controlled variables u_f , i_m , and i_g with their respective references. It is observed that the predictive controller has excellent transient



(a) Filter capacitor voltage and their references (dash-dotted lines).



(b) Main inductor currents and their references (dash-dotted lines).



Figure 5.8: Dynamic response of the grid-connected three-phase sFCI with FCS-MPC for a prediction horizon of $N_p = 3$.



Figure 5.9: Reference tracking of the grid currents.

response, compared to the SFCC, with less overshoot and settling time. In addition, all the variables follow their references implying superior reference tracking. The grid current tracking performance is shown in Figures 5.9(a) and (b). It can be observed that, as soon as the reference value changes the controller is able to follow with minimal overshoot. From Figures 5.9(a) and (b) we can conclude that the sFCI can deliver both active and reactive power, as and when the need arises.



(b) Average switching frequency of the converter.

Figure 5.10: Closed-loop simulation results of the sFCI with FCS-MPC.

Figure 5.10(a) highlights the voltage control of flying capacitors during system transients. Additionally, it also depicts the reference tracking of the desired active and reactive power references. With an average switching frequency of 38 kHz (see Figure 5.10(b)) the performance of the proposed MPC is comparable and even better than the state-feedback controller.

5.4 Drawbacks of classical MPC

Finite-control-set model predictive control (FCS-MPC) has received a lot of attention (see e.g., [14, 16, 28, 31, 72]). It exploits the finite number of the switching states of a power converter and combines current control and modulation into one computational step. Simple concept, faster dynamics and flexibility in terms of the control target realization are the advantages. In the past decade, research on FCS-MPC has spread across various fields, e.g., renewable energy systems, multi-level converters, and electrical drives [35].

The concept of the classical FCS-MPC³, has already been introduced in section 5.2. To ease the problem description, this concept is depicted in Figure 5.11, and denoted as classical MPC.



Figure 5.11: Classical MPC illustration [35].

FCS-MPC evaluates a given cost function for each element of an admissible (finite) set S. The switching sequence which minimizes the cost function will be chosen and applied for a whole control interval. Typically, this technique *enumerates* all the admissible switching states to perform the state prediction and cost minimization, i.e., using "exhaustive search" manner to solve the optimization problem. This leads to extremely *heavy computational efforts*, in particular for multilevel power converters [31, 35].

5.4.1 Drawback 1: Heavy computational efforts due to enumeration

One of the major drawbacks of FCS-MPC for multi-level inverters is that: the computation time increases drastically as the number of switching states (e.g. 27 switching states for a 3L-sFCI) increases and therefore implying that real-time implementation may not be feasible [26, 35].

Recent literature, e.g., [31, 55], have already shown that long prediction horizons result in a drastically improved system performance: e.g., lower current THD is obtained. However, the computation time increases exponentially as the number of prediction steps increases, and hence real-time realization is not possible. This emphasizes the necessity to investigate alternate methods for the optimization problem. Therefore, many of the recent research works proposed new methods that reduce the the computational burden and enable real-time implementation. For example, [73,74] proposed a sphere decoding method to control a 3L-NPC inverter-fed induction machine. It was proven to be effective in particular for multilevel and multiple prediction cases. In [75] three computationally efficient schemes, namely, move blocking, extrapolation, and event-based horizon MPC, have been discussed.

5.4.2 Drawback 2: Variable switching frequency

Another drawback of the FCS-MPC scheme is that it gives a variable switching frequency. Generally, in case of an LCL filter, the average switching frequency of the converter should be at least twice the resonant frequency, in order to avoid the excitation of resonant modes. This requires the system

³also known in literature as direct model predictive control (DMPC)

to sample at a very fast rate, e.g., the sampling time of FCS-MPC in the previous section was 3.5 µs, which makes the real-time implementation difficult. Additionally, the inherently variable switching frequency of the FCS-MPC approach gives rise to variable switching losses.

A possible solution to these drawbacks is to use a modulator with a fixed switching frequency, where the input duty-cycle of the modulator is generated from the predictive controller. In the following section, a control scheme known as continuous-control-set model predictive control, is proposed for a three-phase 3L-sFCI.

5.5 Continuous-control-set model predictive control (CCS-MPC)

This section presents a MPC scheme using the concept of unconstrained minimum, see [55, Section 5.2]. Also known as CCS-MPC, this scheme allows the use of long prediction horizon, i.e., $N_p > 3$ and simplifies real-time implementation, because most of the calculation can be done off-line. Moreover, when using long horizon approach for control of grid-connected inverters, use of additional resonance damping methods for the LCL filter is not necessary. First the prediction model is presented and then the control scheme is discussed in detail.

5.5.1 Continuous-time mathematical model

A continuous-time model of the LCL filter/grid, in the (α, β) -frame, has already been presented in section 5.2.1. Instead of treating the grid voltages as disturbance inputs, a simple oscillator can be used to model their sinusoidal nature [76]. Assuming the voltages u_g^{α} and u_g^{β} to be sinusoidal signals of fixed frequency $\omega_g = 2\pi f$, an oscillator can be modeled using a two-state linear system as

$$\begin{bmatrix} \dot{x}_{osc_1} \\ \dot{x}_{osc_2} \end{bmatrix} = \underbrace{\begin{bmatrix} 0 & \omega_g \\ -\omega_g & 0 \end{bmatrix}}_{\mathbf{T}_{osc}} \begin{bmatrix} x_{osc_1} \\ x_{osc_2} \end{bmatrix}$$
(5.16)

Combining the matrices **F**, **T** from (5.1), (5.2) and **T**_{osc} from (5.16) gives an augmented system. The state vector becomes $\boldsymbol{x} = [i_m^{\alpha} \ i_m^{\beta} \ u_f^{\alpha} \ u_f^{\beta} \ i_g^{\alpha} \ i_g^{\beta} \ u_g^{\alpha} \ u_g^{\beta}]^{\top} \in \mathbb{R}^8$ and the system model can be written as

$$\frac{\mathrm{d}}{\mathrm{d}t}\boldsymbol{x}^{\alpha\beta}(t) = \mathbf{F}_{a}\boldsymbol{x}^{\alpha\beta}(t) + \mathbf{G}_{a}\boldsymbol{u}_{m}^{\alpha\beta}(t)$$
(5.17a)

$$\boldsymbol{y} = \mathbf{C}_a \boldsymbol{x}^{\alpha\beta}(t) \tag{5.17b}$$

where the matrices of the augmented system are

$$\mathbf{F}_{a} = \begin{bmatrix} -\frac{R+R_{m}}{L_{m}} & 0 & -\frac{1}{L_{m}} & 0 & \frac{R}{L_{m}} & 0 & 0 & 0\\ 0 & -\frac{R+R_{m}}{L_{m}} & 0 & -\frac{1}{L_{m}} & 0 & \frac{R}{L_{m}} & 0 & 0\\ \frac{1}{C_{f}} & 0 & 0 & 0 & -\frac{1}{C_{f}} & 0 & 0 & 0\\ 0 & \frac{1}{C_{f}} & 0 & 0 & 0 & -\frac{1}{C_{f}} & 0 & 0\\ 0 & \frac{1}{L_{g}} & 0 & -\frac{R+R_{g}}{L_{g}} & 0 & -\frac{1}{L_{g}} & 0\\ 0 & \frac{R}{L_{g}} & 0 & \frac{1}{L_{g}} & 0 & -\frac{R+R_{g}}{L_{g}} & 0 & -\frac{1}{L_{g}} \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & \omega_{g}\\ 0 & 0 & 0 & 0 & 0 & 0 & -\omega_{g} & 0 \end{bmatrix}, \mathbf{G}_{a} = \begin{bmatrix} \frac{1}{L_{f}} & 0\\ 0 & \frac{1}{L_{f}} & 0\\ 0 & 0\\ 0 & 0\\ 0 & 0\\ 0 & 0\end{bmatrix}$$

$$\mathbf{C}_a = \mathbf{I}_{8 imes 8}, \ \ ext{and} \ \ \boldsymbol{\xi}^{lpha eta} = \mathbf{T}_{ ext{c}} \ \boldsymbol{\xi}^{abc}.$$

(5.18)

5.5.2 Discrete-time model

Again, invoking assumptions (A.4), (A.5) and (A.9), the continuous-time model (5.17), (5.18) can be discretized using zero-order-hold method. The exact discrete-time model is written as

$$\boldsymbol{x}^{\alpha\beta}(k+1) = \mathbf{A}\boldsymbol{x}^{\alpha\beta}(k) + \mathbf{B}\boldsymbol{u}_m^{\alpha\beta}(k)$$
(5.19a)

$$\boldsymbol{y}(k) = \mathbf{C}\boldsymbol{x}^{\alpha\beta}(k) \tag{5.19b}$$

where k is the discrete-time index, T_s is the sampling interval, and the discretized system matrices are

$$\mathbf{A} = e^{\mathbf{F}_a T_s}, \ \mathbf{B} = \left(\int_0^{T_s} e^{\mathbf{F}_a \tau} \cdot e^{\mathbf{j}\omega_g \tau} d\tau\right) \mathbf{G}_a, \ \mathbf{C} = \mathbf{C}_a.$$
(5.20)

5.5.3 Control objectives

The main objective of the CCS-MPC is to regulate the grid current, main inductor current and filter capacitor voltage along their references. Additionally, the flying capacitor voltage must be maintained within the allowable limits. Moreover, the switching frequency of converter is constant because a modulator is used. To achieve the mentioned goals, the control algorithm calculates the optimal duty-cycle using the concept of unconstrained minimum, which is then applied to the PWM block for subsequent generation of gating signals.

5.5.4 Control and optimization problem

The control problem of MPC with reference tracking over a finite prediction horizon of length N_p was presented in (5.9), and is rewritten here as

$$J(k) = \sum_{l=k}^{k+N_p-1} \|\boldsymbol{y}_{\text{ref}}(l+1) - \boldsymbol{y}(l+1)\|_{\boldsymbol{Q}}^2 + \lambda_s \|\Delta \boldsymbol{s}(l)\|_2^2.$$
(5.21)

In (5.21) $\boldsymbol{y}_{\text{ref}} \in \mathbb{R}^8$ encompasses the references for LCL filter states and the grid variables. The first term of the cost function implements the objective of reference tracking with \boldsymbol{Q} as the weighting factor matrix. The second term implements the minimization of switching effort, where penalization is carried out using a non-negative weighting factor λ_s .

The dynamic evolution of the discrete-time prediction model (5.19) and (5.20) can be included in the cost function (5.21) using an optimization problem. Therefore, the remainder of this subsection closely follows the derivation of the optimization problem provided previously in [55, Section 5.2].

5.5.4.1 Derivation of optimization problem in vector form

By successively using the state equation (5.19a) (neglecting the superscripts) over the prediction horizon, we can write

$$\begin{aligned} \boldsymbol{x}(k+1) &= \mathbf{A}\boldsymbol{x}(k) + \mathbf{B}\boldsymbol{u}_m(k) \\ \boldsymbol{x}(k+2) &= \mathbf{A}\boldsymbol{x}(k+1) + \mathbf{B}\boldsymbol{u}_m(k+1) \\ &= \mathbf{A}^2\boldsymbol{x}(k) + \mathbf{A}\mathbf{B}\boldsymbol{u}_m(k) + \mathbf{B}\boldsymbol{u}_m(k+1) \\ &\vdots \end{aligned}$$

$$\boldsymbol{x}(k+N_p) = \mathbf{A}^{N_p} \boldsymbol{x}(k) + \mathbf{A}^{N_p-1} \mathbf{B} \boldsymbol{u}_m(k) + \dots + \mathbf{A}^0 \mathbf{B} \boldsymbol{u}_m(k+N_p-1)$$
(5.22)

The output trajectory vector for a prediction horizon of N_p was previously defined as

$$\mathbf{Y} = [\boldsymbol{y}^{\top}(k+1) \ \boldsymbol{y}^{\top}(k+2) \ \dots \ \boldsymbol{y}^{\top}(k+N_p)]^{\top}.$$
(5.23)

69

Substituting (5.23) in (5.19b) yields

$$\mathbf{Y}(k) = \mathbf{\Gamma} \boldsymbol{x}(k) + \mathbf{\Upsilon} \mathbf{S}(k), \tag{5.24}$$

where the matrices Γ and Υ are defined as

$$\boldsymbol{\Gamma} = \begin{bmatrix} \mathbf{C}\mathbf{A} \\ \mathbf{C}\mathbf{A}^2 \\ \vdots \\ \mathbf{C}\mathbf{A}^{N_p} \end{bmatrix} \quad \text{and} \quad \boldsymbol{\Upsilon} = \begin{bmatrix} \mathbf{C}\mathbf{B} & \mathbf{0}_{8\times 2} & \dots & \mathbf{0}_{8\times 2} \\ \mathbf{C}\mathbf{A}\mathbf{B} & \mathbf{C}\mathbf{B} & \dots & \mathbf{0}_{8\times 2} \\ \vdots & \vdots & & \vdots \\ \mathbf{C}\mathbf{A}^{N_p-1}\mathbf{B} & \mathbf{C}\mathbf{A}^{N_p-2}\mathbf{B} & \dots & \mathbf{C}\mathbf{B} \end{bmatrix}. \quad (5.25)$$

If we define the output tracking error as $\Delta y = y_{ref} - y$, the first term in the cost function can be written as

$$J_{1} = \sum_{l=k}^{k+N_{p}-1} \|\Delta \boldsymbol{y}(l+1)\|_{\boldsymbol{Q}}^{2} = \sum_{l=k}^{k+N_{p}-1} \Delta \boldsymbol{y}^{\top}(l+1) \boldsymbol{Q} \Delta \boldsymbol{y}(l+1)$$
(5.26a)

$$= [\Delta \boldsymbol{y}^{\mathsf{T}}(k+1) \dots \Delta \boldsymbol{y}^{\mathsf{T}}(k+N_p)] \boldsymbol{Q}_e [\Delta \boldsymbol{y}^{\mathsf{T}}(k+1) \dots \Delta \boldsymbol{y}^{\mathsf{T}}(k+N_p)]^{\mathsf{T}}$$
(5.26b)

$$= (\mathbf{\Xi}(k))^{\top} \mathbf{Q}_{e} (\mathbf{\Xi}(k)) = \|\mathbf{\Xi}(k)\|_{\mathbf{Q}_{e}}^{2}, \qquad (5.26c)$$

where $Q_e = \text{diag}(Q, \dots, Q)$ is a diagonal matrix and $\Xi(k)$ is the output error trajectory. Substituting (5.24) into $\Xi(k) = \mathbf{Y}_{\text{ref}}(k) - \mathbf{Y}(k)$ in (5.26c), we get

$$J_1 = \|\mathbf{Y}_{\text{ref}}(k) - \mathbf{\Gamma} \boldsymbol{x}(k) - \mathbf{\Upsilon} \mathbf{S}(k)\|_{\boldsymbol{Q}_e}^2.$$
(5.27)

In a similar way the second term of the cost function (5.21) can be rewritten as

$$J_2 = \lambda_s \|\mathbf{WS}(k) - \mathbf{Zs}(k-1)\|_2^2$$
(5.28)

with the matrices

$$\mathbf{W} = \begin{bmatrix} \mathbf{I}_{2} & \mathbf{0}_{2\times 2} & \dots & \mathbf{0}_{2\times 2} \\ -\mathbf{I}_{2} & \mathbf{I}_{2} & \dots & \mathbf{0}_{2\times 2} \\ \mathbf{0}_{2\times 2} & -\mathbf{I}_{2\times 2} & \dots & \mathbf{0}_{2\times 2} \\ \vdots & \vdots & & \vdots \\ \mathbf{0}_{2\times 2} & \mathbf{0}_{2\times 2} & \dots & \mathbf{I}_{2} \end{bmatrix} \quad \text{and} \quad \mathbf{Z} = \begin{bmatrix} \mathbf{I}_{2\times 2} \\ \mathbf{0}_{2\times 2} \\ \mathbf{0}_{2\times 2} \\ \vdots \\ \mathbf{0}_{2\times 2} \end{bmatrix}.$$
(5.29)

Combining (5.27) and (5.28) yields the cost function in vector form:

$$J = \|\mathbf{Y}_{\text{ref}}(k) - \mathbf{\Gamma} \boldsymbol{x}(k) - \mathbf{\Upsilon} \mathbf{S}(k)\|_{\boldsymbol{Q}_e}^2 + \lambda_s \|\mathbf{W}\mathbf{S}(k) - \mathbf{Z}\boldsymbol{s}(k-1)\|_2^2.$$
(5.30)

The first term in (5.30) penalizes the reference tracking error, while the second term penalizes the switching effort. After some further algebraic manipulations (see [55, Appendix 5.B]) the cost function can be written in a compact form

$$J = (\mathbf{S}(k))^{\top} \mathbf{H} \mathbf{S}(k) + 2(\mathbf{\Theta}(k))^{\top} \mathbf{S}(k) + \theta(k)$$
(5.31)

with

$$\mathbf{H} = \mathbf{\Upsilon}^{\top} \boldsymbol{Q}_{e} \mathbf{\Upsilon} + \lambda_{s} \mathbf{W}^{\top} \mathbf{W}$$
(5.32a)

$$(\boldsymbol{\Theta}(k))^{\top} = -(\mathbf{Y}_{ref}(k) - \boldsymbol{\Gamma}\boldsymbol{x}(k))^{\top}\boldsymbol{Q}_{e}\boldsymbol{\Upsilon} - \lambda_{s}(\mathbf{Z}\boldsymbol{s}(k-1))^{\top}\mathbf{W}$$
(5.32b)

$$\theta(k) = \|\mathbf{Y}_{\text{ref}}(k) - \mathbf{\Gamma} \boldsymbol{x}(k)\|_{\boldsymbol{Q}_e}^2 + \lambda_s \|\mathbf{Z} \boldsymbol{s}(k-1)\|_2^2.$$
(5.32c)

The cost function (5.31) consists of three terms. The first term is quadratic in the switching sequence $\mathbf{S}(k)$. The Hessian matrix \mathbf{H} is time-invariant if the system parameters are time-invariant. Also, the

Hessian is symmetric and positive definite for $\lambda_s > 0$. The second term is linear in the switching sequence $\mathbf{S}(k)$. Here the time-varying vector $\mathbf{\Theta}(k)$ is a function of the state vector at time step k, the output references $\mathbf{Y}_{ref}(k)$, and the previously chosen switch position $\mathbf{s}(k-1)$. The third term is a time-varying scalar that has the same arguments as $\mathbf{\Theta}(k)$ [55].

By completing the squares, (5.31) can be rewritten as

$$J = (\mathbf{S}(k) + \mathbf{H}^{-1}\mathbf{\Theta}(k))^{\top}\mathbf{H}(\mathbf{S}(k) + \mathbf{H}^{-1}\mathbf{\Theta}(k)) + \operatorname{const}(k).$$
(5.33)

The constant term in (5.33) is independent of $\mathbf{S}(k)$ and will not have any effect on the optimal solution. Omitting the constant term, we can reformulate the optimization problem as [55]

$$\mathbf{S}_{opt}(k) = \arg \min_{\mathbf{S}(k)} (\mathbf{S}(k) + \mathbf{H}^{-1} \mathbf{\Theta}(k))^{\top} \mathbf{H}(\mathbf{S}(k) + \mathbf{H}^{-1} \mathbf{\Theta}(k))$$
(5.34a)

subject to
$$S(k) \in [-1, 1].$$
 (5.34b)

5.5.4.2 Solution in terms of the unconstrained minimum

The optimization problem (5.34) is solved by minimization after neglecting all constraints, if any (hence termed as *unconstrained minimum*), and thus allowing $\mathbf{S}(k) \in [-1, 1]$. According to [55], the unconstrained minimum is the optimal solution of the problem (5.34), and can be calculated at every time step k as

$$\mathbf{S}_{unc}(k) = -\mathbf{H}^{-1}\boldsymbol{\Theta}(k). \tag{5.35}$$

 $\mathbf{S}_{unc}(k) \in \mathbb{R}^{N_p \times 2}$ contains a sequence of values that signify the required duty cycle for operating the inverter at the next time step. In order to provide feedback and control the system, only first set of values from the generated solution $\mathbf{S}_{unc}(k)$ is used for further application to the modulator.

The optimal solution used for feedback $\mathbf{S}_{unc,opt}(k) \in \mathbb{R}^{1 \times 2}$ contains only two values corresponding to the bridge voltage reference in (α, β) -frame. These values are transformed to the (abc)-reference frame using the inverse Clarke transformation and fed to the PWM block. The desired duty-cycle, which acts as the input to the PWM block, is defined as

$$\boldsymbol{u}_{abc,ref}(k) = \underbrace{\begin{bmatrix} 1 & 0\\ -\frac{1}{2} & \frac{\sqrt{3}}{2}\\ -\frac{1}{2} & -\frac{\sqrt{3}}{2} \end{bmatrix}}_{=:\mathbf{T}_{c}^{-1} \in \mathbb{R}^{3 \times 2}} (\mathbf{S}_{unc,opt}(k))^{\top} .$$
(5.36)

5.5.5 Controller block diagram

The scheme of the proposed CCS-MPC is illustrated in Figure 5.12. As can be observed, the control block takes in all the measurements and predicts the future outcomes of all the controller variables, i.e., grid current, main inductor current, and the filter capacitor voltage. The reference signals are separately calculated (see section 5.2.4) and fed to the optimization stage where the optimal solution is generated using the optimization problem (5.34). Transforming the optimal solution $\mathbf{S}_{unc,opt}(k)$ to the (*abc*)-reference frame yields the required duty cycle $u_{abc,ref}$, which is supplied to the modulator for generation of switching signals s_{abc} . Additionally, the feed-forward compensation of *dc*-link voltages, defined previously in section 4.2.4, is used for scaling of the duty cycle inside the PWM block. The modulator generates the requisite gating signals and ensures desired system performance.

5.5.6 Discussion

From the above derivations we can observe that a MPC scheme based on this approach can be used to implement long horizons with less computational effort. Most of the matrices required for the calculation of the *unconstrained* solution can be calculated off-line. This removes the enormous



Figure 5.12: Continuous-control-set model predictive control with long prediction horizons for the three-phase sFCI using a modulator.

computational burden on the controller compared to the direct MPC approach. Also, the switching frequency of the system is fixed because of the modulator and hence the sampling frequency does not have to be very high. Both of these advantages make this control scheme feasible for real-time implementation.

5.6 Performance evaluation of CCS-MPC

The performance of the proposed CCS-MPC scheme was verified with simulations in Matlab/Simulink and Plecs blockset. The system under consideration is a three-phase sFCI rated for 5 kW. In addition to the system parameters, given is Table 3.1, Table 5.2 contains the implementation and controller data for the CCS-MPC scheme. A level shifted sinusoidal PWM (SPWM) (see Figure 4.7) with a carrier frequency of 40kHz is employed to generate the switching signals.

Parameter	Symbol	Value
Switching frequency	f_{sw}	$40\mathrm{kHz}$
Resonant frequency	ω_r	$2\pi \cdot 10.155 \mathrm{rad}\mathrm{s}^{-1}$
Sampling time	T_s	$25\mu{ m s}$
Controller step time	T_{ctl}	$25\mu{ m s}$
Nominal grid voltage	u_q	$230\mathrm{V(rms)}$
Grid Inductance	L_{grid}	$0.01\mathrm{mH}$
Grid Resistance	$\tilde{R_{qrid}}$	0.1Ω
Weighting factors	$ec{m{Q}},\lambda_s$	Q = diag(10, 10, 1, 1, 90, 90, 1, 1)
		$\lambda_s = 1e^{-4}$

 Table 5.2:
 Implementation and controller data.

5.6.1 Steady-state performance

For an active-power reference of 4.9 kW and zero reactive power, the grid-current reference (peak) $i_{g,\text{ref}}$ is set to 10 A. The steady-state performance of the three-phase sFCI is shown in Figures 5.13(a) and (b). The voltage of all the flying capacitors is maintained below the allowable limit of 450 V, as seen in the Figure 5.13(a). Additionally, the flying capacitors discharge below the input *dc*-link voltage and hence the sFCI works as desired.



(a) Voltage waveforms for the *dc*-link capacitors. (b) Converter bridge voltage and states of the LCL filter.

Figure 5.13: Steady-state performance of the continuous-control-set model predictive control with prediction horizon $N_p = 15$ for a grid-connected three-phase sFCI.

Despite the variation in the input dc supply, the grid current remains sinusoidal with an average THD of 2.3% (see Figure 5.13(b)). Unlike the SFCC where the grid current deviates form the sinusoidal behavior due to an uneven bump around zero crossing (see Figure 4.9), the current response of the CCS-MPC is superior and has comparable THD. Also, in comparison to the FCS-MPC scheme where the sampling time was set to 3.5 µs (see Section 5.2), the CCS-MPC scheme has a larger sampling time and thus makes it feasible for hardware implementation. Moreover, external methods for resonance damping are not required as this is taken care by the use of long-horizons.

5.6.2 Dynamic performance

The dynamic performance of the CCS-MPC strategy is shown in Figures 5.14, 5.15, and 5.16. At 0.005 s the active power demand is stepped to 4.9 kW and at 0.035 s the reactive power demand is set to 1.5 kVAr (see Figure 5.16). Figure 5.14 compares the three controlled variables u_f , i_m , and i_g with their respective references. As can be seen, the CCS-MPC has excellent transient response with less overshoot and very less settling time, and the performance is comparable to that of FCS-MPC. In addition, all the variables follow their references implying superior reference tracking. The grid current tracking performance is shown in Figures 5.15(a) and (b). It can be observed that, as soon as the reference value changes the controller is able to follow with minimal overshoot. The capability of the sFCI to deliver both active and reactive power is clearly demonstrated in Figures 5.15(a) and (b).



(a) Filter capacitor voltage and their references (dash-dotted lines).



(b) Main inductor currents and their references (dash-dotted lines).



(c) Grid currents and their references (dash-dotted lines).

Figure 5.14: Dynamic response of the grid-connected three-phase sFCI with CCS-MPC for a prediction horizon of $N_p = 15$.



(a) Active power reference is set to 4.9 kW at 5 ms. (b) Reactive power reference is set to 1.5 kVar at 0.035 s.

Figure 5.15: Reference tracking of the grid currents.

Figure 5.16(a) highlights the voltage control of flying capacitors during system transients. Additionally, it also depicts the reference tracking of the desired active and reactive power references. Unlike FCS-MPC, both SFCC and CCS-MPC have a fixed switching frequency due to the use of a modulator. From the simulation results we can conclude that the performance of the proposed CCS-MPC is better than the state-feedback controller, although additional resources would be required for real-time implementation.



Figure 5.16: Closed-loop simulation results of the sFCI with FCS-MPC, showing capacitor voltages and output power response.

5.7 Summary

This chapter presented two control strategies based on model predictive control for the three-phase sFCI: 1) finite-control-set model predictive control (FCS-MPC) using exhaustive enumeration, and 2) continuous-control-set model predictive control (CCS-MPC) using long horizon approach. To achieve a better dynamic and steady-state performance, all the three states of the LCL filter are controlled using reference tracking.

The first control strategy is based on the conventional MPC where the cost function penalizes the errors between the measured values and the reference values. The simulation results highlight the superior performance of the FCS predictive controller compared to the state-feedback controller. The results also show that the controller is able to achieve resonance damping, for the LCL filter resonance, using a simple passive damping resistor R_d . However, real-time implementation of this method is not practically feasible as the sampling frequency is quite high, and the optimization procedure becomes computationally demanding for longer control horizons.

To overcome the limitations of FCS-MPC, a more sophisticated optimization procedure is derived using the concept of unconstrained minimum. Unlike the conventional scheme, long horizon control can be easily implemented using this method as most of the calculation can be done *off-line*. Moreover, the switching frequency is fixed and therefore the control step time and sampling time of the system can be set to a practically feasible values. Hence, this control scheme is feasible for real-time implementation.

Based on the presented results, it can be concluded that MPC schemes have a better dynamic performance compared to a linear controller (state-feedback control), shorter settling times, and also achieve superior reference tracking. Additionally, model predictive control also gives a better performance at the steady-state operating conditions.

Chapter 6

Model Predictive Control using a Non-linear Converter Model

In the previous chapters the working of the sFCI was analyzed and control strategies based on state-feedback control and model predictive control were proposed. The control strategies were based on a linearized model of the filter/grid system, where the behavior of the flying capacitor and its interaction with the converter bridge was neglected. In order to capture the behavior of the flying capacitor, a non-linear converter model is derived in this chapter. To verify the correctness of the non-linear model, a direct model predictive control strategy is implemented for a single-phase sFCI.

6.1 Motivation

Although, using a flying capacitor in the sFCI allows to decrease the input dc-link requirement, additional problems arise due to the behavior of flying capacitor under no-load condition. One of the main drawbacks of the flying capacitor is that its voltage experiences a continuous increase when the load/grid is disconnected, see Figure 6.1(a). For the sFCI to work properly, the flying capacitor voltage has to be maintained below the maximum limit of 450 V. Moreover, the flying capacitor must discharge during the negative cycle to a voltage less than the input dc-link. When the load/grid is not present, the discharging path is open through the LCL filter and hence voltage on the flying capacitor voltage is observed (termed as voltage fly-away), which is detrimental to the operation of the inverter.



Figure 6.1: Voltage of the flying capacitor C_{FC} in no-load or grid-disconnected mode. (a) Voltage-flyaway condition for the flying capacitor, (b) voltage control when the reverse switch S_5 is used.

One of the solutions to this problem is to use an additional RB-IGBT (S_5) in an anti-parallel configuration with the switch S_4 , see Figure 6.2. The gating signals for the switch S_5 are the same



Figure 6.2: Phase leg of the three-level sFCI with an additional switch S_5 .

as the switch S_4 . In this manner the flying capacitor is allowed to discharge during the zero state of the negative cycle and hence the voltage can be maintained within the limits, see Figure 6.1(b). However, the reverse switch S_5 experiences high current stress and hence heats up very quickly.

Another solution is to model the behavior of the flying capacitor, in order to capture all possible dynamics of the system and design a control strategy where the voltage of the flying capacitor is controlled. In this method, the flying capacitor voltage can be defined as a state, and a direct MPC algorithm can be adopted to handle multiple control objectives, i.e., the regulation of the main inductor current, grid current and the capacitor voltage to their reference values, and control of the flying capacitor voltage under no-load condition.

In the following sections, a detailed analysis of the working states is presented and a non-linear model of the sFCI is derived, to capture the dynamics of the converter bridge, the flying capacitor and the LCL filter.

6.2 Analysis of the six working states of the sFCI

Unlike the three operating states proposed in [3, 13] and described in section 3.1.3, the inverter operation can be classified into six cases. Although the positive and negative states essentially remain the same, the zero state can be further classified into four cases, introduced in section 3.2.4. This classification is based on the cycle of operation and some additional conditions. In the following sections the inverter operation is analyzed in separate cases and mathematical equations that govern the respective case are derived.

6.2.1 Positive cycle

The following three cases exist during the positive cycle of the inverter operation. Switch S_2 remains off for the complete positive cycle. Note that the duty cycle from the controller is defined as

$$d(t) = \begin{cases} 1 & \text{if switching state is P or N,} \\ 0 & \text{if switching state is O.} \end{cases}$$
(6.1)

6.2.1.1 Case 1 : Positive state

In this case, the input dc-link supplies the load directly, providing both active and reactive power support. Only the switch S_3 is turned on (see Figure 6.3). The mathematical equations that govern the operation of this case are

$$i_{dc} = i_m; \qquad i_{FC} = 0;$$

$$u_m = u_{dc} d(t); \qquad \frac{d}{dt} u_{FC} = 0 \qquad (6.2)$$



Figure 6.3: Schematic diagram for Case 1, with respective current flow denoted as: ----- active current path, and ----- reactive current path.

6.2.1.2 Case 2 : Zero-state with $i_m > 0$

During the zero state both the switches S_1 and S_4 are turned on, although current flow in these switches depends on some additional factors. In this case $i_m > 0$, and the operation of the system can be described by the following mathematical equations

$$i_{dc} = i_m = i_{FC} \neq 0;$$

$$u_m = (u_{dc} - u_{FC}) \cdot d(t);$$

$$\frac{d}{dt}u_{FC} = \frac{1}{C_{FC}} \cdot i_{FC} = \frac{1}{C_{FC}} \cdot i_m = \frac{1}{C_{FC}} \cdot i_{dc}$$
(6.3)

From Figure 6.4 it is observed that whenever $i_m > 0$, charging current i_{FC} of the flying capacitor flows via $S_1 \rightarrow C_{FC} \rightarrow L_m$, while S_4 does not conduct any current. This behavior is due to the reverse biasing of the series diode in switch S_4 . Since the flying capacitor voltage is higher than the input *dc*-link, a negative voltage appears across the switch S_4 , reverse biasing its series diode.



Figure 6.4: Schematic diagram for Case 2, with respective current flow denoted as: active current path, and C_{FC} charging path.

6.2.1.3 Case 3 : Zero-state with $i_m < 0$

This case is observed whenever the current i_m becomes negative, due to its ripple. Again, the switches S_1 and S_4 are both turned on. Here the charging current i_{FC} takes the path $S_1 \to C_{FC} \to S_4$, see Figure 6.5.

Equations describing the operation are summarized as

$$i_{dc} = i_{FC} = 0; \qquad i_m < 0$$

$$u_m = 0; \qquad \qquad \frac{\mathrm{d}}{\mathrm{d}t} u_{FC} = 0$$
(6.4)



6.2.2 Negative cycle

This cycle can also be separated into three cases. Switch S_3 remains off for the complete negative cycle.

6.2.2.1 Case 4 : Negative state

During this state, only the switch S_2 is turned on. The flying capacitor, which charges during the zero state, acts as a virtual dc-link for the inverter. Figure 6.6 highlights the current flow paths during the negative state. Equations describing the operation can be summarized as

$$i_{m} = i_{FC} \neq 0, \qquad i_{dc} = 0;$$

$$u_{m} = -u_{FC} d(t); \qquad (6.5)$$

$$\frac{d}{dt}u_{FC} = \frac{1}{C_{FC}} \cdot i_{FC} = \frac{1}{C_{FC}} \cdot i_{m}$$



Figure 6.6: Schematic diagram for Case 4, with respective current flow denoted as: ------ active current path, and ------ reactive current path.

6.2.2.2 Case 5 : Zero-state with $u_{dc} < u_{FC}$

During the negative cycle, turning the switch S_2 off activates the zero state. When the flying capacitor voltage is higher than the input dc-link, the charging current is negligible and therefore no current flows via the switch S_1 . However, the negative current i_m flows via S_4 (see Figure 6.7), and the bridge voltage is equal to the voltage drop across the RB-IGBT, i.e., $u_{S_4} = (u_{dc} - u_{FC})$. The mathematical equations governing this case can be summarized as

$$i_{FC} = i_{dc} = 0$$

$$u_m = u_{S_4} = (u_{dc} - u_{FC}) < 0; \text{ will be neglected!}$$

$$\frac{d}{dt} u_{FC} = 0$$

$$u_{dc} + C_{in} S_2 + C_{FC} + U_{FC} + S_4 + U_m + C_f + U_f + U_g$$

$$u_{dc} + C_{in} S_2 + C_{FC} + U_{FC} + S_4 + U_m + C_f + U_f + U_g$$

$$(6.6)$$

6.2.2.3 Case 6 : Zero-state with $u_{dc} > u_{FC}$

When the flying capacitor voltage drops below the input dc-link, the charging current flows via $S_1 \rightarrow C_{FC} \rightarrow S_4$, see Figure 6.8. Additionally, current i_m also flows through S_4 , and the bridge voltage is again equal to the voltage drop across the RB-IGBT, i.e., $u_{S_4} = (u_{dc} - u_{FC})$. The governing equations for this case can be summarized as

$$i_{FC} = i_{dc} \neq 0;$$

$$u_m = u_{S_4} = (u_{dc} - u_{FC}) > 0; \text{ will be neglected!}$$

$$\frac{d}{dt} u_{FC} = \frac{1}{C_{FC}} \cdot i_{FC} = \frac{1}{C_{FC}} \cdot i_{dc}$$

$$i_{dc}$$

$$i$$



6.3 Non-linear mathematical model

The continuous-time linear system model of the sFCI, described by the system of equations (3.22), can now be extended to include the dynamics of the flying capacitor. Considering the flying capacitor voltage u_{FC} as an additional state, the new state vector becomes $\boldsymbol{x} = [i_m \ u_f \ i_g \ u_{FC}]^\top \in \mathbb{R}^4$. Based on the governing equations presented in the previous section a mathematical model can be written for each case, as follows:

Case 1:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_1 \boldsymbol{x}(t) + \mathbf{G} u_{dc}(t) + \mathbf{T}_1 u_g(t)$$
(6.8a)

Case 2:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_2 \boldsymbol{x}(t) + \mathbf{G} u_{dc}(t) + \mathbf{T}_1 u_g(t)$$
(6.8b)

Case 3:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_1 \boldsymbol{x}(t) + \mathbf{T}_1 u_g(t)$$
(6.8c)

Case 4:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_2 \boldsymbol{x}(t) + \mathbf{T}_1 u_g(t)$$
(6.8d)

Case 5:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_1 \boldsymbol{x}(t) + \mathbf{T}_1 u_g(t)$$
(6.8e)

Case 6:
$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}_1 \boldsymbol{x}(t) + \mathbf{T}_1 u_g(t) + \mathbf{T}_2 i_{dc}(t)$$
(6.8f)

here
$$\mathbf{F}_{1} = \begin{bmatrix} -\frac{R+R_{m}}{L_{m}} & -\frac{1}{L_{m}} & \frac{R}{L_{m}} & 0\\ \frac{1}{C_{f}} & 0 & -\frac{1}{C_{f}} & 0\\ \frac{R}{L_{g}} & \frac{1}{L_{g}} & -\frac{R+R_{g}}{L_{g}} & 0\\ 0 & 0 & 0 & 0 \end{bmatrix}; \quad \mathbf{G} = \begin{bmatrix} \frac{1}{L_{m}} \\ 0\\ 0\\ 0\\ 0 \end{bmatrix}; \quad \mathbf{T}_{1} = \begin{bmatrix} 0\\ 0\\ -\frac{1}{L_{g}} \\ 0 \end{bmatrix}$$
$$\mathbf{F}_{2} = \begin{bmatrix} -\frac{R+R_{m}}{L_{m}} & -\frac{1}{L_{m}} & \frac{R}{L_{m}} & -\frac{1}{L_{m}} \\ \frac{1}{C_{f}} & 0 & -\frac{1}{C_{f}} & 0\\ \frac{R}{L_{g}} & \frac{1}{L_{g}} & -\frac{R+R_{g}}{L_{g}} & 0\\ \frac{1}{C_{FC}} & 0 & 0 & 0 \end{bmatrix}; \quad \mathbf{T}_{2} = \begin{bmatrix} 0\\ 0\\ 0\\ \frac{1}{C_{FC}} \end{bmatrix}.$$

The output equation, for all the six cases, remains the same, i.e.,

W

$$\boldsymbol{y}(t) = \mathbf{C}\boldsymbol{x}(t)$$
 where $\mathbf{C} = \mathbf{I}_{4 \times 4}$. (6.9)

Note that u_g and i_{dc} are considered as external disturbances, and I denotes an identity matrix. Figure 6.9 depicts the various regions in reference to the grid voltage and also specifies the operation cases that are valid in respective regions.



Figure 6.9: Operation cases used in the mathematical model of the sFCI.

6.3.1 Continuous-time model of the system

The various cases of the system model (6.8) can be combined into one model that precisely describes the dynamics of the sFCI when operating in different states. To do so, three auxiliary variables d_{aux_1} , d_{aux_2} , and d_{aux_3} are introduced. These variables together with the cycle of operation, the state variable and operation modes indicate the operating case.

$$d_{aux_1} = \begin{cases} 1 & \text{if} \quad p_1 \cdot s_z = 1 \text{ and } i_m > 0, \text{ or } p_2 \cdot s_n = 1 \\ 0 & \text{if} \quad p_1 \cdot s_p = 1, \text{ or } p_1 \cdot s_z = 1 \text{ and } i_m < 0, \text{ or } p_2 \cdot s_z = 1 \end{cases}$$
(6.10)

$$d_{aux_2} = \begin{cases} 1 & \text{if} \quad p_1 \cdot s_p = 1, \text{or} \quad p_1 \cdot s_z = 1 \text{ and } \quad i_m > 0 \\ 0 & \text{if} \quad p_2 = 1, \text{or} \quad p_1 \cdot s_z = 1 \text{ and } \quad i_m < 0 \end{cases}$$
(6.11)

$$d_{aux_3} = \begin{cases} 1 & \text{if} \quad u_{FC} < u_{dc} \text{ and } i_m < 0 \\ 0 & \text{if} \quad \text{else} \end{cases}$$
(6.12)

Here the binary variables p_1 and p_2 denote the positive and negative half-cycles of operation (w.r.t the grid voltage), respectively. The binary variables s_p , s_n and s_z represent the positive (P), negative (N) and zero (O) states of operation, respectively. Note that $p_1 = (1 - p_2) = S^+$, where S^+ has been defined in (5.5). Taking all the above into account, the complete model of the system is written as:

$$\frac{\mathrm{d}\boldsymbol{x}(t)}{\mathrm{d}t} = \mathbf{F}\boldsymbol{x}(t) + \mathbf{G}u_{dc}(t) + \mathbf{T}_1 u_g(t) + \mathbf{T}_2 i_{dc}(t)$$
(6.13a)

$$\boldsymbol{y}(t) = \mathbf{C}\boldsymbol{x}(t) \tag{6.13b}$$

where the system matrices of the complete model are

$$\mathbf{F} = \begin{bmatrix} -\frac{R+R_m}{L_m} & -\frac{1}{L_m} & \frac{R}{L_m} & -\frac{d_{aux_1}}{L_m} \\ \frac{1}{C_f} & 0 & -\frac{1}{C_f} & 0 \\ \frac{R}{L_g} & \frac{1}{L_g} & -\frac{R+R_g}{L_g} & 0 \\ \frac{d_{aux_1}}{C_{FC}} & 0 & 0 & 0 \end{bmatrix}; \mathbf{G} = \begin{bmatrix} \frac{d_{aux_2}}{L_m} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix}; \mathbf{T}_1 = \begin{bmatrix} 0 \\ 0 \\ -\frac{1}{L_g} \\ 0 \end{bmatrix}; \mathbf{T}_2 = \begin{bmatrix} 0 \\ 0 \\ 0 \\ \frac{d_{aux_3}}{C_{fc}} \end{bmatrix}.$$
(6.14)

Figure 6.10 depicts the sFCI represented as an automaton. As can be seen, the transition from one case to another is specified by the auxiliary variables and the state-variables of the inverter.

6.3.2 Discrete-time model of the system

Based on the assumptions (A.4), (A.5) and (A.10), the continuous-time model (6.18), (6.19) is discretized using zero-order-hold method. The discrete-time model is written as

$$\boldsymbol{x}(k+1) = \mathbf{A}\boldsymbol{x}(k) + \mathbf{B}u_{dc}(k) + \mathbf{E}_1 u_g(k) + \mathbf{E}_2 i_{dc}(k)$$
(6.15a)

$$\boldsymbol{y}(k) = \mathbf{C}\boldsymbol{x}(k) \tag{6.15b}$$

where k is the discrete-time index, T_s is the sampling interval, and the discretized system matrices are

$$\mathbf{A} = e^{\mathbf{F}T_s}; \ \mathbf{B} = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{j\omega_g\tau} d\tau\right) \mathbf{G}; \ \mathbf{E}_v = \left(\int_0^{T_s} e^{\mathbf{F}\tau} \cdot e^{j\omega_g\tau} d\tau\right) \mathbf{T}_v \text{ where } v \in \{1, 2\}.$$
(6.16)

6.4 Direct model predictive control

In section 5.2 a detailed discussion on the design of a controller using the direct MPC approach was presented. The controller was tasked to control the grid current while regulating the main inductor



Figure 6.10: sFCI presented as a continuous-time automaton.

current and filter capacitor voltage along their references. A similar approach is adopted to design the predictive controller using the non-linear model-derived in the previous section. However, in addition to controlling the states of the LCL filter, voltage of the flying capacitor must also be controlled, particularly for no-load operation. Therefore, control of the sFCI is split up into two schemes: 1) when the sFCI is under no-load or grid-disconnected condition, and 2) when the sFCI is on-load/grid-connected and/or steady-state operating conditions exist.

In a broader sense, the main objectives of the controller remain the same, i.e., grid current control with regulation of the main inductor current and filter capacitor voltage. To handle the voltage fly-away condition of the flying capacitor, the controller works in the scheme-I. Whenever the system is connected to the grid, control changes to scheme-II where flying capacitor voltage does not require further regulation.



Figure 6.11: Block diagram of the direct MPC for a single-phase sFCI based on the automaton approach.

6.4.1 Control scheme-I

A three-level converter has three switching states, i.e., $\{N, 0, P\} \approx \{-1, 0, 1\}$ (see section 2.1.2). Due to the switching constraints in direct MPC, the transitions between the upper and the lower rails are forbidden, i.e., between -1 and 1. So the converter can either switch between $\{-1, 0\}$ or $\{0, 1\}$. However, for scheme-I we relax the switching constraints and allow the converter to switch between any of the three switching states, i.e., the constraint

$$\Delta s(k) = |s(k) - s(k-1)| \le 1 \qquad \text{where} \qquad s \in \mathcal{S} = \{-1, 0, 1\}$$

is not imposed on the controller during scheme-I. This allows the controller to generate a switching pattern wherein the flying capacitor can be discharged even during the positive cycle or the continuous charging operation can be prevented.

6.4.2 Control scheme-II

For scheme-II, the switching constraints are reinstated so that the converter does not switch between the upper and the lower rail, i.e., the transitions between -1 and 1 are forbidden. Additionally, the weighting factor for flying capacitor voltage control is set to zero and thus the system works as a three-level inverter with only three control objectives, i.e., control of i_g , i_m and u_f .

6.4.3 Control problem

The scheme of the proposed controller for the single-phase sFCI is illustrated in Figure 6.11. The desired system performance can be achieved by directly manipulating the inverter switches, without using a modulator. The proposed MPC algorithm first computes the evolution of the plant over

the prediction horizon (i.e. the trajectories of the variables of concern) based on the measurements of the grid current, main inductor current, filter capacitor voltage, flying capacitor voltage, and grid voltage. Following this, the optimal control action (i.e. the switching signals) is chosen by minimizing a performance criterion in real time.

As highlighted previously, constraining the flying capacitor voltage below 450 V is crucial for the working of the sFCI. Control scheme-I is primarily aimed at maintaining the flying capacitor's voltage near a reference value, as it tends to fly-away. This is possible if a discharging path is available or if the flying capacitor is not charged continuously (see the description of zero state in section 3.1.3), during the positive cycle. When the switching constraints are removed, the converter is allowed to switch between -1 and 1. Due to this the flying capacitor does not undergo continuous charging and mitigation of the voltage fly-away condition is possible. However, the converter bridge output for the positive cycle and the negative cycle are different. Figure 6.12(a) shows the converter bridge output when the controller works in scheme-I. As can be seen, the sFCI behaves like a two level inverter for the positive cycle, because of the cost minimization procedure of the direct MPC.

6.4.4 Cost function

The discrete-time model (6.15), (6.16) is used to predict the output y(t) of the system. At time-step k, the cost function that penalizes the error of the output variables over the finite prediction horizon of N_p time steps is written as

$$J(k) = \sum_{l=k}^{k+N_p-1} \|\boldsymbol{y}_{\text{ref}}(l+1) - \boldsymbol{y}(l+1)\|_{\boldsymbol{Q}}^2.$$
(6.17)

In (6.17) $\boldsymbol{y}_{\text{ref}} \in \mathbb{R}^4$ is a vector encompassing the reference values of the controlled variables (main inductor current, filter capacitor voltage, grid current, and flying capacitor voltage), i.e.

$$\boldsymbol{y}_{\text{ref}} = \begin{bmatrix} i_{m,\text{ref}} & u_{f,\text{ref}} & i_{g,\text{ref}} & u_{FC,\text{ref}} \end{bmatrix}^{\top}.$$
(6.18)

According to the automaton (see Figure 6.10), the operation case changes based on the auxiliary variables and the measured states of the system. Operation of this type of system where the state-space model of the converter changes continuously depending on binary variables, can be compared to piecewise affine systems.

The diagonal matrix $\mathbf{Q} \in \mathbb{R}^{4\times 4}$ is a positive semidefinite matrix and its diagonal entries are chosen in a way that the tracking accuracy among the four controlled variables is prioritized. In addition to relaxing the switching constraints during the scheme-I, priority is given to the flying capacitor voltage control by penalizing the corresponding error more heavily. This allows the control of u_{FC} and thus serves as an efficient solution compared to the use of a reverse switch, as highlighted previously in section 6.1.

Control scheme-II has only three control objectives, i.e., control of the three states i_g , i_m and u_f . The diagonal matrix Q is set to highly penalize the grid current error, and offer small penalization to the errors in main inductor current and filter capacitor voltage, while the weighting factor related to flying capacitor voltage is set to zero.

The optimal sequence of control actions is then computed by minimizing the cost function (6.22). Only first element of this optimal sequence is utilized, whereas the rest of the elements are discarded. At the next time step k + 1, the complete procedure is repeated for the updated measurements over a one-step shifted horizon, according to the receding horizon policy [55].

6.5 Performance evaluation of DMPC

Parameter	Symbol	Value
Nominal power (base power)	P_n	3.3 kVA
Converter side inductance	L_m	680 µH
Converter side resistance	R_m	$70\mathrm{m}\Omega$
Grid side inductance	L_{g}	80 µH
Grid side resistance	R_{g}	$8.5\mathrm{m}\Omega$
Filter capacitance	C_{f}	$5\mu\mathrm{F}$
Flying capacitor	C_{FC}	$700\mu\mathrm{F}$
Grid inductance	L_{grid}	$0.01\mathrm{mH}$
Grid resistance	R_{grid}	0.1Ω
DC link voltage	u_{dc}	$400\mathrm{V}$
Nominal grid voltage	u_g	$230\mathrm{V(rms)}$
Sampling time	T_s	$3.5\mu{ m s}$
Controller step time	T_{ctl}	$3.5\mu{ m s}$
Weighting factor matrices	$oldsymbol{Q}_1,oldsymbol{Q}_2$	$Q_1 = \text{diag}(10, 40, 1, 10)$
		$\boldsymbol{Q}_2 = \mathrm{diag}(20, 90, 1, 0)$

6.5.1 System description

Table 6.1: System parameters and controller implementation data.

To investigate the performance of the proposed controller using the non-linear converter model, we consider a modified system with the system parameters given in the Table 6.1. The system parameters were modified to decrease the resonant frequency of the LCL filter to 8.29 kHz and make it feasible for implementation of the proposed control approach. In addition to system parameters, the controller implementation data is also provided in the table.

The system under consideration is a single-phase sFCI rated for 3.3 kVA. The working of this control scheme is similar to the direct MPC approach used in Section 5.2. Therefore, the controller step time is set to 3.5 µs as the average switching frequency must be comparable to the state-feedback control. Note that the matrices Q_1 and Q_2 are the diagonal matrices for penalization of errors and correspond to the schemes I and II, respectively.

6.5.2 Performance of DMPC during control scheme-I

As highlighted earlier, the main aim of the scheme-I is to control the voltage of the flying capacitor during grid disconnected mode or no-load condition. Figure 6.12 compares the output bridge voltage and flying capacitor voltage during the schemes I and II. From Figure 6.12(a) we observe that the bridge voltage output during the positive cycle resembles a two-level inverter. This is due to the fact that the switching constraints have been relaxed and converter is allowed to switch between the upper and lower rails, i.e., -1 and 1.

When the grid is disconnected the controller is tasked to control the voltage of the flying capacitor with the weighting factor matrix Q_1 . This imposes a large cost on the flying capacitor voltage error and therefore the controller commands the converter to switch between the positive and negative states, in order to prevent continuous charging during the intermediate zero states, and hence the voltage fly-away condition can be mitigated. Figure 6.12(c) shows the voltage increase for 10 cycles of operation when the control scheme-I is used. Compared to the three-level operation (see Figure 6.12(d)) where the capacitor undergoes a voltage increase ≈ 30 V, the operation in scheme-I only sees a voltage rise ≈ 4 V. In this manner the voltage fly-away condition can be mitigated without the use of an extra anti-parallel switch.



Figure 6.12: (a),(b): Converter bridge voltages, and (c),(d): flying capacitor voltages for the control schemes I and II during grid disconnected mode.



Figure 6.13: Dynamic performance of the control scheme-I after grid connection.

The control scheme-I is not intended for inverter operation during grid-connected mode. Nevertheless, the performance after grid connection is depicted in Figure 6.13. It is observed that the controller

has the following features

- flying capacitor voltage is maintained around 400 V.
- converter bridge voltage alternates between the two-level and three-level behavior.
- Upon grid connection, the reference tracking of the controller is very fast. Additionally, the dynamic performance is clearly visible when the current reference is stepped from 6 A to 10 A at 0.105 s.

The control scheme-I performs reasonably well even after grid connection. However, THD of the grid current is quite high due to the uneven switching of the converter. Hence, the scheme-II addresses the operation in grid-connected mode.

6.5.3 Performance of DMPC during control scheme-II

Whenever the gird-connected state is detected, controller is switched from the operation scheme-I to scheme-II. In this scheme the switching constraints are reinstated. Therefore the converter is forbidden to switch between the two extreme states and hence the bridge voltage is three level, see Figure 6.12(b). In the following discussion the steady-state and dynamic performances of the proposed DMPC scheme are presented.



6.5.3.1 Steady-state performance

Figure 6.14: Steady-state performance of the DMPC scheme-II (with $N_p = 3$) for grid-connected single-phase sFCI.

The performance of the controller is verified for a reference current of $10 \text{ A} \approx 1.6 \text{ kW}$ (per phase), i.e., half of the nominal power. The steady-state performance results simulated in Matlab/Simulink are given in Figure 6.14. As can be seen, the *dc* side variables are stable, the flying capacitor voltage has a constant ripple and is maintained within the limits. The bridge voltage is uniform and the converter switches evenly between the three levels and therefore follows the grid implying proper grid synchronization. Since the reactive power demand is set to zero, the LCL filter states can be seen to

be in phase to the grid voltage in Figure 6.14(b). The control scheme gives an average switching frequency of $38 \,\mathrm{kHz}$.

6.5.3.2 Dynamic performance

The dynamic performance of the proposed control scheme is shown in Figures 6.15, 6.16 and 6.17. At 0.025 s the active power reference is set to 1.3 kW which is increased to 1.6 kW at 0.065 s. As can be seen, the converter operation is very stable and robust during system transients. In addition to this the transient operation of the grid current and its reference tracking is depicted in Figures 6.16 and 6.17. It is observed that the proposed control approach exhibits very good transient response, small overshoot, less settling time and above all superior reference tracking.



Figure 6.15: Dynamic performance of the DMPC scheme-II (with $N_p = 3$) for single-phase sFCI.



(a) Grid current, its reference and the grid voltage when active power reference is set to 1.3 kW.

(b) Comparison of grid current and its reference.

Figure 6.16: (a) Transient performance of the grid current as its reference is stepped to 6 A. The grid current is in phase with the grid voltage. (b) Grid current reference tracking of the controller with small overshoot and low settling time.



(a) Grid current, its reference and the grid voltage when active power reference changes to 1.6 kW.



Figure 6.17: (a) Transient performance of the grid current as its reference changes from 6 A to 8 A . The grid current remains in-phase with the grid voltage. (b) Grid current reference tracking of the controller.

Figures 6.16 and 6.17 show that the output grid current accurately tracks its reference, and produces an average output current THD of 2.5%.

6.6 Summary

In this chapter the main drawback of using the flying capacitor (voltage fly-away condition) has been highlighted and two solutions to mitigate this issue have been presented. The first solution where an extra switch is employed to discharge the capacitor violates the basic construction of the sFCI topology. Hence, an alternative solution whereby the flying capacitor voltage is separately controlled as a state is presented. The proposed control algorithm uses the non-linear converter model and works in two schemes, which have specific control motives. In this manner the grid disconnected and grid-connected modes are taken care of separately.

The performance of the proposed method was investigated in simulations. It has very good performance in both steady-state and transient operating conditions. On one hand the hybrid control technique with its operating scheme-I can control the voltage of the flying capacitor during off-grid conditions, while on the other hand it can work as a reference tracking controller during the scheme-II. It is also shown that the proposed strategy results in grid current THD of 2.5% for a prediction horizon of 3 steps.

Chapter 7

Hardware Implementation of State Feedback Control

In this chapter the implementation of the state-feedback controller on a single-phase hardware testbench is discussed. Experimental results show the working of the converter in grid-connected mode and confirm that the newly proposed sFCI topology can indeed be controlled. Due to time constraints and hardware limitations, only preliminary results obtained at lower power are presented here that prove the working of the designed controller.

7.1 Overview

For hardware implementation, the SFCC designed in section 4.1 was chosen due to limitations on the processing power of the microcontroller and testbench availability. The implementation was carried out using the controller TMS320F28379D from the C-2000 microcontroller family of Texas Instruments (TI). The hardware design of the inverter was carried out previously in [37]. To ease the working of the inverter, two separate PCBs were designed: 1) power PCB which includes the converter bridge and the dc-links, and 2) control PCB which includes the various components required for grid connection, measurements circuits and the microcontroller. For details of the design see [37, Ch. 5]. The hardware parameters can be found in Table 3.1, and the testbench is shown in Appendix A.

7.2 Description of the testbench

7.2.1 Hardware setup

Figure 7.1 shows an overview of the setup used for the implementation of the state-feedback current controller. It shows the various components used for enabling grid connection and control of power flow between the inverter and the grid. The control algorithm with associated modules runs on the microcontroller unit (MCU). An additional evaluation board is used to observe the behavior of the MCU and requires a separate software called *Online Oszi* (Fraunhofer internal). Furthermore, for interacting with the controller and setting different values to enable operation a separate in-house software called ISEMON is used which functions via the serial peripheral interface (SPI).

- **DC power supply**: A configurable *dc* power supply from Regatron AG is used to supply variable *dc* input to the sFCI converter.
- sFCI converter: A three-phase testbench of the inverter, designed in [37], is available for the controller implementation. However, due to the voltage-flyaway condition of the flying



Figure 7.1: Schematic representation of the test setup.

capacitor using the reverse switch during startup is crucial and hence only one of the phases could be used.

- Control card: A Delfino F28379D controlCARD (TMDSCNCD28379D) from Texas Instruments (TI) is used for implementation of the contorl algorithm. This 180-pin controlCARD contains the microcontroller, a JTAG emulator and some additional circuitry. The microcontroller is quite versatile as it contains 200 MHz dual C28x CPUs and dual CLAs. This allows independent software development on the CPU₁ and CPU₂ [77].
- Grid simulator: To generate customized grid voltages, a 4-quadrant amplifier is used made by Spitzenberger & Spies GmbH & Co. KG. It gives a very clean sinusoidal voltage output.
- Measurements: In order to control the system, voltage and current measurements are carried out using analog/digital (A/D) circuits and sigma-delta filter modules (SDFM). These circuits are designed separately onto the control PCB and the measured signals are supplied to the microcontroller.

7.2.2 Data acquisition and synchronization

The different voltage and current sensors used in the sFCI supply analog measurements that have to be processed and supplied to the DSP for subsequent use in the control algorithm. For conversion of analog sensor data to digital form, A/D converters (ADC) are employed. The ADCs used consist of data converters based on SAR (successive approximation) and sigma/delta filter modules (SDFM). Furthermore for grid connection, the filter capacitor voltage must be synchronized in phase and magnitude with the grid voltage. This is achieved using a single-phase PLL.

7.3 Implementation of the controller using C-code

For the implementation of the state-feedback current controller (SFCC) on the microcontroller, various additional modules are required which were previously developed at Fraunhofer ISE. Some blocks are described below:

- **DAQ**: This module takes care of the data acquisition using the ADCs and SDFMs. It contains the code to acquire measurement signals from respective channels and conversion of the data to meaningful values. The contribution to this module was the calibration of the different measurement signals.
- State machine: For the proper functioning of the converter and to increase the flexibility of using the microcontroller, a state machine is defined. It consists of different states that

accomplish various objectives. The previously developed state machine at Fraunhofer ISE was modified to suit the working of the controller. More specifically, modules named dc-link pre-charge, u_f ramp-up and grid connection were used frequently.

- **Debugging**: It includes modules that allow communication via SPI. This enables the use of *Online Oszi* and ISEMON.
- **ePWM**: The microcontroller has internally provided modules named "enhanced PWM". These are programmed in such a way that the duty cycle generated from the current controller block acts as their reference input.

To enable control of the inverter three modules were developed, i.e., the current controller, phaselocked-loop (PLL), and the observer. To simplify the procedure, code generation feature of Simulink was used to generate the C-code for all of these modules. Later this code was integrated with the modules described above.

7.3.1 Current controller

The state-feedback controller is based on the structure presented in Figure 4.2 and described in sections 4.1 and 4.2. The gain vector is calculated using Matlab scripts, as per the design specifications in section 4.2. Simulink code generation is set up to generate a *non-reusable* function with output argument as the calculated duty cycle. In a *non-reusable* function, the model data structure is statically allocated and can be accessed by model entry point functions directly from the model code.

Note that this module is specified to run from the RAM area of the system memory, as it should be run in parallel to the high speed loop which has a sampling time of 25 µs.

7.3.2 PLL

A single-phase PLL is designed using SOGI based structure presented in [78]. It has a very simple implementation and offers inherent filtering characteristic. This module generates a sine wave that is in-phase with the grid and calculates the magnitude of the grid. Figure 7.2 shows the output of the PLL w.r.t the grid-voltage measurement.

Note that this module is placed into the control law accelerator (**CLA**) (for details see [77]). At each time step the grid voltage measurement is supplied to the *CLAdata*, and the estimated grid magnitude and sine wave are copied into the main memory area.



Figure 7.2: Experimental output of the PLL.



Figure 7.3: Grid connection procedure.

7.4 Grid connection procedure

Before closing the grid-side relays to connect the inverter with the grid, several steps have to be followed. The various steps followed during the connection procedure are depicted in Figure 7.3 and described below.

7.4.1 Pre-charging the *dc*-link capacitors

The *dc*-link of the sFCI is composed of two capacitors, i.e., C_{in} and C_{FC} . The first step is to charge both of these capacitors to the peak of the grid voltage. This is accomplished via the pre-charge circuits and the commands are initiated manually, to avoid any harzards. Once the voltages on both of these capacitors is equal, we can proceed to the next step. Moreover, the PLL is initialized to start the grid tracking.

Note: Switch S_5 is used in this step to prevent the fly-away condition, and its gating signal is equivalent to the gating signal of the switch S_4 . However, switching signals of S_4 and S_5 are generated using different ePWM modules to allow separate control.

7.4.2 Filter capacitor voltage ramp-up

In this step the grid voltage measurement is used as the duty cycle for the inverter. However, the value starts from zero and is ramped up in steps. This charges the filter capacitor to the peak of the grid voltage and sets up a sinusoidal voltage on it, which is in-phase with the grid and has the same magnitude. For this step, the switch S_5 is necessary as the inverter is still off-grid. Figure 7.4 shows the ramp-up of u_f in comparison to the grid voltage. Figure 7.5 shows the measurements after ramp-up procedure has been completed. As can be seen, the grid voltage and filter capacitor voltage have the same magnitude and phase.


Figure 7.4: Filter capacitor voltage during ramp-up compared to the grid voltage.



(a) u_f and u_g measurements after ramp-up procedure. (b) Osci

(b) Oscilloscope measurements after ramp-up. Grid voltage (dark blue) and filter capacitor voltage (pink).



7.4.3 Grid connection and current control

This is the last step in this procedure. Once the filter capacitor has a sinusoidal voltage on it the state machine is changed to grid connected mode. In this mode the following steps happen in a sequence

- PWM signals for all the switches are disabled, including S_5 .
- Current controller is initialized and the grid side relays are closed.
- After one cycle delay the PWM signals for switches S_1 to S_4 are enabled, while for S_5 the PWM is disabled.
- Based on the current reference the closed-loop controller generates the desired duty-cycle and the inverter starts modulating accordingly.

In this manner the sFCI is connected to the grid and supplies the necessary amount of power as specified by the current reference.

7.5 Experimental measurements

This section presents the experimental measurements for the open-loop and closed-loop operation of the sFCI. Since the converter is relatively new and issues that arise during grid-connection have not been investigated, we start with low values of grid voltage, generated using a grid simulator.

To start with, the operation of the converter was verified in open-loop for an input dc-link voltage of 35 V (see Figure 7.6(a)) and subsequently tested for closed-loop grid connected operation with a grid voltage of 20 V(rms), see Figure 7.6(b). After the measurements (in Figure 7.6) confirmed the operation of the single-phase converter and the designed controller, further experiments with more voltage on the grid are presented below. Table 7.1 contains the nomenclature of the oscilloscope measurements, unless specified otherwise.

Measurement signal	Color
Load voltage/Grid Voltage	dark blue
Converter bridge voltage	pink
Grid-side inductor current	light blue
Voltage ripple of flying capacitor	green

 Table 7.1: Nomenclature of oscilloscope measurements.



(a) Open-loop measurements for a resistor of 34Ω with input *dc*-link voltage = 42 V.

(b) Closed-loop measurements with a grid voltage of 20 V (rms) and a reference current of 1.5 A.

Figure 7.6: Preliminary measurements for control of the single-phase sFCI. Here main-inductor current is depicted in light blue and filter capacitor voltage is depicted in pink.

7.5.1 Converter Response for 50 V grid

Figure 7.7 shows the open-loop and closed-loop measurements when a grid voltage of 50 V(rms) is selected. Figure 7.7(a) shows the operation of the converter in open-loop with a load resistor of 34Ω . In this case the input *dc*-link is set to 92 V and the current drawn by the resistor is 1.47 A(rms), while the load voltage is 50.57 V(rms). In order to compare the closed-loop operation of the inverter, the current reference is set to a peak value of 2.4 A while the grid simulator provides a grid voltage of 50 V(rms). Figure 7.7(b) shows the performance of the controller in grid connected mode.

7.5.2 Converter Response for 80 V grid

Figure 7.8 shows the open-loop and closed-loop measurements when a grid voltage of 80 V(rms) is selected. Figure 7.8(a) shows the operation of the converter in open-loop with a load resistor of 34Ω . In this case the input *dc*-link is set to 140 V and the current drawn by the resistor is 2.3 A(rms), while the load voltage is 79.27 V(rms). The grid simulator provides a grid voltage of 80 V(rms) and



(a) Open-loop measurements for a resistor of 34Ω with input *dc*-link voltage = 92 V.

(b) Closed-loop measurements with a grid voltage of 50 V (rms) and a reference current of 2 A.

Figure 7.7: Experimental measurements for implementation of the state-feedback control on the single-phase sFCI with 50 V grid.

the current reference is set to a peak value of $3.3 \,\text{A}$. Figure 7.8(b) shows the performance of the controller in grid connected mode.

From Figure 7.8(a) it can be observed that the load voltage and current is not purely sinusoidal, as highlighted previously in section 3.2.2. From Figure 7.8(b) it can be seen that the grid current is in-phase with the grid voltage. Additionally, it is observed that the deviation from sinusoidal behavior is present at the transition from positive and negative cycle.



(a) Open-loop measurements for a resistor of 34Ω with input *dc*-link voltage = 140 V.

(b) Closed-loop measurements with a grid voltage of 80 V (rms) and a reference current of 3.3 A.

Figure 7.8: Experimental measurements for implementation of the state-feedback control on the single-phase sFCI with 80 V grid.

7.6 Discussion and summary

This chapter presented the experimental measurements for the operation of sFCI with the statefeedback controller. It can be concluded that the Siwakoti-H inverter can be controlled in gridconnected mode using SFCC.



Figure 7.9: Thermal image of the sFCI converter with temperature probing spot on switch S_4 .

The measured current response contains a lot of distortions due to various contributing factors. Firstly, the topology contains two different switches, i.e., MOSFETs and RB-IGBTs which have a lot of difference in switching behavior, hence more investigation is required in this direction. In addition to this, the hardware test-bench, built at Fraunhofer ISE, was not designed to incorporate RB-IGBTs, instead it used series diodes and MOSFETs. Therefore, the gate driver circuits used previously are not completely compatible and require further improvement. Moreover, a dead-time of 300 ns is used in the PWM block, while the controller does not contain any dead-time compensation.

Experimental measurements could not be carried out at higher voltages as the case temperature of RB-IGBTs (particularly S_4) approached the maximum allowed limit of 110 °C, for a *dc*-link voltage of 190 V (see for example Figure 7.9) and were mostly destroyed due to thermal runaway. Further investigation into this behavior could not be carried out due to time constraints. To sum up, the presented results prove the working of the state-feedback controller for the grid-connected sFCI. The response of the controller can be further improved by enhancing the data acquisition module, tuning the controller parameters and introducing dead-time compensation.

Chapter 8

Conclusion and Future Outlook

In this work, a recently proposed transformerless inverter, i.e., the Siwakoti-H flying capacitor inverter has been investigated. The operating principle, working modes, and performance in openloop/closed-loop has been discussed. The Siwakoti-H inverter utilized in this thesis has the potential of becoming an attractive alternative to conventional topologies as it can be realized using lesser and smaller components. However, it presents an interesting problem for system modeling and control design due to its additional non-linearities. Therefore, both conventional and modern control techniques for the grid-connected sFCI have been investigated. Emphasized efforts have been put into the predictive control due to its attractive properties (e.g., simple concept and fast control dynamics, etc.) and promising potential. All the presented control algorithms have been validated in simulation. Finally, the state-feedback approach has been verified experimentally for a single-phase sFCI using a microcontroller.

8.1 Conclusion

The work carried out in this thesis can be briefly summarized in the following points.

- In Chapter 2, a literature review was conducted with regards to the conventional threelevel inverter topologies and the popular control strategies used in grid-connected inverter applications. The basics of three-level inverters in terms of the switching vectors, was also discussed. Additionally, the sFCI was introduced as an attractive alternative to the conventional topologies. Furthermore, the theoretical and mathematical background of state-feedback control and model predictive control was introduced to build the foundations for the design of these controllers.
- In Chapter 3, first the design of the sFCI and its working in different modes was discussed as proposed by the designer. Then, a detailed analysis of the operating modes was carried out and the additional non-linearities of this topology were highlighted. In order to understand the different voltage and current responses of the sFCI during open-loop operation, a comparison was drawn with the conventional NPC converter. Finally, the three-phase sFCI, equipped with a LCL filter, was introduced and derivation of the mathematical model was presented, which is a prerequisite for the design of any control technique.
- In Chapter 4, a current control scheme based on the concept of state-feedback was designed separately for the single-phase and three-phase sFCI connected to the grid via an LCL filter. The controller designed in the (a, b, c)-reference frame (per-phase basis) was able to fulfill all objectives for the single-phase system. However, for the three-phase system, only a controller designed in (d, q)-reference frame was able to achieve all the control objectives. This highlighted the fact that there exists an in-built coupling between the flying capacitors of the three phases of the sFCI.

A pole-placement based design procedure was presented to achieve better dynamic and steadystate performance. The simulation results show that the resonance of the LCL filter is well damped, and the dynamic performance specified by direct pole placement is obtained for the reference tracking, grid voltage disturbance rejection, and balancing of flying capacitor voltages.

In Chapter 5, a direct MPC—as a current controller—was proposed for the three-phase sFCI, and designed in the $(\alpha\beta)$ -reference frame. The main motivation of using MPC was the uneven deviation in the current response of the state-feedback controller while transitioning from positive to negative cycle. The proposed MPC strategy was set to simultaneously control all the three states of the LCL filter via reference tracking. In a nutshell, the conventional MPC uses the cost function to penalize the errors between the measured values and the reference values. The simulation results highlighted the superior performance of the direct predictive controller compared to the state-feedback controller. The results also proved that the controller was able to achieve resonance damping using a simple passive damping resistor R_d .

Furthermore, two major drawbacks of the classical direct model predictive control were discussed, namely *heavy computational efforts* and *variable switching frequency*. To overcome the heavy computational efforts, a sophisticated optimization procedure was derived using the concept of unconstrained minimum, which facilitates the use of long horizon control. Additionally, the problem of variable switching frequency was resolved by employing a modulator. This restricts the converter to a fixed switching frequency and hence the controller step time and sampling time were set to realistic values. The presented simulation results show that MPC schemes have a better dynamic performance compared to a linear controller (state-feedback control), shorter settling times, and also achieve superior reference tracking.

■ In Chapter 6, one of the main drawbacks of using the flying capacitor, i.e., the voltage fly-away condition was highlighted. It was observed that the C_{FC} undergoes continuous charging under no-load/grid disconnected condition. This behavior is primarily due to the continuous charging procedure of the zero state. To overcome this issue two solutions were proposed. A simple, but effective solution was to use an extra switch S_5 —anti-parallel to the switch S_4 , with both these switches having the same gating signals. This method was shown to achieve the desired goal of maintaining the flying capacitor voltage near 400 V. However, this violated the actual design of the sFCI converter and was therefore not preferred.

An alternative solution to this problem was proposed, in which the flying capacitor itself was considered as a system state. With this proposition, the system model was redefined to incorporate the dynamics of the flying capacitor and the converter bridge. This led to the derivation of the non-linear converter model, wherein the system changes cases based on various conditions. For verification of the derived non-linear model, a direct MPC technique was proposed for a single-phase sFCI. The control approach was split up into scheme-I and scheme-II, giving rise to a hybrid controller. During scheme-I the controller was tasked to maintain the C_{FC} voltage around its reference, while during scheme-II the controller worked as a reference tracking controller. In this manner the voltage fly-away condition was mitigated and using an extra switch was avoided.

The simulation results highlighted the superior performance of the proposed hybrid controller. On one hand the hybrid control technique—with its operating scheme-I—can control the voltage of the flying capacitor during off-grid conditions, while on the other hand it can work as a reference tracking controller—during the scheme-II. The presented results verify the correctness of the non-linear model and working of the proposed hybrid controller.

■ Finally in Chapter 7, implementation of the state-feedback controller on the hardware testbench was presented for the single-phase sFCI. After a brief description of the test-bench, the design of the control algorithm using C-code was discussed and various modules required for the same were highlighted. Additionally, the procedure adopted for grid connection was detailed. Furthermore, the response of the various modules was also discussed to highlight the performance of the designed control algorithm and affirm the fact that this recently proposed topology can be controlled to deliver desired power in grid connected mode.

Experimental results show the performance of the SFCC for grid current reference tracking and voltage control of the flying capacitor. The performance of the controller in terms of current response is satisfactory, but leaves a lot of room for improvement.

In a nutshell, the presented simulation and experimental results prove that this thesis was able to fulfill the objectives that were stated in the introduction, by providing control algorithms to facilitate grid connection of the sFCI. Although the sFCI has a very simple and robust design, using it during grid-connected mode poses some challenges due to the presence of the flying capacitor. Nevertheless, we can conclude that this newly proposed topology has a lot of potential to serve as an alternative in transformerless inverter applications, but requires further investigation and improvements in the proposed control algorithms.

8.2 Future outlook

Finally, I would like to put forth some recommendations for future endeavors in this topic.

- ♦ In the presented work, the state-feedback control was designed using pole-placement approach which does not give optimal pole locations. An alternative approach would be to use a linear-quadratic-regulator (LQR) for calculation of the feedback gains and then compare the performance of the two approaches.
- ♦ In this thesis, the direct MPC scheme used a simple enumeration based approach. However, alternative techniques to reduce the computational burden could be utilized like branch-and-bound technique, move blocking strategy, and switch and extrapolate method. Accordingly, DMPC with longer prediction horizons could be implemented, which results in a remarkable improvement in the system performance and has the possibility of real-time implementation.
- ♦ The non-linear converter model of the sFCI resembles the piecewise affine systems and can be further refined. To improve the performance of the proposed direct MPC, a move blocking strategy can be employed which can predict the future responses for very large horizons. This has the potential to improve the performance of the proposed control approach and decrease the sampling frequency. Additionally, an integral state can be incorporated into the controller to improve the steady-state response.
- ♦ Finally, the CCS-MPC approach has a potential for real-time implementation, given the fact that it uses a modulator and the sampling time, the controller step time can be set to realistic values. Therefore, this control scheme can be implemented on the hardware test-bench and its response compared to the SFC.

Appendices

Appendix A

Description of the Test Bench



(a) Three level, three-phase prototype of the sFCI constructed at Fraunhofer ISE.



(b) Experimental setup for implementation of state-feedback control using single-phase sFCI.

Figure A.1: Test-bench.

Figure A.1(a) shows the inverter prototype developed previously at Fraunhofer ISE. This setup was used for verification of the inverter operation during open-loop. Figure A.1(b) depicts the single-phase experimental setup that was used for the testing of the controller. The different components

are described below:

- 1, 2, 3 are the phases a, b, and c of the sFCI, respectively.
- 4, 5, 6 are the flying capacitors of phase a, b, and c, respectively.
- 7, 8, 9 are the input capacitors of phase a, b, and c, respectively.
- 10, 11 is the main inductor for the sFCI.
- 12 is the control PCB. It mainly consists of the controller and different measurement circuits.
- 13 is the power PCB. It consists of the topology and the two dc-links.
- 14, 15 are the voltage measurements for flying capacitor and the input dc-link, respectively.

Appendix B

Abbreviations and symbols

B.1 List of abbreviations

Abbreviation	Meaning
ac	alternating current
ADC	analog to digital converter
CMV	common-mode voltage
CCS-MPC	continuous-control-set model predictive control
CLA	control law accelerator
dc	direct current
DVFC	dc voltage feedforward compensation
DMPC	direct model predictive control
FCS-MPC	finite-control-set model predictive control
\mathbf{FC}	flying capacitor
LCL	inductor-capacitor-inductor filter
LQR	linear quadratic regulator
MPC	model predictive control
MLI	multi-level inverter
MCU	microcontroller unit
NPC	neutral point clamped
PV	photovoltaic
\mathbf{PR}	proportional resonant
PLL	phase locked loop
PCC	point of common coupling
PWM	pulse width modulation
PCB	printed circuit board
RB-IGBT	reverse blocking IGBT
sFCI	Siwakoti-H flying capacitor inverter
SFC	state-feedback control
SFCC	state-feedback current control
SOGI	second order generalized integrator
SDFM	sigma-delta filter module
THD	total harmonic distortion
VSI	voltage source inverter
VOC	voltage oriented control
ZOH	zero-order-hold

B.2 List of symbols

Symbol	Meaning
u_{dc}	voltage across dc -link capacitor
u_{FC}	voltage across the flying capacitor
u_m	output voltage of the converter bridge
u_f	voltage across filter capacitor
u_g	grid voltage
i_{dc}	input current to the converter
i_m	current flowing in the main-inductor
i_g	current flowing in the grid-side inductor

Bibliography

- [1] Office of Energy Analysis, U.S. Department of Energy, Washington, DC, U.S. Energy Information Administration: International Energy Outlook 2016, 2016.
- [2] J. Kukkola et al., Estimation Methods for Grid-Voltage Sensorless Control of Converters Equipped with an LCL Filter. PhD thesis, Aalto University, 2016.
- [3] Y. P. Siwakoti and F. Blaabjerg, "Common-ground-type transformerless inverters for singlephase solar photovoltaic systems," *IEEE Transactions on Industrial Electronics*, vol. 65, no. 3, pp. 2100–2111, 2018.
- [4] S. Kouro, J. I. Leon, D. Vinnikov, and L. G. Franquelo, "Grid-connected photovoltaic systems: An overview of recent research and emerging PV converter technology," *IEEE Industrial Electronics Magazine*, vol. 9, pp. 47–61, Mar. 2015.
- [5] Wikipedia contributors, "Growth of photovoltaics Wikipedia, the free encyclopedia." https:// en.wikipedia.org/w/index.php?title=Growth_of_photovoltaics&oldid=859628823, 2018. [Online; accessed 7-September-2018].
- [6] R. González, E. Gubía, J. López, and L. Marroyo, "Transformerless single-phase multilevelbased photovoltaic inverter," *IEEE Transactions on Industrial Electronics*, vol. 55, no. 7, pp. 2694–2702, 2008.
- [7] T. Bruckner, S. Bernet, and H. Guldner, "The active npc converter and its loss-balancing control," *IEEE Transactions on Industrial Electronics*, vol. 52, no. 3, pp. 855–868, 2005.
- [8] Y. Gu, W. Li, Y. Zhao, B. Yang, C. Li, and X. He, "Transformerless inverter with virtual dc bus concept for cost-effective grid-connected pv power systems," *IEEE Transactions on Power Electronics*, vol. 28, no. 2, pp. 793–805, 2013.
- [9] M. Gommeringer, F. Kammerer, A. Schmitt, and M. Braun, "A transformerless single-phase pv inverter circuit for thin-film or back-side contacted solar modules," in *Industrial Electronics Society, IECON 2014-40th Annual Conference of the IEEE*, pp. 1148–1153, IEEE, 2014.
- [10] S. V. Araújo, P. Zacharias, and B. Sahan, "Novel grid-connected non-isolated converters for photovoltaic systems with grounded generator," in *Power Electronics Specialists Conference*, 2008. PESC 2008. IEEE, pp. 58–65, IEEE, 2008.
- [11] O. Stalter, P. Wellnitz, and B. Burger, "Flying-capacitor topology for grounding of single-phase transformer-less three-level photovoltaic inverters," in *Power Electronics and Applications* (EPE'14-ECCE Europe), 2014 16th European Conference on, pp. 1–9, IEEE, 2014.
- [12] S. V. Araújo, P. Zacharias, and R. Mallwitz, "Highly efficient single-phase transformerless inverters for grid-connected photovoltaic systems," *IEEE Transactions on Industrial Electronics*, vol. 57, pp. 3118–3128, Sept 2010.

- [13] Y. P. Siwakoti and F. Blaabjerg, "A novel flying capacitor transformerless inverter for singlephase grid connected solar photovoltaic system," in *Power Electronics for Distributed Generation* Systems (PEDG), 2016 IEEE 7th International Symposium on, pp. 1–6, IEEE, 2016.
- [14] N. Panten, N. Hoffmann, and F. W. Fuchs, "Finite control set model predictive current control for grid-connected voltage-source converters with lcl filters: A study based on different state feedbacks," *IEEE Transactions on Power Electronics*, vol. 31, no. 7, pp. 5189–5200, 2016.
- [15] X. Jinming and X. Shaojun, "Lcl-resonance damping strategies for grid-connected inverters with lcl filters: a comprehensive review," *Journal of Modern Power Systems and Clean Energy*, vol. 6, no. 2, pp. 292–305, 2018.
- [16] H. Miranda, R. Teodorescu, P. Rodriguez, and L. Helle, "Model predictive current control for high-power grid-connected converters with output lcl filter," in *Proc. 35th Annual Conf. of IEEE Industrial Electronics*, pp. 633–638, Nov. 2009.
- [17] R. N. Beres, X. Wang, M. Liserre, F. Blaabjerg, and C. L. Bak, "A review of passive power filters for three-phase grid-connected voltage-source converters," *IEEE Journal of Emerging* and Selected Topics in Power Electronics, vol. 4, pp. 54–69, Mar. 2016.
- [18] G. Shen, D. Xu, L. Cao, and X. Zhu, "An improved control strategy for grid-connected voltage source inverters with an lcl filter," *IEEE Transactions on Power Electronics*, vol. 23, no. 4, pp. 1899–1906, 2008.
- [19] J. Dannehl, C. Wessels, and F. W. Fuchs, "Limitations of voltage-oriented pi current control of grid-connected pwm rectifiers with *lcl* filters," *IEEE Transactions on Industrial Electronics*, vol. 56, no. 2, pp. 380–388, 2009.
- [20] R. Teodorescu, F. Blaabjerg, M. Liserre, and P. C. Loh, "Proportional-resonant controllers and filters for grid-connected voltage-source converters," *IEE Proceedings-Electric Power Applications*, vol. 153, no. 5, pp. 750–762, 2006.
- [21] J. Kukkola and M. Hinkkanen, "Observer-based state-space current control for a three-phase grid-connected converter equipped with an lcl filter," in *Proc. IEEE Energy Conversion Congress* and *Exposition*, pp. 1371–1378, Sept. 2013.
- [22] J. Kukkola, M. Hinkkanen, and K. Zenger, "Observer-based state-space current controller for a grid converter equipped with an lcl filter: Analytical method for direct discrete-time design," *IEEE Transactions on Industry Applications*, vol. 51, no. 5, pp. 4079–4090, 2015.
- [23] J. Dannehl, F. W. Fuchs, and P. B. Thøgersen, "Pi state space current control of grid-connected pwm converters with lcl filters," *IEEE Transactions on Power Electronics*, vol. 25, no. 9, pp. 2320–2330, 2010.
- [24] C. Dirscherl, J. Fessler, C. M. Hackl, and H. Ipach, "State-feedback controller and observer design for grid-connected voltage source power converters with lcl-filter," in *Control Applications* (CCA), 2015 IEEE Conference on, pp. 215–222, IEEE, 2015.
- [25] J. Scoltock, T. Geyer, and U. Madawala, "Model predictive direct power control for a gridconnected converter with an lcl-filter," in *Industrial Technology (ICIT)*, 2013 IEEE International Conference on, pp. 588–593, IEEE, 2013.
- [26] S. Kouro, M. A. Perez, J. Rodriguez, A. M. Llor, and H. A. Young, "Model predictive control: Mpc's role in the evolution of power electronics," *IEEE Industrial Electronics Magazine*, vol. 9, no. 4, pp. 8–21, 2015.

- [27] A. Ayad, Advanced Control Techniques of Impedance Source Inverters for Distributed Generation Applications. PhD thesis, Technische Universität München, 2017.
- [28] S. Vazquez, J. Rodriguez, M. Rivera, L. G. Franquelo, and M. Norambuena, "Model predictive control for power converters and drives: Advances and trends," *IEEE Transactions on Industrial Electronics*, vol. 64, no. 2, pp. 935–947, 2017.
- [29] J. Rodriguez, J. Pontt, C. A. Silva, P. Correa, P. Lezana, P. Cortés, and U. Ammann, "Predictive current control of a voltage source inverter," *IEEE transactions on industrial electronics*, vol. 54, no. 1, pp. 495–503, 2007.
- [30] H. Abu-Rub, J. Guzinski, Z. Krzeminski, and H. A. Toliyat, "Predictive current control of voltage-source inverters," *IEEE Transactions on Industrial Electronics*, vol. 51, no. 3, pp. 585– 593, 2004.
- [31] P. Karamanakos, T. Geyer, N. Oikonomou, F. D. Kieferndorf, and S. Manias, "Direct model predictive control: A review of strategies that achieve long prediction intervals for power electronics," *IEEE Industrial Electronics Magazine*, vol. 8, pp. 32–43, Mar. 2014.
- [32] A. Nabae, I. Takahashi, and H. Akagi, "A new neutral-point-clamped pwm inverter," *IEEE Transactions on industry applications*, no. 5, pp. 518–523, 1981.
- [33] S. Khomfoi and L. M. Tolbert, "Multilevel power converters.," in *Power Electronics Handbook: Devices, Circuits, and Applications* (M. H. Rashid, ed.), pp. 455 486, Butterworth-Heinemann, 2011.
- [34] A. El-Hosainy, H. A. Hamed, H. Z. Azazi, and E. El-Kholy, "A review of multilevel inverter topologies, control techniques, and applications," in *Power Systems Conference (MEPCON)*, 2017 Nineteenth International Middle East, pp. 1265–1275, IEEE, 2017.
- [35] Z. Zhang, On control of grid-tied back-to-back power converters and permanent magnet synchronous generator wind turbine systems. PhD thesis, Technische Universität München, 2016.
- [36] T. Meynard and H. Foch, "Multi-level conversion: high voltage choppers and voltage-source inverters," in *Power Electronics Specialists Conference*, 1992. PESC'92 Record., 23rd Annual IEEE, pp. 397–403, IEEE, 1992.
- [37] A. Palanisamy, "Development of a three-level three-phase boost inverter," Master's thesis, Coburg University of Applied Sciences, 2018.
- [38] M. P. Kazmierkowski and L. Malesani, "Current control techniques for three-phase voltagesource pwm converters: A survey," *IEEE Transactions on industrial electronics*, vol. 45, no. 5, pp. 691–703, 1998.
- [39] J. Rodriguez and P. Cortes, Predictive control of power converters and electrical drives, vol. 40. John Wiley & Sons, 2012.
- [40] M. Liserre, R. Teodorescu, and F. Blaabjerg, "Stability of photovoltaic and wind turbine grid-connected inverters for a large set of grid impedance values," *IEEE transactions on power electronics*, vol. 21, no. 1, pp. 263–272, 2006.
- [41] F. Huerta, D. Pizarro, S. Cobreces, F. J. Rodriguez, C. Giron, and A. Rodriguez, "Lqg servo controller for the current control of *lcl* grid-connected voltage-source converters," *IEEE Transactions on Industrial Electronics*, vol. 59, no. 11, pp. 4272–4284, 2012.

- [42] E. Wu and P. W. Lehn, "Digital current control of a voltage source converter with active damping of lcl resonance," in Proc. APEC 2005. Twentieth Annual IEEE Applied Power Electronics Conf. and Exposition, vol. 3, pp. 1642–1649 Vol. 3, Mar. 2005.
- [43] M. Xue, Y. Zhang, Y. Kang, Y. Yi, S. Li, and F. Liu, "Full feedforward of grid voltage for discrete state feedback controlled grid-connected inverter with lcl filter," *IEEE Transactions on Power Electronics*, vol. 27, no. 10, pp. 4234–4247, 2012.
- [44] M. Erhard, G. Frison, and M. Diehl, Energy Systems: Hardware and Control Lecture Notes for Control Part. University of Freiburg, Nov. 2017.
- [45] D. Rowell, Handouts: Analysis and Design of Feedback Control Systems, p. Sec 2.14. MIT OpenCourseWare, 2002.
- [46] B. Stellato, T. Geyer, and P. J. Goulart, "High-speed finite control set model predictive control for power electronics," *IEEE Transactions on Power Electronics*, vol. 32, pp. 4007–4020, May 2017.
- [47] J. M. C. Geldenhuys, "Model predictive control of a grid-connected converter with lcl-filter," Master's thesis, Stellenbosch: Stellenbosch University, 2018.
- [48] P. Cortes, J. Rodriguez, C. Silva, and A. Flores, "Delay compensation in model predictive current control of a three-phase inverter," *IEEE Transactions on Industrial Electronics*, vol. 59, no. 2, pp. 1323–1325, 2012.
- [49] M. Rivera, J. Rodriguez, B. Wu, J. R. Espinoza, and C. A. Rojas, "Current control for an indirect matrix converter with filter resonance mitigation," *IEEE Transactions on Industrial Electronics*, vol. 59, no. 1, pp. 71–79, 2012.
- [50] S. Mariéthoz and M. Morari, "Explicit model-predictive control of a pwm inverter with an lcl filter," *IEEE Transactions on Industrial Electronics*, vol. 56, no. 2, pp. 389–399, 2009.
- [51] S. Almér, S. Mariethoz, and M. Morari, "Sampled data model predictive control of a voltage source inverter for reduced harmonic distortion," *IEEE Transactions on Control Systems Technology*, vol. 21, no. 5, pp. 1907–1915, 2013.
- [52] T. Nguyen-Van and N. Hori, "New class of discrete-time models for non-linear systems through discretisation of integration gains," *IET Control Theory & Applications*, vol. 7, no. 1, pp. 80–89, 2013.
- [53] P. Karamanakos, T. Geyer, and S. Manias, "Direct voltage control of dc-dc boost converters using model predictive control based on enumeration," in 2012 15th International Power Electronics and Motion Control Conference (EPE/PEMC), pp. DS2c.10–1–DS2c.10–8, Sept 2012.
- [54] M. A. W. Begh, S. Agarwal, and Z. Zhang, "Predictive current control of modular multilevel converter for hvdc system with online weighting factor tuning using fuzzy logic," tech. rep., Technische Universitat München, 2017.
- [55] T. Geyer, Model predictive control of high power converters and industrial drives. John Wiley & Sons, 2016.
- [56] E. Camacho and C. Bordons, *Model predictive control*. Springer Verlag, 1999.
- [57] T. Bruckner, S. Bernet, and H. Guldner, "The active npc converter and its loss-balancing control," *IEEE Transactions on Industrial Electronics*, vol. 52, pp. 855–868, June 2005.

- [58] C. Dirscherl, C. M. Hackl, and K. Schechner, "Modellierung und regelung von modernen windkraftanlagen: Eine einführung," in *Elektrische Antriebe – Regelung von Antriebssystemen* (D. Schröder, ed.), pp. 1540–1614, Berlin: Springer-Verlag, 2015.
- [59] R. H. Park, "Two-reaction theory of synchronous machines generalized method of analysis-part i," *Transactions of the American Institute of Electrical Engineers*, vol. 48, no. 3, pp. 716–727, 1929.
- [60] A. Julean, "Active damping of lcl filter resonance in grid connected applications," Master's thesis, Aalborg University, 2009.
- [61] P. A. Dahono, "A method to damp oscillations on the input lc filter of current-type ac-dc pwm converters by using a virtual resistor," in *Telecommunications Energy Conference*, 2003. INTELEC'03. The 25th International, pp. 757–761, IEEE, 2003.
- [62] W. Gullvik, L. Norum, and R. Nilsen, "Active damping of resonance oscillations in lcl-filters based on virtual flux and virtual resistor," in *Power Electronics and Applications*, 2007 European Conference on, pp. 1–10, IEEE, 2007.
- [63] P. W. Lehn and M. R. Irvani, "Discrete time modeling and control of the voltage source converter for improved disturbance rejection," *IEEE Transactions on Power Electronics*, vol. 14, no. 6, p. 1028–1036, 1999.
- [64] F. Rodriguez, E. Bueno, M. Aredes, L. Rolim, F. A. Neves, and M. C. Cavalcanti, "Discretetime implementation of second order generalized integrators for grid converters," in *Industrial Electronics*, 2008. IECON 2008. 34th Annual Conference of IEEE, pp. 176–181, IEEE, 2008.
- [65] P. Rodriguez, A. Luna, I. Candela, R. Teodorescu, and F. Blaabjerg, "Grid synchronization of power converters using multiple second order generalized integrators," in *Industrial Electronics*, 2008. IECON 2008. 34th Annual Conference of IEEE, pp. 755–760, IEEE, 2008.
- [66] X. Yuan, W. Merk, H. Stemmler, and J. Allmeling, "Stationary-frame generalized integrators for current control of active power filters with zero steady-state error for current harmonics of concern under unbalanced and distorted operating conditions," *IEEE transactions on industry applications*, vol. 38, no. 2, pp. 523–532, 2002.
- [67] J. A. Rossiter, "Videos on modelling, control and analysis." https://www.sheffield.ac.uk/acse/ staff/jar/feedbackchapter, 2018. [Online; accessed 24-July-2018].
- [68] G. F. Franklin, J. D. Powell, and M. L. Workman, *Digital control of dynamic systems*, vol. 3. Addison-wesley Menlo Park, CA, 1998.
- [69] IEEE, *IEEE recommended practice for utility interface of photovoltaic (PV) systems.* IEEE, 2000.
- [70] G. Ellis, "Chapter 4 the luenberger observer: Correcting sensor problems," in Observers in Control Systems (G. Ellis, ed.), pp. 67 – 96, San Diego: Academic Press, 2002.
- [71] R. Teodorescu, M. Liserre, and P. Rodriguez, Grid converters for photovoltaic and wind power systems, vol. 29. John Wiley & Sons, 2011.
- [72] S. Kouro, P. Cortés, R. Vargas, U. Ammann, and J. Rodríguez, "Model predictive control—a simple and powerful method to control power converters," *IEEE Transactions on industrial electronics*, vol. 56, no. 6, pp. 1826–1838, 2009.

- [73] T. Geyer, P. Karamanakos, and R. Kennel, "On the benefit of long-horizon direct model predictive control for drives with lc filters," in *Energy Conversion Congress and Exposition* (ECCE), 2014 IEEE, pp. 3520–3527, IEEE, 2014.
- [74] T. Geyer and D. E. Quevedo, "Performance of multistep finite control set model predictive control for power electronics," *IEEE Transactions on power electronics*, vol. 30, no. 3, pp. 1633–1644, 2015.
- [75] T. Geyer and D. E. Quevedo, "Multistep finite control set model predictive control for power electronics," *IEEE Transactions on power electronics*, vol. 29, no. 12, pp. 6836–6846, 2014.
- [76] B. Stickan, S. Reichert, and M. Diehl, "Model predictive control of a 3-phase grid-connected b6-inverter with lcl-filter in abc-coordinates," in 2017 19th European Conference on Power Electronics and Applications (EPE'17 ECCE Europe), pp. P.1–P.9, Sept 2017.
- [77] T. Instruments, Delfino TMS320F28379D controlCARD R1.3 Information Guide, Oct. 2015.
- [78] M. Ciobotaru, R. Teodorescu, and F. Blaabjerg, "A new single-phase pll structure based on second order generalized integrator," in *Power Electronics Specialists Conference*, 2006. *PESC'06. 37th IEEE*, pp. 1–6, IEEE, 2006.